

The Role of Governmental Educational Institutions in Implementing Talents' Development Programs in the UAE: An Explanatory Study

دراسة حول دور المؤسسات التعليمية الحكومية في تنفيذ برامج تنمية المواهب في دولة الإمارات العربية المتحدة: دراسة استكشافية

by BASEMA YOUNIS

A thesis submitted in fulfilment of the requirements for the degree of DOCTOR OF PHILOSOPHY IN EDUCATION

at

The British University in Dubai

November 2019



The Role of Governmental Educational Institutions in Implementing Talents' Development Programs in the UAE: An Explanatory Study.

دراسة حول دور المؤسسات التعليمية الحكومية في تنفيذ برامج تنمية المواهب في دولة الإمارات العربية المتحدة: دراسة استكشافية

By

BASEMA YOUNIS

A thesis submitted to the Faculty of Education

In fulfilment of the requirements for the degree of

Ph.D. IN EDUCATION

at The British University in Dubai November 2019

Thesis Supervisor

Prof Eman Gaad

Name	Name
Designation	Designation
Name	Name

Annroyed for award.

DECLARATION

I warrant that the content of this research is the direct result of my own work and that any use made in it of published or unpublished copyright material falls within the limits permitted by international copyright conventions.

I understand that a copy of my research will be deposited in the University Library for permanent retention.

I hereby agree that the material mentioned above for which I am author and copyright holder may be copied and distributed by The British University in Dubai for the purposes of research, private study or education and that The British University in Dubai may recover from purchasers the costs incurred in such copying and distribution, where appropriate.

I understand that The British University in Dubai may make a digital copy available in the institutional repository.

I understand that I may apply to the University to retain the right to withhold or to restrict access to my thesis for a period which shall not normally exceed four calendar years from the congregation at which the degree is conferred, the length of the period to be specified in the application, together with the precise reasons for making that application.

Signature of the student

COPYRIGHT AND INFORMATION TO USERS

The author whose copyright is declared on the title page of the work has granted to the British University in Dubai the right to lend his/her research work to users of its library and to make partial or single copies for educational and research use.

The author has also granted permission to the University to keep or make a digital copy for similar use and for the purpose of preservation of the work digitally.

Multiple copying of this work for scholarly purposes may be granted by either the author, the Registrar or the Dean only.

Copying for financial gain shall only be allowed with the author's express permission.

Any use of this work in whole or in part shall respect the moral rights of the author to be acknowledged and to reflect in good faith and without detriment the meaning of the content, and the original authorship.

Abstract

UAE educational institutions implement gifted and talented development programs; therefore, this study was the first to examine the application of these programs in UAE schools. The main research question of the study was to examine the efficiency of this implementation and the accuracy of outcomes of the applied programs over several stakeholders. The study adopted a qualitative explanatory method design to collect the data. The collection method included interviews with parents of the gifted and talented, teachers, programs' managers, experts in the field, gifted and talented students and adults, in addition to a focus group with parents, teachers, and talented individuals A review of documents was also conducted. The conclusions highlighted diverse deficits in the implementation of several program components that have arisen from findings, and room for thorough improvement is necessary. The last chapter included a set of recommendations presented to better serve the highly able students in the UAE in future.

Keywords:

Talented individuals- Institutions.

Abstract (Arabic)

تقوم المؤسسات التعليمية في دولة الإمارات العربية المتحدة بتنفيذ برامج خاصة لاكتشاف وتنمية الموهوبين في المدارس؛ وبالتالي، كانت هذه الدراسة هي الأولى من نوعها لاستكشاف دور خمس مؤسسات تعليمية في تنفيذ برامج تحديد الموهوبين وتطوير هم. كانت الأسئلة البحثية الرئيسية للدراسة هي مراجعة دور هذه المؤسسات في تنفيذ هذه البرامج وكيفية تقييم نتائجها من خلال اجراء مقابلات للعديد من أصحاب المصلحة. تم اعتماد تصميم طريقة البحث النوعي التوضيحي. تضمنت الأساليب المستخدمة لجمع البيانات مقابلات مع أولياء الأمور والموهوبين والمعلمين ومديري البرامج والخبراء في مجال الموهوبين والطلاب الموهوبين، بالإضافة إلى مجموعة تركيز مع أولياء الأمور والمعلمين والخبراء والطلاب الموهوبين. كما تم إجراء مراجعة للوثائق الرسمية التابعة لهذه المؤسسات. ووجدت الاستنتاجات عدة فجوات في هذه البرامج مع وجود مجال للتحسين اللازم. وتقدم الدراسة في الختام مجموعة من التوصيات لخدمة الطلاب الموهوبين والمتفوقين في دولة الإمارات العربية المتحدة بشكل أفضل في المستقبل.

الكلمات الرئيسية: الموهوبون-المؤسسات

Dedication

To the loving mother, brothers, and sisters and children who have given all the support needed to reach my dream, I dedicate this thesis.

Acknowledgement

I want to express my sincere gratitude to My advisor, Prof Eman Gaad, for her care and dedication within the whole process of my thesis, for her continuous support, for her patience, motivation, enthusiasm, and immense knowledge. Her guidance helped me in all the time of research and writing of this thesis, her selfless time and care were sometimes all that kept me going.

Exceptional gratitude goes out to Prof Abdulla Al Shamsi, Vice-Chancellor of British University in Dubai, who has encouraged and supported me all the way.

I want to thank my mother, brothers and sisters whose love and guidance are with me in whatever I pursue.

Most importantly, I wish to thank my elder brother Dr Basem for his support, encouragement, insightful comments, and hard questions.

Special gratitude goes to my amazing children Mohammed Ali, AbdulRahman, Hazzaa and AlYazia who provide constant inspiration.

Table of Contents

Table of Contents	<i>i</i>
List of Tables	vi
List of Abbreviations & their Explanations	viii
List of Appendices	ix
Chapter One. Introduction	1
Introduction of the Chapter	1
Key Words	6
Background of the study	6
Purpose and Rationale	9
Research Questions	9
Significance and Research Problem	11
Limitations and anticipated Challenges	12
Organisation of Chapters	
Chapter Two. Literature Review	14
Introduction	14
Global Gifted and Talented Debates	
Theoretical Framework	25
Definition of Talent	25
Evolution of the definition	26
History of Definition Evolution	29
Identification of Gifts &Talents	32
Development of Gifts &Talents	35
Stages of Talent Development	37
Assessing Achievement	38
Gifted education & Development Programs	39
The US Experience	39
The Experience of China	45
Intelligence and talents	49

Defining Intelligence	
Nature of Intelligence	55
Intelligence tests	
Criticising intelligence view	58
Theories of Intelligence	62
Robert Sternberg: Triarchic Theory of Intelligence	64
Criticising Robert Sternberg's Theory of Intelligence	66
Howard Gardner. The Theory of Multiple Intelligence	67
Criticising Howard Gardner's Theory of Multiple Intelligence	69
Gagne's Differentiated Model (DMGT)	73
Criticizing Gagne's Differentiated Model (DMGT)	76
Three-Ring Conception of Giftedness	77
Criticising the Three Rings Model of Giftedness	79
Employing the Theory of Multiple Intelligences	81
Chapter 3. Institutions related to supporting gifted and talented programs in the UAE	85
Introduction	85
Rationale and Purpose	85
The Role of Educational Institutions in the UAE	88
Sources of Information	89
An Overview	90
Gifted Programs	92
Definition of Gifted Programs	94
Gifted Programs Requirments	94
Gifted programs in the UAE	96
Identification of the gifted and talented in the UAE	98
Assessment of progress	100
UAE Educational Institutions	100
At the federal level	101
The Ministry of Education (MOE)	101
At the local level. Emirate of Abu Dhabi	101
The (Abu Dhabi Department of Education and Knowledge) (ADEK)	101
Emirate of Dubai	101
Knowledge & Human Development Authority (KHDA)	101
Emirate of Sharjah. Sharjah Special Education Authority (SSEA)	102
In the Northern Emirates	102

Education in the UAE	102
Present & Future	102
The UAE National Agenda	102
First: (MOE) Ministry of Education	104
Emirates Foundation for School Education	106
Development of Gifted and talented	106
MOE Definition & Identification of Gifted and Talented	107
Example of a development program at (MOE)	108
Masterpieces	108
(2) Our Ambassadors Program	108
Talents Identification for Masterpieces	109
Second: (MCKD) Ministry of Culture & Knowledge Development	109
Gifted and talented identification and development	110
Policy for Talented	110
Example of an (MCKD) Development program	111
Nurturing Talented Authors	111
Third: (ADEK) Abu Dhabi Department of Education and Knowledge	113
The role of (ADEK) in Education	114
(ADEK) Gifted & Talented Development Initiatives	114
Definition of gifted and talented	114
(ADEK) Development Programs & Initiatives	115
An example of a development program at (ADEK)	116
Mawhibaty Programs	116
Fourth: (KHDA) Knowledge & Human Development Authority	117
The Role of (KHDA) in Education	118
KHDA definition & Identification of gifted and talented	118
(KHDA) Gifted & Talented Developmental programs	119
Examples of (KHDA) development program	119
Human Talent Map	119
(3) The National Institute for Vocational Education (NIVE)	120
(4) Young editors	
Fifth: (HA) Hamdan Foundation for Distinguished Academic Performance	
The role of (HA) in Education	
The Role of (HA) in implementing Gifted & Talented Development	
Definition & identification of the gifted	

Stages of ide	entification	
National Pla	n for Gifted programs	124
Examples of	(HA) development programs	125
(1) The F	Professional Diploma Program in Gifted Education	125
(2) The F	Hamdan Center for Talent and Creativity	126
The relevance	ce of the Institutions Focused Study	126
Chapter Fou	ır: Methodology	128
Introduction.		128
Research Ap	proach	128
Research Str	rategy & Design	130
The Case Stu	udy Strategy	132
Case Study a	advantages & Limitations	134
Research De	esign-Structure	135
Research pai	radigm	146
Population, i	research sample and sampling methods	147
Purposeful s	ampling	148
Ethical Cons	siderations	149
Utilitarian		151
Confidential	ity	152
Relational E	thics	152
Data Collect	tion Methodology	135
Library rese	arch and document analysis	136
Field researd	ch: Focus groups and interviews	139
Designing th	ne interview	140
Participants	in the interviews	141
Registration	and recording of information	143
Validity, Tru	stworthiness, and Reliability	154
Types of vali	idity	154
Reliability		
Construct va	ılidity	156
Other measu	res of validity	157
Reliability of	f the study instrument	157
Reliability in	the Cases Studies	157
Triangulatio	n	158
Focus Groun	os & Protocol	

Facilitation & Note Taking	161
Data Collection Challenges	
Data Analysis	
Choice of data analysis	
Conclusion	
Chapter Five. Presentation of the data	
Introduction	169
Results from the Interviews	
Findings from the interviews	
The Interviews Analysis	
(1) Knowledge and awareness	
(2) Definitions and identification of Giftedness & Talents	179
A-Definition	
B-Identification	
(3)-Development & support	
(4) Challenges	
(a)Challenges in knowledge and awareness	
(b)Challenges in definition and identification	199
(c) Challenges in development and process	200
Results from focus groups	201
Findings from Focus Groups	202
Analysing focus group discussions	203
Evidence from Focus Groups discussions	207
The Parents Group	207
Evidence from the parents' group	207
The Experts' group	209
Evidence from the experts' group	209
The Gifted and Talented focus group	210
Evidence from the Gifted and Talented focus group	211
Evidence from Documentation	213
Knowledge and Awareness	213
(2) Definition & Identification	217
Definition of talents (MOE)	217
Definition of talents (MCKD)	217
Definition of talents (ADEK)	218

Definition of talents (KHDA)	218
Definition of talents (HA)	219
Identification of talents (MOE)	219
Identification of talents (MCKD)	220
Identification of talents (ADEK)	220
Identification of talents (KHDA)	221
Identification of gifted (HA)	222
(3) Development & Processes	223
Development of Talents in the UAE	223
Development of Talents (MOE)	224
Development of Talents (MCKD)	225
Development of Talents (ADEK)	225
Development of Talents (KHDA)	226
Development of Gifts (HA)	226
(3) Challenges	227
Chapter Six. Discussions, Recommendations & Conclusion	230
Introduction	230
Discussing Key Findings	231
Awareness and knowledge,	231
Gifted and talented ecosystem	231
RQ(1)	232
Recommendations	244
Table 39.Interview Questions in alignment with recommendations	244
National Policy	245
Unified Definition	249
Identification Process	251
Development Process & Systems	253
Recommendation for Future Research	257
Conclusion	258
References	261
Online Resources	275
Appendix A	281
Appendix B	283
Appendix C	284

List of Tables

Table 1. Vision & Mission in General Rules for the provision of Special Education Table2. Developmental Theories	12 42
Table 3.Differences between talent and intelligence	54
Table 4. Sternberg's classification of giftedness	69
Table 5. Summary of Theories of Intelligence	88
Table 6. List of UAE Educational Institutions	93
Table 7. Three stages of the identification process by (NPGP)	103
Table 8. MOE Vision & Mission- Strategic Objectives 2017-2021	109
Table 9. MCKD Vision & Mission	113
Table.10. (MCKD) Talents' Nurturing Policy Themes	115
Table11. ADEK Vision & Mission-Strategic Objectives (adek.gov.ae)	117
Table 12. KHDA Vision & Mission-Strategic Objectives	121
Table 13: (HA) Hamdan Foundation Vision & Mission-Strategic Objectives.	125
Table 14. Tools and Metrics for Gifted Students at	128
Table 15. (NPGP) Programs	129
Table 16. (PDPG) Objectives (ha.ae)	129
Table 17. Hamdan Center for Talent and Creativity -Objectives	130
Table 18. Denscombe's Elements of a case study	137
Table 19. Advantages & Disadvantages of a Case Study	139
Table 20. Strengths and weaknesses in document analysis	140
Table21. Some of the documents reviewed for the sake of this study	141
Table 22. Summary of advantages of Triangulation	163
Table 23. Key themes addressed by the interviews	163

Table 24. Summary of Focus Groups (advantages & disadvantages)	167
Table25. Summary of stages for data analysis	172
Table 26. Alignment of data sources, participants and research questions	173
Table 27. Participants in the interviews	174
Table 28. Interview questions in alignment with the $RQ(1)$	176
Table.29. Interviews' results in alignment with (RQ1)	177
Table 30. Interview questions in alignment with RQ (2)	183
Table 31.Interview results in alignment with (RQ2)	184
Table 32. Interview questions in alignment with RQ (3)	190
Table 33. Interview results in alignment with $RQ(3)$	190
Table 34. Interview questions in alignment with $RQ(4)$	199
Table 35. Interview results in alignment with $RQ(4)$	200
Table 36. Participants in Focus Groups	205
Table 37. Themes analysis	209
Table 38. Theme Analysis in alignment with research questions	247
Table 39.Interview Questions in alignment with recommendations	248

List of Abbreviations & Explanations

Abbreviation	Explanation	Origin
UAE	United Arab Emirates	UAE
UAEU	United Arab Emirates University	UAE
MOE	Ministry of Education	UAE
MCKD	Ministry of Culture & Knowledge Development	UAE
ADEK	Abu Dhabi Education and Knowledge	UAE
KHDA	Knowledge & Human Development Authority	UAE
HA	Foundation of Hamdan Awards	UAE
NPGP	The National Plan for Gifted Programs	UAE
PDPG	Program for Teachers' Development in Gifted Education	UAE
НСТС	Hamdan Center for Talents and Creativity	UAE
MCD	Ministry of Community Development	UAE
MOP	Ministry of Possibilities	UAE
DoT	Department of Talents	UAE
CSS	UAE Centre for strategic Studies	UAE
Ted	Talents Education	UAE
<i>ESRCS</i>	Emirates Center for Strategic Studies and Research	UAE
ADEZ	Abu Dhabi Educational Zone	UAE
DCT	Department of Tourism and Culture	UAE
NIVE	National Institute for Vocational Education	UAE
NQA	National Quality Authority	UAE
GCC	Gulf Countries Council	Gulf & Universal
WCGTC		Universal
IEP		Universal
ALP	Advanced Learning Plan	Universal
GATE	Gifted & Talented Education	Universal
GAT		Universal
G&T	Gifted & Talented	Universal
SAT	Secondary Assessment Test	Universal
IB	International Baccalaureate	Universal
AP	Advanced Placement	Universal
IQ	Intelligence Quotient	Universal
AI	Artificial Intelligence	Universal
IEP	Individual Education Program	Universal
STEAM	Science, Technology, Engineering, Arts and Mathematics	Universal
R&D	Research & Development	Universal
NAGC	National Association for Gifted Children	US
NCLB	The No Child Left behind Act	US
MI	Multiple Intelligence Theory	US
DMGT	Gagne's Differentiated Model of Giftedness and Talent	France
APA	American Psychological Association	US

List of Appendices

Appendix A: Interview Questions Guide: Experts-Teachers-Programs Leaders.

Appendix B: Interview Questions Guide: Parents of Talented Students.

Appendix C: Interview Questions Guide: Talented Students.

Chapter One. Introduction

1 Introduction of the Chapter

The oil depression and the enormous scientific developments urged a competition in the globe and made worldwide countries eager to build a knowledge-based society, that will move on with the future advancements. To stay within the platform of the advancing countries, the UAE entered the global competition towards a future that looks at creativity and innovation as a new, more reliable economy rather than oil. This competition raised the awareness of the importance of human resources and specifically the highly able persons' creativity and innovations. Gifted and talented individuals, therefore, are the country's main assets, investments and key to success. Consequently, this mission required the UAE to set an ecosystem that can support a deliberate, consistent search for gifted and talented individuals. This ecosystem should support the country's transiting from being an oil-reliable nation to profit from gifted and talented skills. Accordingly, this requires adopting a sustainable strategy that promotes the creative industries and supports the highly able productivity in this sector. The way to achieve success, as usual, starts from and infusing and inspiring education. The mission of the UAE focused primarily on transforming its educational institutions from providing education only to become a search engine and a skills booster for gifted and talented students(government.ae). Once again, this frantic transmission required more than a desire or a coherent system to find more gifted and talented individuals and to inspire their productivity. It needs a substantial well-set ecosystem, built on numerous successful plans and strategies and a different type of education that would achieve the UAE dream. Subsequently, the UAE government stepped in and established several educational institutions

delegated specifically to reform the existing educational system, including the surrounding environment, students and their educators' well-being.

These institutions exerted efforts and took responsibility to apply reforms to the existing educational system. They invested in additional identification programs to discover more highly able students. However, and in the light of not having ongoing researches or studies that document these achievements, such efforts are not seen nor evident for many others. A few pieces of research' findings and results would not inform how these institutions achieved and are achieving their roles. What did they do to implement the required improvements in the educational system? How did they prepare the ecosystem to thrive the highly able creativity and inspire them to innovate? Without having continuous studies or reports, it is more challenging to know how well they were able to organise and collaborate their efforts towards a national mission as well; what did they do to boost the discovery and development of gifted and talented students in schools? Primarily, gifted and talented students have always been a rising issue in the realm of education not only in the UAE but in the entire world. To identify students as exceptionally or profoundly gifted, according to (Feldhusen, 1993) this requires a unique individual, differentiated learning plan. The responsibility for discovering and developing an individual's giftedness or talents has always been a joint responsibility shared between the individual who has the talent and the surrounding ecosystem. Today, serving talented and gifted individuals is not limited to, identification tools or development programs only. Development programs or individual learning plans aims solely at sharpening talents abilities and skills, however, the essential development interprets by designing an utterly thriving life journey that would ensure ongoing systematical service to meet all moral, psychological, educational, economic, emotional, and future needs of this category, at all stages and ages. Prominent scholars like (Alfred Binet, Terman, Gagne, Galton,

Hollingworth, Tannenbaum, Talyor & Renzulie) were of the pioneers in this field by promoting the needs of the gifted and talented students need. They didn't stimulate and inspire their countries' researchers and decision-makers only but their theories and researches were able to spread in many other countries as well to support the exceptional human resources by inspiring them to initiate various talent's development programs. Their researches and theories were followed by different researchers and they were able to open the way for other theorists who came with them or after them such as Gardner, The actual value of being talented and gifted, according to them, is by emphasising giving them differentiated care, attention, and education. Such programs are supposed to identify those gifted and talented individuals and satisfy their potentials (Hodge & Kemp, 2000). Moreover, (Watson, 2006 & Hew, 2007) pointed out that in the shadow of the digital revolution and if the conventional educational system is not capable of satisfying the needs of the society, though, what about the needs of those with extraordinary abilities? Is the current ecosystem qualified to nurture them, or is it unable to do so anymore?

Consequently, several questions wandered amongst interested researchers and educators in the area of the highly able individuals. (Watson, 2006 & Hew, 2007) For example, suggested nurturing these highly able students by a specialised gifted education that offers them a differentiated type of training. This education, according to (Watson 2006 & Hew, 2007), depends on various methods and programs such as acceleration and other motivational tools and applications. Their suggestion drew the field's attention to the possibility of implementing a differentiated educational system. This system separates the highly able students from their peers, and it should be able to satisfy their potential and unique demands as well. This suggestion, on the contrary, created criticism by other educators who opposed the idea. The critics assumed the problem is not in implementing a differentiated educational system, but it lies in other real factors that should analyse in more depth.

The conflated ideas regarding differentiation have led to inequality in schools and classrooms in contrary to it expects. Differentiation that might appear to be successful and a great solution, is not helping to look at other real facets affect these highly abled such as the environment, surrounding the society, attitudes and sociobiology(Taylor, S., 2017).

Furthermore, other questions enquired about what happens if the identification tool or method failed to identify a talented or a gifted student? What if this individual fell inadvertently because his or her talent was challenging to identify by that specific tool?"

It again tempted the scholars to look at the other side of the challenge. This time it was the definition itself. The definition, according to a long history of alterations and evolutions, is causing a conflict. The numerous definitions proposed by several scientists and through different theories tried to describe the concepts of talents and giftedness. The descriptions were dissimilar and should be defined separately (Gagne, 2004).

Theorists saw it challenging to agree on one inclusive definition that can clearly define the significance of talents and could explain the meaning of giftedness. It was more challenging in their opinion than unifying an international definition that aims to describe individuals with special needs. Subsequently, the conflicts and massive views heated the debates again, and it focused more on the definition and the identification conflicts. According to several researchers in this area, the differences between the gifted and talented basis on their intellectual abilities and interests and not on their needs or behaviour (Betts, G.T. and Neihart, M., 1988). Such dispute urged more educators and experts to search for alternatives to support their views and provide this category with their needs. These alternatives were to design advanced educational development programs and introduce them to these highly able students as enrichment programs. However, and as these

enrichment programs do not comprise a curriculum, they are usually subject to the concept of "take it or leave it".

Therefore, the current study was born to explore these institutions achievements in their roles to define, identify and implement the development of the gifted and talented students in the UAE; to examine their mechanisms regarding collaborating to fulfil the national requirements in this regard; what challenges these highly able students faced and what can be suggested to help them overcome these challenges.

Finally, and for the sake of the topic, it was necessary to obtain the information from reliable, reputed resources. Consequently, and in light of the enormous shortage in researches and studies in the UAE, in this area and many others, this study has chosen to explore the reality of gifted and talented identification and development through a variety of methods such as interviews, focus groups, library search and document reviews. The five educational institutions considered the leading entities responsible for primary educational initiatives in the UAE were the subject of the study. The inspiring reason for wanting to study and review these institutions is because: (1) they implement regular and special education and developmental programs for entirely students in the whole UAE. (2) They have been among the prominent chief educational institutions since the inception of the UAE, (MOE inception was in 1971). (3) Since their establishment, these institutions have shown a concrete and sustainable role in the educational field in the UAE and have assumed responsibilities related to education in general and gifted and talented students' initiatives precisely. (4) The main reason for reinforcing the importance of referencing these sources is their responsibility for supporting the discovery and nurturing of talented and gifted students from different aspects and areas. (5) Being part of the UAE governmental institutions with full support to carry out their missions that go in line with the country's vision and ambitious

aspirations of the future. (6) Moreover, these five educational institutions enjoy plenty of opportunities and considerable funds to do everything possible to improve their operations and advance their outputs. In brief, these educational institutions are the utmost suitable institutions to answer this study's questions regarding the fundamentals of how they implement identification and development for the gifted and talented students in the UAE.

Key Words

Gifted and talented Individuals: Individuals with motivation and potential, committed to achieving in different areas and domains of gifts and talents at high levels compared with their peers from the same age groups.

Educational Institutions: Entities that provide a large variety of learning environments and programs where people gain an education, and gifted and talented students receive services like enrichment programs and initiatives.

1.1 Background of the study

The development of the educational institutions in the UAE is the responsibility of the educational institutions, subject of this study. Their role is to provide citizens and residents in the UAE with their needs for the appropriate education. This role is crucial. They have to provide ordinary students with their needs in a regular curriculum, in addition to giving the gifted and talented students their equal needs for development including gifted and talented education (GATE) and individual development programs (TAG). These institutions, if compared to global specialised institutions' in charge of gifted and talented development, can maintain their role successfully if they had a strategy to organise their roles appropriately (davidsonistitute.org). They implement countless enrichment programs considered as the main tools for the development of gifted and

talented. However, the benefit of such programs is not evident because they do not follow a differentiated particular framework, in addition to the absence of a policy that regulates their implementation and imposes standards and periodical evaluations to estimate their benefits.

The development programs in these institutions formulate numerous developmental initiatives and programs that intend to challenge those student's capabilities and fulfil their curiosity. However, stakeholders do not recognise such initiatives and programs as crucial or necessary. Having a policy that organises, safeguards and provides this category with their needs of qualified programs that ought to inspire them and boost their creativity is a key issue in this process (Sebring & Tussey, 1992). Furthermore, the absence of a national policy suggests the absence of the basic principles that should take place while implementing any development programs to nurture those extraordinary students. This study, consequently, takes the development side of the argument. It is born to support the UAE's ambitions towards nurturing the gifted and talented students and individuals as their future income boosters. The UAE sought this dream since its inception and exerted efforts to grasp its achieved success in various missions since then. It was able to become the first in the Arab World, and the (41st) in the global index of innovation in 2016. One aspect of the UAE's success was its support given to individuals with different domains of gifts and talents. The improvements imposed on the educational system was one of the essential objects to provide this category with care and specialised services, and the UAE's support given for creativity and talent has advanced its global ranking six levels in 2016 compared to 2015. On the other hand, and according to a survey field study conducted through (2012-2016) in collaboration amongst Hamdan Bin Rashid Talents Award (HA) and the UAE Centre for Strategic Studies (CSS), the study that explored the gifted and talented' status in the UAE said that their situation is still vague, and their specialized services appear as inconsistent." (Libary.ha.ae, 2016). Such opinion links the

(CSS) study results with the main issue that is continuously infuriating the field regarding the absence of federal policy. This policy that guides, specify, and clarifies the appropriate services for this category would be a solution to understand the current situation and be able to assess it. (Al Obaidly, 2006). The other shortage, and according to the same research, (Library.ha.ae, 2016) comes from the lack of collaboration amongst these educational institutions. Their missing collaboration to identify and develop the gifted and talented students in the UAE is making the scene more blurry and unorganised. Having multiple unlinked efforts leads to the existence of vague, unclear roles (A Field Study, MCKD. 2012).

This (MCKD, 2012) field study conducted in 2012, was one of the leading studies that investigated the (MOE) General Rules for the provision of Special Education Programs and services in Public and Private Schools in the UAE. The study conducted and published by the Ministry of Culture & Knowledge Development (MCKD) indicated that the (MOE) General Rules include regulations, and a law customised to safeguard the rights and education of disabled individuals only. The (MOE) accordingly implements a unified federal national policy to empower, nurture, and safeguard the students and people with special needs and disabilities. The gifted and talented included with special needs by the (MOE) mission. However, excluded from their policy, and there is no specific policy that arranges and regulates services and rights of the gifted and talented compared with what the people with disabilities have.

Table 1. Vision & Mission in General Rules for the provision of Special Education Programs and Services (Public & Private Schools). (MOE).

Vision for Special Education	Mission for Special Education
To provide educational programs and associated	To use all available resources and to plan, implement,
services to all students with special needs and gifts and	and monitor the provision of special programs. To plan,
	implement, and watch the related services provided to
talents in public and private schools in the UAE. These	students with special needs, gifts, and talents. To ensure

services will reflect the best international standards and practices to prepare them to be productive members of society. (school for all,moe.gov.ae)

they all children in the UAE receive an Individual Education Program (IEP) or Advanced Learning Plan (ALP) based on their strengths and needs to enhance their social competence, to enable them to maximise their contributions to their communities. (school for all, moe.gov.ae)

This study, therefore, targets to review the roles and practices of five educational institutions in charge of educational development in the UAE. The review is in respect of their responsibility to achieve the UAE's vision 2021 by exerting more efforts to find and nurture gifts and talents within all individuals living and residing in the country. It also aims to understand the extent of their collaboration to achieve the UAE's National Agenda requirements and to satisfy the country's high-level ambitions.

1.2 Purpose and Rationale

This study intends to explore the implementation of the development programs through five of the UAE leading educational institutions. It aims to make an impact by contributing to the educational system in the UAE, which went through fundamental improvements to support the National Agenda, the UAE Vision 2021 and the UAE Centennial 2071. For this purpose, this study will carry out an intensive investigation and will provide researchers in the field the opportunity to study and research for more in-depth issues within the area of gifts and talents in the UAE.

1.3 Research Questions

The study answers the following questions

1. What is the role of development programs implemented by governmental, educational institutions to develop gifted and talented students in the UAE?

Sub Questions:

- 2. How are gifted and talented students defined and identified in the UAE?
- 3 How is the development of gifted and talented students implemented in the UAE?
- 4. What are the main challenges facing the gifted and talented identification and development in the UAE?

Based on the findings, the research will seek to offer the decision-makers a set of recommendations regarding gifts and talents' identification and development in the country. It will participate in the efforts working for the improvement of gifted and talented developmental services in the country. Such a study focuses on supporting the UAE's Vision and National Agenda. The vision and the agenda aim to put the UAE amongst the advanced countries in the world within the Human Development Index. They also aim for the UAE to be the happiest among all nations and to make all citizens and residents feel proud they belong to the UAE. Education, skills training and inspiring innovation and creativity is accordingly the basic construction for the required development and the most excellent guaranteed investment in the youth and gifted and talented students in particular. This study explores the UAE's efforts and methods in nurturing and appreciating gifts and talents. It examines the development projects applied to achieve that, and the expectations of these projects in the future.

Finally, the research builds on data obtained by different methods and tailored in the form of literature review and case studies to present the efforts of the Educational Institutions' and to learn how they work to achieve such a goal.

1.4 Significance and Research Problem

This study is a relative comparison to equate the functions and operations of the leading educational institutions in the UAE, their practices, and policy implementation regarding gifted development, education and initiatives. Areas examined herein include the definition, process of identification, and gifted and talented development procedures. This study is born at a time of crucial need to discover and nurture gifted and talented individuals, not in the UAE, but also throughout the globe. The UAE is accordingly launching and funding more institutions to undertake the responsibility of talented identification and development. However, and pre the establishment of other institutions, there is a need to understand and examine the actual position of identification and development of the gifts and talents in the UAE and what their needs are.

The governmental concern and the confidence given to these institutions assume that enormous efforts made by them in respect of the development of this area and the improvements applied to boost the educational system.

The study, henceforth, intended to clear the view in this area regarding the identification and development of gifts and talents in the country. It looks forward to providing the extraordinary learners', the concerned leaders and educators, the experts, and decision-makers with clarifications regarding the situation of the identification and development programs and initiatives in the UAE. It also intends to endorse how to improve these practices according to some of the best similar performs in the world, if needed. It will also provide officials and decision-makers in the UAE with in-depth information and suggestions for any future improvements with recommendations. In short, if the UAE is counting on its gifted and talented program to support achieving its vision and mission in the coming years, it is worth looking at such programs closely.

1.5 Limitations and anticipated Challenges

The researcher is working at the Ministry of Culture and Knowledge Development (MCKD) and in charge of Department of Talents at the Ministry of Possibilities. The researcher confronted numerous challenges due to the following issues:

- (1) The topic of gifts and talents, their definition, identification and development is complicated and challenging, not only in the UAE but also around the globe.
- (2) (MCKD) as one of the examined institutions in this study, is where the researcher currently work.
- (3) The Department of Talents (DOT) at the Ministry of Possibilities (MOP) is conducting similar research to explore and examine the same situation.
- (4) This study will rely on document reviews, interviews and focus groups. It is not to criticise the institutions or their efforts in this area, but to share these institutions' their appreciated national efforts and to collaborate with their ambitions to achieve the UAE's vision and to support their future improvements by reliable researches.

This study, therefore, does not attempt to evaluate the effectiveness of the UAE educational institutions or how beneficial are their efforts in implementing various educational systems and their improvements in the UAE; instead, it aims to explore specific gifts and talents oriented initiatives practices, launched to identify and develop students' gifts and talents in the UAE. It also seeks to identify the used practices by these UAE Educational Institutions in respect of capable students' identification and development compared with the identification and development practices employed by other similar talents'-oriented initiative internationally. The aim of this study's construction has initially been to base its data collection primarily on previous studies and

researches in this area, responses from stakeholders who benefit from these programs, and data obtained from different resources. However, and due to some administration obstacles, the study depended on the retrieved information collected from preliminary data published mainly on their official websites or the government official websites besides the data generated from the research applied methods such as interviews and focus group. The study utilised the open data facility and the results of researches and interviews gained from the currently ongoing investigations in this area. Such sources are all presumed to be the most reliable, trusted, and up-to-date data that can be found about the gifted and talented issue in the UAE in addition to the matched destinations and aspirations towards the same results that will support this category in the UAE.

1.6 Organisation of Chapters

An overview of the area of gifts and talents in the United Arab Emirates, purpose, and rationale of the study, significance and research problems, limitations, and anticipated challenges were in this chapter, followed by Chapter 2 reviews the literature related to this study. Chapter 3, the institutions' case studies. Chapter 4 presents the methods and procedures used to create this study. Chapter 5 presents the results and data discussion, and Chapter 6 contains the findings, recommendations, and conclusion.

Chapter Two. Literature Review

Introduction

This chapter analyses the relevant literature for this study. For this literature pursuit, the utilised researches obtained from international countries are studies conducted by prominent scientists and theorists in this field. They are from nations like England, America, and the UAE. The chapter discusses definitions of gifts and talents and any associated issues. These issues include identification process, development programs, their need for special provision, and what it should involve. Therefore, and to understand what type of improvement these developmental programs require, the chapter refers to some older texts and use them as references, despite updating some of the models in them. This literature review aims to try to track the development in the field of gifts and talents, their identification, development programs and practices. It includes an assessment of related literature in the areas of global debates, definition of talent, definition evolution, identification processes, and development procedures, stages of talents development, assessing achievement, special education and gifts and talents development programs and presenting two experiences of two leading countries known for being prominent in the identification and development of gifts and talents (US & China experiences). A discussion regarding intelligence and talents, defining intelligence, nature of intelligence, intelligence tests, criticising intelligence view, and theories of intelligence are also present. This chapter discussed and criticised four of the pioneering theories of intelligence. These theories include The Triarchic Theory of intelligence; The Theory of Multiple Intelligence; Gagne's Differentiated Model of Giftedness and Three -Ring Conception of Giftedness. It indicates the employment of The Theory of Multiple Intelligence as well.

2.1 Global Gifted and Talented Debates

Scholars have always wanted to understand the meaning of giftedness, how to identify giftedness and talents, how to develop them, and to make other scholars and researchers explore, understand and learn more regarding them. Earlier theories and other empirical investigations have usually been dependable on previous works and researches. These studies appeared complementing or conflicting over conceptions from time to time. The heated topic in the conflicts was regarding the concept and the portion of the responsibility of each stakeholder towards explaining the conception and the development of this category as well. In a study by (Subtonik, Olszewski, Pand Worrell, 2011), for example, a set of interrelated arguments about talents was discussed. It has confirmed that the abilities do matter and that the different talent domains have different development routes. On the other hand, the study indicated that the society is responsible as well for supporting the development process of talents development. Gifted and talented individuals are also responsible for their abilities and skills development, and they share this responsibility with the community and other stakeholders as well. The leading goal is to allow gifted and talented individuals to reach an outstanding and high level of personal satisfaction and self-actualisation. They should produce several benefits to themselves and society. (Subtonik & Arnold, 1994) said that talented individuals need special education to thrive, and the mentors of the gifted students believe that even highly intelligent students benefit from exceptional educational opportunities, and they may also require them to satisfy and foster their potential.

Several studies were very famous in this area, like the popular debate in the history of psychology title "nature vs nature "by (Baker, 2012). This debate made it hard to decide how talented abilities development happen. The question was if it predisposes in the DNA, or influenced by the talent's life experiences and the surrounding environment? Several studies examined "the influence of

physical ability, mental and emotional aspects of the talents' development. (Rita, Curlos, 2015) Examined home influence, school, and culture affect and discussed an athlete's performance enhancement by several factors. The study results showed that many factors affect development, and they are essential. She considered things like supervising, mentoring and coaching are part of the athletes' development. The cognitive-behavioural therapy and attention to aspects of health psychology are included in her study results as well. Other factors, such as neuropsychological interventions are according to her part of this development. Another study by (Olszewski, P., Kulieke, M.and Buescher, T., 1987) indicated that many factors boost gifts and talents development, including family environment and parents' value as well. Many conducted types of research were to examine the causes of talents' achievements and the effect of the environment on their success, such as (Subtonick, 1988). Subotnick studied (94) American winners identified from a popular American science talent search. The longitudinal study that started from the time when those winners awarded in 1983 until 1993, investigated the factors that led those winners' achievements in science and math. She also studied the retention of this achievement. According to her mentors helped a lot in supporting those scientists and making their achievement sustainable. Mentors constituted an essential part of those scientists winning. (Subtonick, 1988) found that curiosity was a great motivation that led them to win. Exposing the classroom to scientific debates helped much in supporting their success. She also compared between males and females and found that both had different views of success. According to Subotnick, men's success refers to their intelligence and creativity, while women's success refers to their hard work and dedication. The study also indicated that the main factor that led those scientists to decline continuing their studies in these fields was because they believed that scientists' lives are not interesting. The lack of reasonable mentoring and the quality of education were also reasons for making these winners

decline science (Subotnik, 1994). By 1988, several winners dropped out of science. The number of (41% out of the 73) females and (12% out of the 57) males constituted the dropped out of science. The study indicated that those winners who remained in the science track had received the essential continuous support. They received several motivations to carry out research projects. The resources included tools to engage in researches. They also had the opportunity of receiving mentoring services by expert mentors, and they had the chance to learn a tacit knowledge they needed to continue. Such factors helped them succeed in science major and continue in scientific research. Other studies from different countries discussed talents, giftedness, and educational issues. Studies concerned about the development of gifts and talents are usually supporting their development and providing the area with valuable debates and arguments in this regard. This issue is the most important one on a large scale because the area of gifts and talents is in need for special resources including differentiated educational system, special teaching methods and tools that provide them with their needs and fit in with their higher abilities, interests, and intelligence. (Bloom, 1985) conducted a study in which he interviewed (120) gifted and talented outstanding individuals in America who were under (40) from different disciplines of giftedness and talents, including Olympic swimmers, research mathematicians, concert pianists, world-class tennis players, research neurologists, and sculptors. Bloom's research team picked the participants based on the most objective criteria they were able to devise after they consulted experts in each field. The selection was limited to people under their 40s to make sure they still remember how they were able to develop their talents and achieve their success. Bloom,1985 interviewed their parents, trainers, and coaches, too in some cases to get more information. Bloom, 1985 stated that the main point of his research was to confirm that the initial ability was not the thing that made these talented individuals achieve in the future but focusing on a continuous

practice for long years and getting help from others around them. Practice, in other words, was the reason for their success. "They achieved excellent competence in their field because they got help and had the opportunities to practice" (Bloom, 1985). Bloom's type of research would have a better impact if they were as longitude researches that would trace talents' early lives and how did they go through different obstacles plus getting training and support. Another research by (Feldman, 1993) was about a prodigy. The researcher viewed prodigy as a different form of talent viewed in a different focus (Feldman, 1993). Feldman interpreted prodigy as the high performance of a child who is not ten years old yet in a very demanding field of endeavour, and his performance is similar to the performance of a highly trained adult. Feldman (1993) suggests in his interpretation that the child's prodigy is distinct from any other kind of talent. For Feldman, the prodigy was the supporting factor that made the development process more obvious and apparent in children because they shine amongst their peers very easily. Feldman studied six talented boys in different disciplines of talent, such as music, math, writing, and chess playing. The findings of Feldman's study acknowledged the effect of three factors on these children's development: (1) Their families were capable of providing them with good resources for their education; (2) The sort of discovered talent in early years and (3) Motivation played an essential role in their lives (Feldman, 1993). (Stanley, 1971); (Lubinski& Benbow, 1996) as well conducted a planned 50-year longitudinal study that focused on young individuals who were talented in math and science. The study tried to understand the factors that affect talented people to become successful in math and science and how those factors helped them develop their talents. The longitudinal study that Stanley started in 1971 and continued by Lubinski and Benbow in 1996 found that students aged 12-13 succeeded because they had the opportunity to sit for (SAT) test at an early age. The researcher found them through a talent search that looked for students with high scores that top 3% on standardised tests

(Gross, 2006). The study that was long has traced the development of the academic, social and emotional factors of (20) Australians who were highly exceptional with an IQ of (160 and above) (Gross, 2006). The findings of Gross's study noted significant differences that appeared clearly in the participants' life satisfaction, social connections and relationships, and self-esteem. Most of those talented people, deeply accelerated by two years, had professional careers and satisfying social lives. The others who had equal abilities but accelerated by one year or were not allowed to accelerate reported lower levels of life and work and did not show satisfaction compared to the other group (Gross, 2006). The study referred to the development models of (Tannenbaum and Gagne); however, it did not follow these models in an obvious way. The study found that talented children with high abilities were not satisfied because they did not get their academic, social and emotional needs (Gross, 2006). Gross indicated that many factors such as family support, ability and motivation are the factors that contribute to supporting talents' development (Gross, 2006). (Feist, 2006) surveyed to study the nation's best scientists winners of "Westinghouse" competitions in the years (1965, 1975, 1985, and 1995). The study investigated their "career results, gender differences, and age of talent recognition" (Feist, 2006). The first survey question asked, "how they were able to recognise the first sign of talent in them, and when their interest in Science began". The survey also asked about "how their productivity changed over the years and whether their gender influenced their productivity" (Feist, 2006) Results of the survey indicated that both males and females were equal by the number of honours they received, and they had similar levels of education throughout university and postgraduate studies. Both males and females were interested in science in similar ages, though, the similarity changed after high school (Feist, 2006). A study by (Neu& Baum & Cooper. 2011) traced a winner of (Intel Science & Engineering Fair (Intel ISEF). It wanted to understand the factors that led him to achieve and how he was able

to handle the project, fix his issues, and what made him a winner(Neu& Baum & Cooper. 2011). The main issue detected in the student's story was the need to recognise special needs students who are talented in science. Students as well need a differentiated science curriculum and lab to help them study and apply what they study in books. It also focused on the collaboration of students who have the same interests and talents in order to support each other and win. This study, according to (Baum, 2011), has informed both professional development and instructional practice (Baum, 2011). Another study by (Subhi-Yamin, 2009) contained an overview of the Arab Gulf region's different cultures, each with its experience methods regarding the terminology evolution and the concepts of uniqueness. It presented several significant problems in the field of identification and the growth of talent and creativity. (Subhi-Yamin, 2009) discussed multiple issues and challenges faced by educators in both the Arabian Gulf and the Middle East. These challenges include the philosophy of gifted education, the identification of gifts and talents, their development programs, advanced educational technology, and the new digital functionality. The study found out that, the identification and development of gifts and talents would be more complex if this category definition and description is not appropriate and if the educational system is following traditional methods in teaching them (Subhi-Yamin, 2009). Furthermore, according to (Allana, 2010), while Dubai is well known for being prominent in every respective field such as financial services, education, and lifestyle, however, a crucial area in its education systems is yet to be explored and requires more attention. This area focuses on how gifts and talents appreciated and acknowledged in schools. (Allana, 2010) examined the assessments and provisions provided for the gifted children studying in private schools in Dubai. It concluded by indicating the urgent need for recognition, assessments, and several arrangements for the development and education of gifted and talented children in Dubai's private schools(Allana,

2010).(Mansour, 2011) in another study, conducted analytical research to investigate the Status of development programs provided for gifted and talented students in Cycle 2, in public schools in Abu Dhabi. The study examined the processes and procedures of these programs and investigated the challenges they faced and concluded that the main challenges were in how to improve their abilities and the programs used for their development (Mansour, 2011). The study that conducted a thorough examination and comparison among three centres for gifted and talented learners in (ADEZ) identified the standings of these centres and compared them with the situation of similar centres in other advanced countries. The study concluded the need to introduce effective programs to promote gifted learners' thinking skills. Teachers in these centres proposed a set of suggestions included in the findings of the study (Mansour, 2011). In another research (AlGhawi, 2017) focused on the issue of gifts and talents identification and development in the United Arab Emirates through the new education system named as (Ted education). The study interprets the education and development provided to the students who demonstrated some exceptional abilities in seven schools in Dubai - UAE. The study searched these development programs provided for this category in the study's participating schools and looked into how to improve the requirements of such education. The study followed the standards of the National Association for Gifted Children (NAGC) framework in gifted programs and adopted it as an outline. It also discussed the provision of education in Dubai as an advanced education within the last ten years. According to the researcher, there were several improvements required to amend the application of gifted development programs. The researcher emphasised the importance of training the school administrators and teachers and making them aware of the policies issued by the Ministry of Education. The researcher recommended improving the area of gifted students 'development in the UAE (AlGhawi, 2017). Researchers in UAE such as (Al-Jaghaiman, 2018) conducted a study

to learn about gifts and talents identification and methods of identification in the UAE. Al-Jaghaiman's study compared (HA) identification method with the identification methods adopted in other centres and institutions in the GCC countries. He then compared them all to the National Association of talented American children (NAGC) standards and methods. The study presented a set of recommendations to support the development of existing practices (Al-Jaghaiman, 2018). Another study by (Budair, 2018) was the first to examine (ADEK's) Abu Dhabi Education and Knowledge gifted education in the UAE. The study that examined a gifted pilot program implemented in three public schools in Cycle 1, 2 and 3 in Abu Dhabi, focused on examining the application and its efficiency and the accuracy and the outcomes of the program that affects several stakeholders. It also checked their acquiescence with the standards applied by (NAGC) (Budair, 2018). The findings of the study showed the existence of several deficiencies that appeared in the application of the program components. It recommended making intensive improvements in this program to assist the students, to whom this program aim in the UAE(Budair, 2018). The field study conducted by the (Emirates Center for Strategic Studies and Research, 2018 (ESRCS) to research the position of the gifts and talents in the United Arab Emirates highlighted the reality of giftedness in the UAE. The research focused on the idea that the interest of talented students in the UAE is part of the national agenda. It concluded that nurturing this category in the UAE is a priority in the country (ESRCS,2018). The study concentrated on the efforts of the educational institutions that sponsor the development of these exceptional students. The study also discussed a set of concepts and definitions that describe giftedness and the challenges faced by these individuals due to these definitions. It intended to form a comprehensive picture of the area of the study review. It also presented a set of theories and discussed several international experiences. These experiences involved a group of countries that ranked the first six in the index of global

creativity in 2016 (ESRCS,2018). Finally, the study highlighted the positive role of the educational institutions and the civil society to enhance the area of gifts and talents care and the UAE's vision for the future of talent, however, it stressed on the ambiguity in the scene and the necessity of more researches to clarify it (ESRCS,2018).

Consequently, the abovementioned researchers substantiated that highly able individuals should get exceptional services. These services start from providing them with a supportive infrastructure and a nurturing ecosystem that accepts their differentiation. This ecosystem should include the educational, cultural, physical, and social institutions and initiatives, policies, strategies, laws, tools, updated curriculums, new digital tools, media awareness, family awareness, and qualified human resources. The UAE accordingly launched numerous institutions and initiatives to identify and nurture additional gifts and talents within the entire country. However, the lack of studies and researches that help to give us an insight view and enlightens the decision-makers regarding these institutions' achievements makes it more challenging to decide what they accomplished and if they need any improvements.

Therefore, the previous reviews in the literature aimed to look at some international, Arabic, and local studies that investigated the same issue in different methods and approaches. Countries differed in their procedures to identify and develop gifted and talented individuals. Some of the studies focused on younger children; others looked at gifted and talented adults. Some studies limited their focus to the fields of science and math, while others studied the talented people who display creative skills in performing arts. What makes these studies important is that they searched different types of factors and matters in support of the development of students' gifts and talents. It was evident that some elements are outdated or interpreted as variables at a specific date or moment. Some are no longer beneficial, helpful, or even fit for gifted and talented people in this

century. However, what linked them all was focusing on the importance of providing gifted and talented at all times with differentiated services, mainly a definition that reflects their exceptional traits. Searching for some gifted and talented requires special identification tools, while others stand out by their performances. The deliberate practice is, on the other hand, a must to secure gifted and talented sustainable development and to motivate them. A collaboration among the educational institutions intends to help the gifted and talented students' progress and grow within a secured environment. The different opinions, views, and methods used to search for these factors and the discussions of their findings form guidance for this study and many types of research to come.

The present study may not differ from the previously mentioned studies or may resemble some of them or even replicate the topic and the approach of any other. Such similarity, if occurred, is due to a critical factor that the UAE lacks a considerable number of researches in all aspects of the field of education, gifted and talented and their development. The UAE works continuously on launching more initiatives and institutions to discover and nurture additional gifts and talents in the individuals who can infuse innovation and creativity with their extraordinary traits. However, no surveys or collimated studies clarify the techniques used to identify and nurture this category currently. Therefore, this makes studying developmental initiatives, factors that encouraged and motivated their abilities, the achievements of the educational institutions, essential support to the field researchers, highly able students, and similar institutions to launch in the future.

Finally, the issue of gifted and talented is a crucial topic in all aspects and regards. Their role is vital in the development of countries and nations and an essential part in the development of the economy of the nations. They also help the countries achieve their ambitions and fearlessly enter the modernised challenging global competition. The UAE's recent national agendas seek to

support a new industry dependable on creative industries. Such an aim needs to invest in all areas of gifts and talent, starting with the improvement of the quality of life of these individuals in the UAE and the globe. It is obvious now that innovation cannot thrive if there is a shortage in the knowledgeable, creative and have the foresight human resources, while the supportive environment is the second-largest crucial support that would enhance innovation and give it space to thrive. Consequently, this study that reviews the educational institutions, currently responsible for supporting the highly able students' well-being, their identification and development, will provide an additional unique feature of this research and will undoubtedly support and highlight an unexplored part in the area of development of gifted and talented students in the UAE.

2.3 Theoretical Framework

2.3.1 Definition of Talent

"Gifted and talented "terms are usually open to multiple interpretations. They have been problematic for every stakeholder related to gifts and talents or deals with them(Al Obaidly,2006). Stakeholders in this study include the teachers, theorists, scientists, scholars, and experts in this area. This term describes persons who display exceptional performance and high levels of competence in different topics and domains such as scientific, academic, and culturally valued domains or socially useful forms of expression. (Encyclopedia of Education.com, 2019). Notwithstanding, the ambiguity of the concepts of gifts and talents resulted in continuous debates and confusion among researchers. (Howe & Davidson & Sloboda, 1998) For example, pointed at certain issues that scientists should avoid when defining talents. These issues are: (1) Not limiting the definition to a certain skill and (2) Not using a definition that needs a shred of evidence to prove the reason for using it(Howe & Davidson & Sloboda, 1998).

Theories of intelligence, on the other hand, focused on searching the origin of gifts and talents. Their debates concentrated on the origin more than anything else did. Theorists were always in conflict and debates to understand if the gift or talent is inherited or nurtured. Many theories that were invented by prominent scholars and theorists such as (Sternberg, 2004; Renzulli, 2005; Gagne, 2005) have presented the views of their inventors on what constitutes (giftedness and talent). Accordingly, this chapter pursues a range of views and opinions that argued and discussed what constitutes a (gift or a talent), starting with the evolution of the definition, the link between the definition, the identification type and the identification tool to the gifted development Programs tailored for the sake of their growth.

2.3.2 Evolution of the definition

The interest in these exceptional individuals raised by a set of ideas generated by researchers and theorists such as (Renzulli, 1978; Cohn, 1981; Gardner, 1983; Sternberg, 1985) and others. In scientific literature, notwithstanding the efforts exerted by scientists and theorists in talented and gifted issues. These efforts continue to face some out tackled and not yet agreed upon problems that keep ongoing flaming debates among researchers, scholars and educators. One of the significant conflicts regarding talents and giftedness is about deciding who is gifted and who is talented. This conflict encouraged scientists and scholars to exert further efforts and try to formulate an acceptable definition that would describe talents and explain the differences between the gifts and talents as well.

(Gagne, 1985), consequently, examined several definitions that were common in the field. Gagne examined (Renzulli, 1979 and Cohn, 1981) definitions. Gagne's research resulted in a reflection on the ambiguity found in the differences that occur in the concepts of giftedness and talents, which

accordingly, made him believe that there is a clear differentiation between both terms (Gagne, 1985). Talents, according to Gagne, connect with areas of abilities that raise the performance in different domains of actions, while giftedness relates to academic success. Latterly, the ambiguity of the concept of talent urged more specialists to try to interpret talent according to the differentiated skills of talented individuals and their different locations. Educators in schools, for example, wanted the term to describe the academically talented students; parents thought it when the child starts talking or crawling, the society interprets talents according to their popularity and fame. Therefore, such variations imposed more ambiguity on the definition of talents, which is supposed to define individuals with extraordinary special skills other than their academic achievements. (Howe & Davidson & Sloboda, 1998) Pointed at certain problems that we must avoid when it comes to defining talents like not limiting the definition to a certain skill, besides not using a definition that needs evidence to prove the reason for using it and to justify it (Howe & Davidson & Sloboda, 1998). Theories of intelligence as well focused on the origin of talents, and their debates primarily concentrated on trying to know whether a person is born talented, or this person talents should find appropriate development to cultivate and thrive. This confusing relation between intelligence and talents developed and advanced in modern theories to prevent the deficit and inadequate recognition of the talent. These theories intended as well to discontinue the confusion that appears while trying to determine whom to define as talented and who is not. Thus, the definitions of talented and giftedness continue to undergo changes and modifications in their concepts and explanation. Every while, the area of gifts and talents witnesses additional attempts of many scientists and educators who are trying to redefine the terms gifted and talented. Several scientists exerted efforts in this aim differently, like (Dr Dai, 2010) who suggested a new context for giftedness and gifted education. He accordingly asked three fundamental questions to

measure our knowledge regarding the role of natural abilities and the experiences in making gifts and talents thrive. Dr Dai's questions were like, "What we know about the role of the personal efforts in talents' development?" "How do we identify them?" "How are the objectives of gifted education achieved?" (Dai, 2010). Researchers then proposed a new definition intended to apply in all domains of giftedness (Subotnick & Olszewski & Worrell, 2011). They suggested that giftedness involves a high-capacity individual who has achievements among other high-capacity people in the same field of talent (Subotnick & Olszewski & Worrell, 2011). They also suggested that achievement is the measurement of giftedness (Subotnick & Olszewski & Worrell, 2011). This definition would be sufficient if it describes the importance of evaluation and if it explains how to assess the high ability and how to recognise it. The second major conflict is about not having a policy that dictates definitions for those gifted and talented individuals (AlObaidly, 2006). Several educators urged the need to define the talented and gifted students to be able to educate them in special schools by distinctive curriculums and exceptional methods and tools (Heller, 2000). Then again, many other scholars said that these exceptional students should not be separated from others, but should merge with common students to benefit the larger number and to allow others to have confidence and exhibit their hidden talents in an encouraging environment. Several questions raised to enquire about the reason for not having a unified definition for gifted and talented while having a definition that nurtures and supports children with disabilities (Shaunnessy, 2003). These questions might find an answer according to the nature of each case, but it is worthy of tracing the history of the evolution of the definition to know how it transformed over time.

Hence, the next section presents the evolution of the definition introduced by several scholars throughout history.

2.3.3 History of Definition Evolution

In this section, a list of definitions demonstrates the definitions' use by several theorists in the field of talents and mirrors the evolution of this term based on time and time variables. However, the list does not exhibit a unified definition that everyone approves of using it.

The term (Talent) in the sixties of this century defined people with special abilities (Dodin & Jarwan, 2018). Some theorists considered talent as an ability that is a synonym for giftedness related to the academic field and limited within certain ages; others interpreted it as the distinguished performance in arts, sports, various craft areas, mechanical skills, and social leadership that exists in any age and at any time.

- (NAGC) Defined talents as synonymous with giftedness. It is according to them what exhibits in one or more areas such as intellectual, creative, artistic, leadership, or displays in a specific academic field like language arts, mathematics or science."
- Lewis Terman (1916) definition of talents is the ability to think abstractly.
- The American definition that was adopted by the federal legislation for talented individuals in the United States in 1971 defined talented or privileged children as Those who are recognised by professionals, and they show a high level of distinct performance, and they are more qualified than the regular school curriculum. School curriculum, therefore, does not meet their needs (Heller, Monks, Passow 1993, Kirk & Gallagher 2011, Sisk 2010, Novello, 2009).
- Renzulli (1977) explained giftedness as the communication between three groups of human characteristics. These consist of higher levels of general abilities, motivation, and creativity.

- The theory of (MI) Multiple Intelligences that was developed in (1983) by Howard Gardner. It identified nine types of intelligence. These are 1) Logical-mathematical, 2) Spatial, 3) bodily-kinesthetic, 4) existential, 5) interpersonal, 6) intrapersonal, 7) musical, 8) linguistic, and 9) naturalistic. Gardner related those different abilities and skills to moral intelligence. Gardner added moral intelligence as one of the nine intelligence.
- Beginning from the mid of the twentieth century, scientists began to evaluate intelligence differently, and this led to several changes in this field. (Daglioglu &Suveren, 2013).
- In the mid-1990, the terms "talented" and "gifted" appeared, but then again, both terms were synonyms. They used to replace each other and not to identify different abilities.
- (Morelock, 1996) tried to clarify the confusion by categorising the term talent and define it as the unconnected specialised abilities with the general intelligence and giftedness.
- The federal definition in the U.S exposes to additions, deletions, and impacts of other explanations.
- According to the International Encyclopedia of Education, the critical problems in the
 definition of talents are: (1) not distinguishing talents from giftedness; (2) several studies
 focused only on schoolchildren who excel in school subjects; (3) Combining giftedness
 with talents believing that talent is only a synonym of giftedness. (Husen & Postlethwaite,
 1994)
- (Freeman,2000) said measuring gifts could be as intellectual features of development but talents cannot be measured because they are discovered and identified by individuals who should be experts in this profession.

- (McKinsey report,2001) related the concept of talent to the size of a persons' abilities and gifts, knowledge, experience, intelligence, skills, attitude, character, judgment, and drive in addition to the persons' ability to learn and grow in work.
- (Munro, 2002) Differentiated between giftedness and talents by indicating that gifted students show their giftedness by displaying exceptional understanding and abilities.
 Talented students, on the other hand, are those who possess extraordinary skills in other domains that they need to learn explicitly.
- In (The No Child Left Behind legalisation, 2002), students with high achievement capability and who exhibit this ability in intellectual areas and capacities such as creating art, or leadership or specific academic fields as gifted. Moreover, they need differentiated services and activities to develop their competences.
- (Colom, Karama, Jung, Haier, 2010) said it is a general mental ability for reasoning, problem-solving, and learning.
- The definition adopted in (HA) in the UAE is "Talented students are those who possess extraordinary abilities and show distinguished performances that exceed the performance of their colleagues in the intellectual skills, academic achievement and or leadership as evaluated by the local community.
- According to (Stephens& Kames, 2000), several definitions that describe gifts and talents seem to be there in the UAE, as every institution has developed its definition.
- (MOE) Description of the highly able students emerges from a combination of international definitions that describe the concept of giftedness and the UAE visions, the labour market needs, and the national agenda (moe.gov.ae).

Finally, the issue of the multiplicity of definitions leaves teachers and parents confused and unable to interpret the notion of talent and work on identifying suitable individuals to design unique growth programs for them. This confusion permitted several schools, universities, and talent development initiatives to interpret the meaning of talents differently and to develop their gifted programs in different ways. This difference might be reasonable to satisfy the needs of these extraordinary students for unique and unusual methods to learn; however, dealing with such disagreement creates more difficulties, especially when it comes to designing the suitable development programs that will be able to grant them the required progression. The central conflict in this subject revolved around the question about what happens if there is no appropriate identification of the skilled people and what if this inappropriate identification places these gifted or talented in different development programs that do not meet their skills(AlObaidly,2006).In fact, not having a proper definition will result in an inappropriate identification. Therefore, the gifted or the talented students will end struggling with others viewing them as underachievers and bullying them at times.

2.4 Identification of Gifts & Talents

The educators' voices that were demanding to apply different methods of identification and development for this category come to be louder. They said that these methods could be in the form of acceleration and allocation of some classes or even entire schools. It resulted in a complete shift from the beginning of 1990 until today. The 20th century accordingly witnessed the establishment of several gifted and talent development centres, activities, and conferences. Many training programs applied to teachers and more researches conducted in this field. This interest increased significantly in the 21st century. It changed the way societies view talented people and altered the significance of literacy and skilled programs (Siegle, 2004). However, the arguments

related to identification and development have been on the table since the nineteenth century and are still an ongoing debate until today(VanTassel-Baska, 2000). Such a discussion is causing a fading in talent development programs.

Consequently, such dispute reduces the services endeavoured for these children and adults with gifts and talents as seen by researchers like (VanTassel-Baska, 2000). It is not difficult to guess that identification, which refers to the action or process of gifts and talents identification, is the most critical stage in this area, according to (Oxford dictionaries,2019). It arranges the scene before implementing any development program. The fact that talented individuals need special, challenging programs, high-powered complex curriculums and flexibility in educational regulations, is what makes the identification process the first step towards success or failure. It is the first step towards the talents' discovery and development (Feldhusen, 1996).

The identification of these highly able individuals is progressively relevant across several different domains of talent (Abbott, Button, Pepping & Collins, 2005). Also, this area needs more attention because it predicts the future performance of the identified person. Upon evaluation of current physical, technical, and psychological qualities, identification should be able to find who is gifted and who is talented. It must not confuse between gifted and talented people who need services and people who do not need them(VanTassel-Baska, 2000). The talent Identification process has been going through a dilemma, and it faced many challenges that made it more complicated. The following points summarise the challenges in identification:

• The way people understand the concept of absolute talent versus relative notions (VanTassel-Baska, 2000).

- Most of the gifted and talented professionals regard talent as a stable characteristic belonging to an individual, while many researchers in the sector have recommended not to consider talent as a simple notion unrelated to anything else (Renzulli, 1980).
- How to identify talented individuals' differences? (VanTassel-Baska, 2000).
- Over-reliance on intelligence tests. The intelligence test is limited to precise measurements that exclude creativity for not being able to measure it (Perkins, 1995).
- IQ test as an identification practice does not reflect the understanding of talent (Coleman, 2003)
- Narrow conceptions of intelligence (Perkins, 1995).
- Confusions appear in defining between talents and giftedness. (Tuttle, 1988).

The identification process has always fueled the concerns of educators regarding how teachers, parents, or any related stakeholders identify gifted and talented individuals? How do they know if this identification is appropriate? The addressed issues related to the process and tools of identification emphasise the importance of reviewing and evaluating their ability and reliability. Described, as better tools for identification are those that help correctly identify who is gifted or who is talented, who requires services, and who does not need any special differentiated services (VanTassel-Baska, 2000). Furthermore, the appropriate identification provides equity for talented individuals when it comes to selecting and enrolling them in development programs or advanced classes (Borland &Wright, 1994).

Subsequently and to accurately identify gifted and talented individuals, according to scholars, the identifying programs must first understand the definitions of talents and giftedness (Hodge, 2013). When schools and talent development initiatives can effectively identify these individuals, they must be able to locate the school's domain of giftedness, describe the students' level of giftedness,

and the fields of talent. (Hodge, 2013). Once an individual identifies as gifted or talented, schools or other concerned organisations should know how to use appropriate developmental, educational strategies and plans that cater to them. (Hodge, 2013). Therefore, and once again, early identification of talented and gifted individuals is an essential mechanism because it provides a way to the full realisation of their potentials. According to (NAGC), no one program fits all-talented and gifted people. However, there are similar characteristics among talented and gifted individuals(Matthews, M.S. and Shaunessy, E.,2010). These differences help to realise gifted and talented students' needs and nurture them upon that.

The main issues needed to consider, according to (NAGC) are the following:

(1) Giftedness is not static, and it is changeable. (2) Identification of giftedness or talent must repeat. (3) One test at a time is not able to distinguish the talents of the individuals and should not say whether someone identifies as talented or gifted. (4) talents and gifts appear through all types, nationalities, races, ethnics, social levels, and diversity groups all over society. (5) Early identification improves talents and giftedness (6) Talents, and giftedness should be identified through different ways and tools as giftedness and talents may exhibit within a specific interest or category. (NAGC, 2019).

2.5 Development of Gifts & Talents

The term "development" used in this study relates to growth that leads to a significant result and describes what talented individuals undergo like training, processes, practising, professional development and evaluation. Development of gifted and talents is what individuals with high abilities and their teachers or trainers require to flourish and to sharpen their skills. They are such as Professional training and talented programs conducted by experts who know the appropriate

methods to serve talents potentials and to motivate them. Well-designed development programs will positively influence the talents of those who are exceptionally able (Lubinski & Webb & Morelock & Benbow, 2001). Scientists and scholars expertise widely agreed that the development of talents has a massive impact on talented individuals and helps them maintain their skills over time and to stay connected to the creative and productive work within and after the program (Matthews, M.S. and Shaunessy, E., 2010).

The study by (Park & Lubinski & Benbow, 2007) tracked (2409) talented adolescents assessed on the (SAT) at the age of (13) and received particular services for more than 25 years, and found that these participants have excellently progressed in their middle age, and their achievements were amazingly recorded (Park & Lubinski & Benbow, 2007). One of this study Participants, for example, got the medal in the field of mathematics, while another participant won (John Bates Clark Medal, encyclopeiedia2.thefreedictionary.com), which honours the most outstanding economist under 40. Among the rest of the participants, (817) participants were able to earn patents and to publish (93) books. Thus, the study by (Park & Lubinski & Benbow, 2007) found that talented individuals must undergo a growth journey to achieve their potentials. Talent development, according to (Csikszentmihalyi, 1996), requires more non-intellectual factors that exceed talented individuals' abilities, like motivation, personality, determination, and attentiveness.

On the other hand, (Neihart, 2002) suggested that talented and gifted individuals possess higher motivation and a stronger drive, accompanied by lower anxiety than other individuals (Neihart, 2002). (Greene, 2006) Stated that while talented individuals demonstrate superior performance, however, counsellors and trainers must be aware of those talented individuals' needs of a unique career and life development issues. These factors might hinder their plans and prevent them from

developing their abilities if they are neglected (Greene,2006). Greene furthermore indicated that these issues could be in "having more than one talent, the early emergence of talent and popularity, personality traits, the overemphasis on academics, and the higher expectations of others."

2.5.1 Stages of Talent Development

Depending on (Gagne, 1991) "basic giftedness," and (Gardner, 1983) nine intelligence, children's talents in early years can be viewed as incomplete and similar. However, the variations in environments at home, school, community, and other surroundings or experiences related to these children impose differentiated situations on every child. Such differentiation makes the child establish more talents, strengths and skills. On the contrary, and in case these circumstances were not supportive nor appropriate, they are expected to hinder the child's talent or weaken it. (Feldhusen, 1996).

According to the abovesaid, a few children could produce creative things in different domains of creative talents and continue their breakthrough successfully. Those innovators are often talented individuals who were inspired and motivated by their families or teachers or other people in their surroundings. They are also the lucky ones who were able to experience early development in their lives. (Feldhusen, 1996).

Moreover, several developmental theories highlighted the nature of the gifts and indicated that they are changeable. They as well indicated that different external factors might interact with the internal factors to help the student produce gifted or talented behaviour (Pfeiffer, 2008).

Consequently, this chapter will include a look at issues of talents' identification and development from a cross-cultural perspective presented in the final section of this chapter.

Table2. Developmental Theories

(Mönks, F.J., 1992) (Multifactor) model of giftedness.											
Mönks modified Renzulli's three-ring model. Mönks added the environmental factors including school, family,											
and peers, to the three factors such as motivation, creativity, and exceptional abilities. Mönks came up with the											
(Multifactor) model of giftedness.											
(Feldman, 1999) Seven dimensions of development											
Cognitive processes			Family aspects		Education and preparation		Characteristics of the domain and field		Socia cultur context issue	al tual	Historical forces, events, and trends
4-10 years Cognitive control through exploration and observation.											
10-13 years Development			oment throug	t through mentors, models, contests, apprenticeships.							
13-18 years Commitment			tment to deve	t to development, idealism, blending self with talent.							
180-22 years	180-22 years The crystallisation of talent with a career choice.										
Francois Gagne (2005) Differentiated Model of Gifted and Talented (DMGT)											
Enforces the important			Non-ii	Non-intellective variables				To transform essential genetically			
environmental influences: Home,				(Motivation and temperament),				determined gifts (intellectual,			
school, parents, activities and			learnii	learning, training, and practising				creative, sensorimotor.) into talents			
encounters.		(language, arts, science,									
Abraham Tannenbaum (1986)											
Five psychological and social linkages concerning promise and fulfilment:											
Superior Exceptional s		nal special		on-intellective		Environmental		Chance or luck			
intelligence aptitude			facilitators influences								

2.6 Assessing Achievement

Assessment of the gifted and talented achievement is a vital part of their identification and development process. It helps ensure that the objectives contribute to the success of the process. The aim of the assessment should be "to educate and improve student performance and not to audit it." (Wiggins, 1998) Assessment can be in different types and forms, such as:

Ability and achievement tests: These tests provide scores to designate the performance. The tests should align with the students 'characteristics. Students' traits may require more than one assessment to distinguish them. Tests are standard assessment tools for identification, and they provide an objective way of identifying gifted children. However, tests are not supposed to be the

only source of identification, according to (NAGC, 2019). These tests become tools to benchmark requirements for enrolment into specific programs. The efficacy of the assessment, according to (Wiggins, 1998), relies on the assessment's relationship with the curriculum as well as how it represents it.

Contests and Competitions: Individuals competitions motivate talented individuals and build on critical interpersonal skills that help prepare people for success and boost their self-esteem and engagement.

2.6.1 Gifted education & Development Programs

Gifted Education includes a variety of programs and processes such as gifted and Talented Education (GATE), gifted and talented programs (TAG). These include a broad group of distinctive training, courses, activities, practices, procedures, and theories used in the education of these exceptional children. Most countries nowadays adopt policies to nurture them and provide them with services as an essential requirement in their development. This section will brief two of the highest advanced countries in the field of Gifted and Talented development. Their experiences may help the readers to understand what supported their advancement and how did they do it.

2.6.2 The US Experience

The United States' experience in (GATE) and (TAG) is a leading experience and one of the prominent ones. It was evident in terms of many programs that support this category and having a federal law to protect and serve their potentials. The long history and the advancements in education and psychology in US-led to credibility and trust to the field too. Moreover, the existence of multiplicity in methods used in implementing gifted education and programs, as well as many

institutions, organisations, and associations that sponsor this category makes the US as one of the crucial experiences in this field. The increasing interest in talents made people recognise their abilities or other related individual's skills.

Furthermore, this resulted from the development in the field since the beginning of the seventies of the twentieth century. Today, the US educational systems include a method for gifted and gifted education. This system comprises assembling, enrichment, and educational acceleration of early admission to kindergartens and primary classrooms. Another system is classroom or curriculum compacting that helps those gifted and talented pass the school systems in less than average years.

This history of care for this category in the US confirms that this type of care has not been formal as it is today, nor the tremendous momentum of attention it currently enjoys. The beginnings were individual experiences. Scientists tried to apply for some accelerator-based programs in 1886 while gifted students acquired a specialised curriculum and additional enriching teaching experience after their graduation in the city of St. Barbara, California, in 1898, as well as in New York in 1900, the city of Cleveland and Ohio, in 1920. This method of unofficial concern was due to the fearing of giving extra interest and care to the gifted and talented individuals, which would make them feel they are (elites). Such a thing would create a grouping of individuals who will line up together. The other reason is the belief that talented individuals could manage and develop individually without the need for intervention. These beliefs hindered the efforts to support this group; therefore, gifted and talented differentiated care was infrequent during the period 1925-1950.

In general, the studies of Lewis Terman and Lita Hollingworth were unique and distinctive in the history of psychology. The field of giftedness was one of the oldest systematic scientific reviews that have helped draw the attention of educators, researchers, and public opinion in the US to the subject of gifts and talents, their physical, mental, emotional, and social characteristics. The educators and scientists gave their attention to the different orientation as well, the importance of their early detection, educational and psychological care, and special education programs for the growth of their giftedness and talents. These studies looked at the lives of academic and professional graduates, their problems, some aspects of their mental health, and their subsequent achievements in their various potentials and specialities. It also paved the way for further studies to reveal the personal characteristics and non-cognitive factors that contribute to increasing educational attainment and academic achievement, to explore the conditions and family and school environments suitable for talent growth and excellence, and to assess the impact of strategies and teaching methods on the development of talented children.

The attempt by (Terman, 1916) to translate and codify an English-language test in the United States published for the first time in 1916, is a vital milestone in the area of giftedness in the US and its history as well. Terman conducted his longitudinal study of more than 1,500 gifted students in various stages of research in California and published the first report in 1925. These results and subsequent reports have had a substantial impact on drawing gifted care, especially about their characteristics and problems. The experiments of (Lita Hollingworth, 1923) supported the attempts of Terman's as well. The results of her studies and the information contained in her books on the education and nurturing of highly educated talents drew attention to the importance of educational enrichment. Hollingworth's studies helped to study subjects, identify the most crucial scholastic, emotional, and socio-psychological problems of gifted students, problems resulting from their feelings of distress and boredom in ordinary schools, and their unsatisfied relationship with their peers and teachers. Hollingworth's studies have also shown the importance of psychosocial

counselling in solving most of the problems of this category. Because of such attempts and scientific studies, Westinghouse Company adopted in 1942 a project to discover talented secondary school students who can be creative scientists in the future. This project and its accompanying efforts led to exhibiting the value of exceptional students in the field of science. It drew US attention again to the importance of talents in their industries.

In 1947, the establishment of the American Association for Gifted Children (AAGC) followed in 1953 by the establishment of the (NAGC) that published "The Gifted Child Quarterly." (NAGC, 2019). All these events were signs of a limited orientation towards the care of the gifted. However, the most memorable event in the United States was the success of the Soviet Union - Russia - in launching the first Sputnik satellite in the fall of 1957. Since then, interest in the United States has grown in nurturing gifted and talented students with exceptional services and education as well. They started thinking about the ways to maximise their abilities and to ensure scientific superiority over the Russians. Such thoughts resulted in (1954) by establishing the National Association of Gifted Children (NAGC), Passing the National Défense Education Act (1958), (the Civil Rights Act passes. It emphasised the equal opportunities of this category, including those in education (1964). The Americans then achieved their fame by invading space and landing the first astronaut on the moon in 1969.

The federal government then funded various projects to improve educational programs to develop students' talents and preparations in science, mathematics, and foreign languages, and to develop mentoring programs. In 1970, Congress mandated the Commissioner of Education, (Sidney Maryland under Act PL 91-230), to conduct an extensive study that investigates the status of their abled children and the adequacy of the federal education assistance programs in meeting their needs and developing their skills. The report published in 1972 highlighted the following points:

- 1. The need to develop unique approaches that highlight the processes of higher thinking, innovative thinking, critical thinking, and high-level concepts.
- 2. The need to develop appropriate teaching strategies that will be able to teach this category and help them find successful solutions to the problems facing society and the nation.
- 3. The need to develop educational services systems and adequate administrative procedures to provide distinct services to groups of gifted individuals, such as individuals with unique talents. (Davis & Rimm, 2004)

The report also produced a broad and comprehensive definition adopted as the first federal definition to include multiple categories.

The Office of the Gifted and talented got its official statues in (1974); Congress passes the Act (Jacob Javits cation Act) in (1988). The issued report title "Nation at Risk" in 1983 prompted officials to accelerate the process of educational reforming. In (1993), The Case of Developing America's Talent issued by the US Department of Education defined how America abandonments the most talented youth. (NAGC) Published Pre-K -12 Gifted Program Standards in (1998). The Act of (No Child Left behind (NCLB) in (2002). A Nation Deceived report on acceleration strategies for advanced learners published in (2004). (NAGC) published (National Gifted Education Standards) (2006).

The scientific research has been active in this area when some academics rushed to research into the fields of nurturing talents and creativity, initiated by Gilford in the early fifties and then Paul Torrance and Calvin Taylor, James Gallagher, Julian Stanley, and Joseph Renzulli, The contributions of these researchers and others have a significant impact in the field of talents.

(Gallegher, 1994). These contributions urged a culture of talent's development through institutional action strategies at the local and international levels. Also, there was a movement at the family level, promoting the need to protect their gifted children whom their teachers and schools frustrated them by their negligence. The movement also promoted the importance of providing the teachers of highly able students with suitable training. The training was to increase these teachers' knowledge and qualify them to be able to support these students' progress. The National Foundation for the Gifted and Creative Children came then to defend and protect the rights of gifted and creative people. The gifted education area remains evolving since then to meet the varying needs of the country in a challenging varying globe.

Forms of development programs in the US:

- (1) The Government provides flexible support to educational departments and schools. The disbursement of grants from schools related to the implementation of special programs and various additional activities that benefit students, including those with special needs and talents.
- (2) Diversity of applied methods.
- (3) Having many specialists, researchers, and researches in the field.
- (4) Writing, publishing, and usage of experimental research methods increased.
- (5) Schools provide services for able students, such as the provision of additional activities and decisions as personal arrangements and initiatives.
- (6) Most high schools in the US offer elective courses accepted as accredited modules for students to cover the course requirements. These courses often focus on teaching foreign languages and physical education, and the provision of such courses depends mostly on the budget and physical status of the school.

Moreover, not all secondary schools in the United States are following the same compulsory system; other systems are created to satisfy the different needs of different categories in the US, such as:

- There are preparatory and secondary schools known as "Honor Classes" for creative students.
- Some of these schools apply a system of tests to determine the level, abilities, and tendencies to sort and accept certain types of high achievers.
- Some schools offer enriching experiences in various arts and crafts.
- In the case of adequate financial support, some schools offer courses in International Baccalaureate (IB) and Advanced Placement (AP) to support early university graduation.
- In some states, there are different community colleges. These colleges allow talented students at the secondary level to enrol in certain subjects in these colleges. These subjects accredited to them after they join the university, and thus, they can graduate early.
- Other alternatives for education are Early Admission in KG to colleges and universities,
 Grade Skipping, compacting, acceleration, correspondence courses, credit and skipping by
 examinations.

2.6.3 The Experience of China

In Taiwan, the word "talented" includes academically gifted and talented individuals and children with outstanding abilities in fine arts, music, dance, and sports. The year (1978) witnessed the establishment of Central schools to provide their exceptional students with special differentiated education. These schools are the schools established within the People's Republic of China, which

combines theoretical study with ordinary daily work and assigned their graduates of national cadres to relevant jobs to serve China. A law passed for these students in (1984), including the establishment of programs for academically capable students in all fields. An expanded program to develop these talents incepted and worked under the supervision of the leaders to adopt the Chinese culture with a general view of the Western culture. China's educational system is one of the oldest systems in the world, yet the modern forms of education from various western educational systems influenced it, and they copied many of these systems during the twentieth century. (Zhang, 2017).

The influence of the west on the Chinese educational testing system divided into two historical periods. (1905-1949) Post, (1980). However, the interest in these bright children in China dates back to ancient history (580 to 1905) when the famous Chinese script system was so prevalent that it made these children and young people the most important national treasure, and they held the hopes of the future. (Zhang, 2017)

Consequently, gifted children in China sit for examinations held in government palaces. Upon passing these special examinations, the gifted and talented students receive a study grant, which qualifies them after graduating to become senior government officials or qualifies them for a job in an advisory position.

The Chinese experience in the field of caring for gifted people was different and unique. The state of China incepted the Palaces of Culture with some reputable and professional institutions to nurture the talented people in the country. Those palaces provided all types of gifted, talented with the appropriate education and training. They offered this education to various groups of individuals, including children, youth, workers, peasants, and teachers, with all educational and appropriate services and programs in the evening. These cultural evening palaces contained

educational services and provided adequate care for talents in a systematically professional way. The attention given to talented and gifted individuals in China focuses on nurturing leaders in the domains of science, literary, technical, mechanical, physical education, and other areas of talents. The followed system in these palaces of culture is the method of grouping gifted individuals who show a tendency to excel in a domain or have the willingness to enrich their knowledge and develop their talents and abilities. The formal education system, however, provides all citizens with regular education without distinction between different abilities or groups of talents. Consequently, the ablest students are subject to the ability of their teachers who can identify them taking into consideration the individual differences of the students in order to identify them appropriately (Li & Delisle, 1990).

The Chinese government believes that education is the means of achieving the scientific and technical development needed by the People's Republic of China in a short period. Therefore, Chinese educational policy has focused on the quality of education at various stages. Accordingly, they developed newly updated programs to satisfy the needs of these individuals and equipped schools with all necessary modern educational tools and resources, as well as the development of the methods of preparing and qualifying teachers in universities. Thus, in a short period, China was able to make its significant progress when it was able to gain the time it lost during the Cultural Revolution, which resulted in the loss of many knowledge and scientific facts essential and necessary for its scientific progress.

In China, academic achievement tests have always been the main emphasis of the country since ancient times. They consider academic achievement as an appropriate tool to help them identify and distinguish those with gifts and talents.

China encourages and motivates gifted and talented students through different methods and structures. It applies a variety of ways, including classroom methods and non-classroom methods. Some of these methods mean establishing special classes for talented and gifted students in different educational stages. Early access to primary school helps to nurture those gifted children who can admit schools. Other ways are like: Implementation of scientific competitions in various educational stages; Early access to universities; Establishment of experimental and technical classes for youth that directly follow universities; Creating pilot classes with educational programs for talented children; Non-class study groups; The quality of the lesson. Chinese psychologists and educators consider the quality of the lesson more than anything else does because they see that it has a significant role in nurturing and encouraging talented students. These lessons identify and motivate this category. Moreover, and in order to raise the level of the lesson's activities, they consider the fundamental educational principles in the following:

- The role and the growth of the talented child, the complexity of the teaching tasks, the variety of illustrative examples, and the diversification of the positions of the assessments and test and others.
- Encouraging quick thinking, stimulating productive thinking, and the ability to articulate and discuss the problem by promoting direct dialogue and self-initiative of students and encouraging students to deal boldly with the contents of the lesson material.
- To provide diverse and in-depth knowledge, information, and mastery of necessary skills as a prerequisite for innovative achievement.
- To raise logical and creative thinking through ideas. Diversifying ideas and providing qualitative illustrations and encouragement through guidance and assistance.

 Tasks and duties assigned to the talented students to stimulate their thinking and to commensurate with their mental development that should be in line with their interests and tendencies.

Thus, China's experience shows that, despite their loss of knowledge and scientific facts necessary for their scientific progress due to the Cultural Revolution, China continues to focus on the quality of education at various levels and seeks to develop qualified programs to nurture the gifted and provide them with all services including materials and human resources. China achieved its fantastic progress in a short period because it stressed the quality of education and development programs.

2.7 Intelligence and talents

"Talent hits a target no one else can hit; Genius hits a target no one else can see." Arthur Schopenhauer.

Talents are special abilities of developmental origin. Intelligence is a mental quality. It consists of abilities to learn, adapt, understand and use knowledge to control the environment (Hilliard, A.G., 1994); (Pinker, S., 2010. Talent definition differs according to who and what it describes. In the world of management, talent is the naturally recurring patterns of thought, feeling and behaviour (www.Gallup.com.Retrieved). The managerial talent is a combination of leadership, strategies, emotional maturity, skills and abilities (Orleman, P. and Borner, H., 2008). In schools, the definition differs and changes according to the educational system and the philosophy of the country. "Children whose ability is significantly above the norm for their age are gifted." (www.nagc.gov)

According to some dictionaries, intelligence is the capacity of learning, reasoning, understanding and aptitude. It is in grasping truths, knowledge, relationships, facts, meaning and the manifestation of high mental capacity(www.dictionaries.com).

Consequently, and according to several scientists, talent and intelligence are different and detached from each other. They have no link because if a person is intelligent, this does not mean he is talented. Besides, some talented individuals may exist among the mentally disabled.

Thus, the term talent is used to indicate a high level of performance reached by an individual in an irrelevant field to intelligence, and subject to genetic factors. Such fact led some theorists to decline to use the term talent in the field of mental superiority and intelligence. Understanding human intelligence had been a significant issue in psychologists' debates for centuries. Some theorists have objected to the use of this term in the field of mental excellence and creativity because they intended to use this concept initially for those who reach a high level in non-academic fields, like arts and games. They considered mathematics, various crafts fields, mechanical skills, and other areas irrelevant to intelligence.

Table 3.Differences between talent and intelligence

Intelligence	Talent
Intelligence involves both genetic and environmental	A natural skill with which a person is more able to achieve.
The ability to understand and process	The ability to achieve something
information.	

2.7.1 Defining Intelligence

Psychologists at different schools provided various definitions of intelligence. Some related the definition to its functions and others related it to the way it works. As a result, multiple definitions of this concept led researchers to classify these definitions into three groups:

First: It highlights the organic basis of intelligence. This group defines intelligence as an organic physiological capacity in which genetic factors play a significant role.

Second: Emphasizes that intelligence result from the interaction between social factors and the individual.

Third: It is a category of definitions that depend on the identification and observation of behavioural manifestations to judge the intelligence of the individual.

With the growth and advancement of science, new views emerged, and the perception of things changed. What is new? A direct relationship between talent and intelligence exists. Whenever there is intelligence, talent is measurable. Mental talent is measurable by specific measures of intelligence. Through a careful reading of theories of intelligence, such as The "Renzulli's three Rings theory", "Sternberg Triple Theory," and "Gardner's multiple intelligence theory", the findings show that talent, according to them, is part of intelligence and a function of it.

Renzulli (1997) indicated that gifted behaviour results from the interaction of three interconnected episodes. These episodes are (above average intelligence, high commitment to challenging tasks, and high level of innovation) that clarifies how intelligence alone is inconclusive to the emergence of talent only through ability. However, there is a Commitment to the difficult tasks, as well as work in the spirit of originality and novelty and profound attempt to change. The skills of talented

individuals are within those three rings mixture. Sternberg's theory (1985) description of intelligence as being linked to the inner components of an individual and consists of a necessary information process occurring within the individual and consisting of other mental processes. These processes are like planning, control, and decision-making, coding, synthesis, comparison, and response, as well as the cognitive processes, acquired to learn things and to learn about their characteristics as well as everything related to them. As for Gardner (1987), he has divided the talent into several sections. He called his theory multiple intelligences, including mathematical, linguistic and logical, space and music, motor and social, and other names that describe the behaviour as a kind of intelligence characterised by a person only.

Like talents and gifted definitions, and despite going through a long history of research and debate, intelligence also has no standard description (Legg & Hutter, 2007). Intelligence, as known, includes the ability to abstract thinking, the collection, and coordination of ideas, the capture of languages, and the speed of learning. It also includes, according to some scientists, the ability to sense and express feelings and understanding the opinions of others. Even Though the general concept of intelligence for people encompasses all these things and may relate to the power of memory, psychology treats intelligence as a behavioural advantage independent of creativity, personality, wisdom, and even the power of the memory portfolio. The availability of many IQ tests could not help defining intelligence. Scientists, therefore, criticised and accused the IQ test of not being able to identify those with high and low intelligence because the theories that exist today confirm the existence of multiple types of intelligence, and these tests are not able to determine geniuses in all areas.

A scientific fact says, "Humans, in general, do not use all parts of their brains to solve a problem or even to think. They use a specific part in the brain to solve the problem on their own".

Accordingly, we cannot criticise intelligence as one thing or one type because it is of "different types," so a person's performance will certainly not be the same as another. According to (Roth & Dicke, 2005)," the intelligence of humans is referred to as a combination of properties and enhancements found in non-human primates such as the theory of mind, imitation, and language." There is not yet a specific definition for intelligence; even intelligence, in its general sense, differs from one location to another and from one environment to another. In the school, the intelligent student is the academic achiever and the one who gets full marks in his studies and has the highest qualifications among his or her peers. In the business sector, the intelligent is the one who can exploit business opportunities and achieve the best gains even if he or she had no education or academic achievements. Many athletes were and still seen as the geniuses of football today because they could read and predict the opponent's movements in advance and translate them by taking

advantage of the best opportunities and win the game.

Intelligence, therefore, in general, is the tool that enables individuals and groups to better adapt to the surrounding circumstances by exploiting what exists in nature or other resources to solve a problem. In the 19th century, the prominent psychologist Francis Galton believed that intelligence was inherited from the father and was, therefore, looking for intelligence in the children of sons of great leaders. In World War I, the United States of America required those who wished to join the army to pass the Intelligence test to distribute them in positions according to their rankings in the tests. James Flynn in (2007) was able to arrive at a logical explanation that "the results of a person's IQ test are based entirely on the economic, cultural, and scientific and life conditions that existed in his predecessors, which will give a more significant, boost to a high score or vice versa. Latterly, in 1999, the scientist William Dickens of the Brookings Institution in Washington developed a theory that has almost complete consensus among scientists today. The method indicates that those

who had genetically inherited traits will have a preference in an area and would be creative if they could continue practising it. For example, if a tallboy can participate in a football team, he will improve his performance and abilities in this game and will have the opportunity to maintain high fitness compared to other children who do not have the same physical characteristics and are not as tall as he is. (Dickens, 1999). The tall boy in the previous example will invent and excel physically and mentally in this area because his traits would support him. Dickens concluded that those who have inherited qualities will enjoy preference in a field over others and will use their attributes to become superior to them. The inherited traits and the surrounding environment, according to Dickens, have a significant role in intelligence and the growth of personal mental skills (Dickens & Flynn, 2006).

Therefore, intelligence defined differently by writers and scientists, and in the study, it will represent the notion of talents, highlighted in this way to mark the reason why theorists linked intelligence with talents and making it closer to readers. Starting from the American Heritage Dictionary in its fourth edition, (2000), which defined intelligence as "The capacity to obtain and apply knowledge. "Intelligence, according to Lumpers, is a common unified capacity for obtaining knowledge, reasoning, and solving problems (Weinberg, 1989). Galton indicated that individuals possess a "g" factor, which is a general intellectual ability. The factor, according to him, demonstrates in various ways like navigating without a compass or programming a computer (Gardner, 1992). Howard Gardner said that intelligence constitutes of many separate mental abilities that work independently. He identified (9) different types of intelligence to prove his theory (Gardner, 1992). The definition of the" American Psychological Association" (APA) says, "there is a difference between individuals in how they understand the complicated ideas and how they efficiently adapt to the environment. How they learn from experiences, engage in various

forms of thinking, and overcome difficulties by taking thought." Encarta World English Dictionary, 2006, defined intelligence as:" The ability to learn facts and skills and apply them."

2.7.2 Nature of Intelligence

(Terman,1920) said that intelligence is inherited and that it cannot be nurtured by any development programs, while researchers, such as Hollingworth, who worked with Terman and agreed on a lot with his belief about giftedness and inheritance, though on the contrary that talents could evolve through practice and become impacted by environment and other educational factors. (Hollingworth, 1926). Such unsolved conflicts have resulted in losing many talented individuals who were unaware of their talents or are indecisive about the importance of deliberate practice to develop them. The opinions of researchers and theorists' regarding the origin of talents and the identification of individuals to get support or not were discussed and criticised for several years. Vast amounts of papers, articles, and longitudinal studies debated the issue, but this issue is unable to settle on one agreed opinion.

Complications arose from some of the prominent scientists in the field of intelligence like Francis Galton (1909). Galton was a child prodigy himself and worked effortlessly to prove that intelligence is inherited. He counted the number of relatives of intelligent men to obtain a wide range of data from many biographies. He disclosed that there is a connection between intelligence and inherit. Galton decided after studying multiple cases that evidence favoured nature rather than nurture. He accordingly invented the method of testing and named it "the hysteron meter" (Galton, 1874). Charles Spearman came after Galton and supported the theory of two factors intelligence, (the general intelligence and the private intelligence) (Spearman, 2005). He developed "the (g factor) of general intelligence and the (s factor) of specific intellectual abilities." Godfrey Thomson

in (1916) criticised Spearman's work saying that his evidence was not decisive, and it is not as it looks (Thomson, 1947). (Shrkin, 1998; Gladwell, 2008) Pointed at some prejudices and assumptions in Terman's studies as well. The complications of Lewis Terman's (1906) studies on gifted people led to criticisms of his work by many scientists. The criticism of his longitude researches and tests were for being (complex); therefore, many of those who worked with him pulled away from his tests, objecting that intelligence was too difficult to define or measure (Burke, 2016). Terman's test was also criticised by (Lippmann,) and by (Mitchell, 2000) for not having a generalizable sample and for granting his research subjects letters of recommendation for jobs and this made their lives' outcome not generalizable according to (Mitchell, 2000). However, Terman was praised for his contributions to the gifted education, his work to alter the Stanford-Binet Intelligence Scales and his initiative of Genetic Studies of Genius. He, according to them, enriched educational researches in the area of giftedness. Terman as well suggested that IQ tests help classify children appropriately, and it is the strongest predictor to know if the person will succeed in life or not (Warne, 2019). The relationship between gifts and talents links with high IQ scores due to such opinions; however, there is still much controversy and much research on this issue since the twentieth century. Classification of gifted children in the 20th century was often via IQ tests, but as other identification procedures are developing, this may help to identify more talented and gifted children who stood out of the crowd without being measured with the IQ tests, and they succeeded without being as intelligent as many others.

Since educators, parents, and talent experts commonly think intelligence is a significant component of high ability, it is crucial to discuss the nature of intelligence and how it impact gifts and talents to understand the relation between them and the reason for it being a highly controversial issue. The issue here lies in how the word talent and the word intelligence are both

used as synonyms in popular daily dialects, and thus like the synonym of gifted and talented. Such synonyms create numerous confusions about the notion of each word and its distinct characteristics. Intelligence studies on the other side were part of the twentieth-century advancements in the field of education and giftedness pioneered by Terman and Leta Hollingworth. They both conducted the first researches in this field and inspired future researchers at that time to investigate more in gifts and talents. These tests intended to support educators' measures of the children's innovations and differentiated abilities since the early stages of their lives. (NAGC, 2019). Plenty of theorists and researches latterly devoted time and money specifically to learn and understand intelligence (Feist & Runco, 1993). While searching in some chief talent development initiatives in the UAE, such as (MOE) and (NPGP), it appears that they were using intelligence tests as the primary tool to decide who is talented and who is not. Such note, therefore, guides this research towards exploring more about intelligence tests and the role of this globally used tool as one of the primary measurements to asses individuals' talent.

2.7.3 Intelligence tests

One of the methods for the identification of talented and gifted individuals is the Intelligence Quotient Tests known as the (IQ) test. The (IQ) test was formed first by a German psychologist named William Stern in 1912 (Stern, 1914) said it is a total score derived from several tests designed to evaluate the intelligence of humans. The (IQ) analysis is the most thoroughly researched and differed about among scholars. According to many of them, IQ is the most trusted test that measures intelligence, while others approve the opinion that confirms that intelligence is challenging to measure. The IQ test estimated a considerable number of individuals from all ages and categories and viewed their intelligence depending on its results. The most asked questions throughout the years were about the reliability of the IQ test and to what extent its effects qualify

those gifted and talented individuals who scored high numbers in the test to be more successful than the others with reduced IQ. This IQ test, or Intelligence quotient, was one of the first attempts to classify humans in a way that tends to be scientific. These tests indicate by a set of points the humans' location in the order of intelligence. The IQ test that was first invented in (1905) by the French Scientists "Alfred Binet and Theodore Simon" to assess the children's intelligence, was used for a long time by different entities as a tool for educational placement, assessment of abilities and as an identification of talented individuals for jobs and many other various testing needs. Spearman in (1904) developed his two factors theory of intelligence to measure general intelligence, and Terman in (1920) supported the theory that having a high IQ score means an outstanding achievement. Terman at that time leapt and developed Alfred's Binet's test that became a standardised measure to compare individuals' intelligence with others from the same age group. Consequently, Binet's IQ tests remain since then as the most reliable standardised tests used in education to measure an individual's cognitive ability (Benson, 2003).

2.7.4 Criticising intelligence view

One of the most critical questions asked then and nowadays is, "Can IQ be measured?"

This question about how intelligence is measured and if it is one of the testable traits, although it is originally hard to define, is still roaming around researches and studies, and faced by huge and intensive debated arguments. The other question is about being able to use certain criteria to classify humans from the brightest to the less intelligent. According to psychologists like (Ceci & Liker (1986), "IQ test is unrelated to realistic forms of cognitive problems that would confirm to some of what that scientists favour as the symbols of intelligent manners." Many factors, therefore, such as life experiences, activities, and structured learning, can change children or adults' IQ. It

suggests that IQ tests should not be the final proof of the person's talent unless repeated in every stage of the life of the talented individual to be reliable. (Ceci & Liker, 1986)

More scientists criticised the IQ method by saying that many "geniuses would drop out of the geniuses list if they tested their intelligence, and many of the people who make it to the top were not top performers in their schools(Sternberg, R.J., 1996)." Dean Simonton(2000) traced the origins and accuracy of intelligence tests, and asked one of the most important questions: "Can we rely on the IQ tests?" (Simonton, 2000). Simonton found four problems in Terman's and Cox 's studies regarding Intelligence (1) many individuals who scored high ranks in the IQ tests that were conducted by Terman had no ambitions to achieve better positions in their lives. (2) The achievements of some political figures who, despite displaying low ranks in the intelligence tests, occupied great positions in their lives, such as Napoleon Bonaparte (his IQ was only 145), but his achievements overpowered everyone around him (Simonton, 2000). (3) Simonton argued other influencing factors away from intelligence, such as the stable personality, which supports the low IQ rankers to achieve higher positions than those who scored higher ranks. (4) Catherine Cox & Terman's tests in the second termites' book were unequal, and both were not measuring the same thing. Simonton indicated that (Binet & Stanford's) intelligence test measures how the individual develops his basic skills such as memory, inference, and other primitive skills that humans expected to gain when they mature. Cox, on the other hand, focused her tests on measuring specific skills that may be rare even on matured people because they are limited to a specific domain of creativity. Others, like (Plomin & Craig, 2001), viewed intelligence in the way Terman has viewed it; however, (Esqierdo & Arreguin-Anderson, 2012) criticised them as being conservative in their views. Simonton cited Howard's Garner's Multiple Intelligence theory, indicating that this theory contradicts Terman's theory, who views linguistic intelligence and logical thinking, as the two

only criteria must measure intelligence. Terman's studies "Termites" criticised for being determined to show that highly intelligent children were the future leaders of the society, and Terman himself conducted the interviews on their behalf while he was supposed to separate himself as a scientist from the subjects. Many of these kids reflected the influence of being a Terman Kid by stating in a later time how this study changed the course of their lives. A famous Journalist named Walter Lippmann said that the intelligence testers were the "Psychological Battalion of Death." (Lippman, 1922). Lippman criticised the IQ test by saying that it was used to assess American Officials in the Army and reflected his opinion regarding making such IQ tests determine regular people's future and success. More critics from researchers like (Gould, 1996) argued and disputed the reliability of the IQ test by saying it cannot be reliable because it cannot predict some achievements, and it neglects other important aspects of mental abilities. (Stanovich, 2010) argued that IQ might not be as accurate as it is thought to be because it is unable to measure border definitions of intelligence, and it is only examining some areas of intelligence and neglecting others that are assumed to be as important as intelligence such as creativity and social intelligence. Stanovich argues that if the IQ test does not measure the rational thought, then it fails to give us accurate results because the rational thought skills and cognitive processes are as important as intelligence and should be valued as the high abilities are valued. (Stanovich, 2010). (Kamin,2012) indicated that the data generated from the IQ tests have always taken place in both social and political context, and the IQ test is the psychological contribution to the continuous efforts to blame social disasters on fixed biological causes. He added that psychology lost its political innocence concerning IQ testing.

Moreover, intelligence tests faced further criticism from (Maccoby & Jacklin, 1974) when they found differences in respect to gender in the verbal IQ tests. The improvement of scores on

standardised tests did not mean that the cognitive ability of the tested, such as memory, attention, and speed, were improved. The fact that many people become successful was because they were talented and not intelligent, while many others were not successful despite scoring high scores in IQ tests, which makes this tool criticized and questioned. Some scientists like (Renzulli, 2005) said that IQ scores are associated with high grades in schools' subjects, but this association accounts for only 16-36%. Others blamed the IQ tests users for extensive and independent use of these tests to identify highly able students, while the test itself was merely testing intelligence according to (Dweck, 2000). The IQ tests have affected several cultural minority students designated for development programs according to (Esquierdo & Arrefhuin-Anderson, 2012). Other writers such as (Eysenck; Jensen, 1972) believed that the ethnic minority's underachievement is due to the inherited nature of IQ. The IQ tests were in the debate to have shown differences in gender (Maccoby & Jacklin, 1974), although others like (Hyde & Linn (1998) rejected that argument.

After Terman's death and before the release of the fifth and final volume of his longitude study, there was a huge surprise over the fact that appeared to disturb him, when he found out that the children in his study did not become what he expected them to become when they grew up. The performance of those who ranked high scores in the IQ tests was average, and their high intelligence turned into natural intelligence; they became doctors, lawyers, and specialists in various fields. Just two of them, Robert Sears and Lee Cronbach were teaching at Stanford University, and these two continued the long study they were part of it. However, the story did not end here, but it bears a curious surprise according to (Simonton, 2012) when two of the children in the study whose IQ scores were not high scored extraordinary successes against expectations and became Nobel Laureates physicists: (Louis Alvarez and William Shockley.) This paradox

bothered Terman because he could not explain it, and Catherina Cox was no less worried than he was because she failed in her testing with several "geniuses" she chose. "Put it simply, if this test had examined Nobel winners before they win, they would not join the list of geniuses" (Simonton, 2012). Other Terman kids said the tests had negative impacts on their lives and complained of the unfair burden to succeed. The label as intelligent at an early age made many of them arrogant and unworried. In conclusion, the problem is the concept of intelligence is that a single dimension defined it earlier (Plomin, 2001). Finally, and although many researchers are exerting efforts to improve the concept and to find the range of inheritability in it, (Sternberg, 1985; and Dweck,2000) found that the injustices that came with this view of intelligence made these tests without credibility to be predicting the future success, on the contrary of what they were designed to do. Therefore, this means other ideas about the nature of intelligence should be thought about to lead to a more substantial understanding of it. "

2.8 Theories of Intelligence

Many theories deal with intelligence, such as the theory of (Jeane Piaget, "The theory of cognitive development"). Piaget's theory, in the area of developmental psychology, was very influential in the 20th century (Huitt & Hummel, 2003) as well as the psychometric theory for understanding the construction of human intelligence (McGrew, 2009). What combines these theories is that they agree that intelligence is an integrated structure that makes performance in a task linked to the performance in other jobs. Most of these theories focus on the external aspect (appearance) of the process of education and learning, and they did not reach the essence of the student and the analysis of his or her actual capabilities such as intelligence and the ability to address situations and problems and find solutions and descriptions of such issues and attitudes. However, in 1983, Howard Gardner came to break the mysteries of human intelligence. His observation of pre-school

children explained that these facts were not correct; saying that many pre-school teachers knew that a child has the skills of interpersonal relationships or so-called social intelligence, while another child has athletic intelligence. These tendencies do not appear suddenly but through the passage of activities comprising this intelligence.

Consequently, intelligence maintains to be a debated definition in psychology. Some researchers defined it as exceptional ability, while others said it includes a variety of talents, skills, and abilities.

The first psychologist, who invented the term (IQ) intelligence quotient, based on the work of Alfred Binet, was the German psychologist and philosopher (William Stern) who pioneered in the field of personality and intelligence. Francis Galton, (1869) pioneered in studying the human differences believing that humans inherit intelligence. He linked the concept of talent with intelligence (Jensen, A.R., 2002). Alfred Binet (1904) was the first to develop the intelligence test (Binet-Simon scale) (Binet.A. & Simon.T, 1916), (Terman, 1925) conducted a forty years study to prove the connection between talent and intelligence. Charles Spearman (1904) said that "intelligence defined as a general ability that can be measured and expressed in numbers" (Charles Spearman, 1904). More studies by (Hollingworth, 1926; Hudson, 1926; Vernon, 1971; Mortinson, 1972; Renzulli, 1977) linked talent to high IQ. In (1984) Renzulli introduced his "Three rings of giftedness theory" to prove that talent depends on the integration of three fields of high intelligence, high creativity, and the ability to implement ideas and solve problems with a high level of achievement (Renzulli,1984). (Davis, 2004) pointed at different characteristics that can identify talents better like curiosity, independency, obligation, and open minds. (Heller, 2004) developed the Munich Model of giftedness, to explain how talent can be developed based on four

differentiated dimensions dependable on each other such as Giftedness, performances, personality, and environment.

2.8.1 Robert Sternberg: Triarchic Theory of Intelligence

Robert Sternberg, in the past two decades, has developed several contemporary theories of intelligence (Sternberg, R.J., 2012). These theories had essential implications for detection and teaching of gifted people. The Triarchic Theory of Intelligence looked at talent from the perspective of leadership(Sternberg, 1997). Sternberg's theory constitutes of three consistent elements: Intelligence, creativity, and wisdom and developed two models of giftedness after observing intelligence from divergent approaches (Sternberg, 1997). Sternberg's Triarchic Theory of Intelligence looked at the factors of (context and cognitive) as what constitutes intelligent behaviour. The theory explained the connection between the individuals and the inner world. It described it as what goes between the inside and the outside world, and it highlighted three central aspects of intelligence (Sternberg, 1997). Its designated intelligence as located within the individual and his behaviour and the context of ethics. Cognitive, according to it, is the component that constitutes the basic unit of intelligence. In 1997, Sternberg presented an improved picture of this view, called the theory of successful intelligence (Sternberg, 1997). Successful intelligence means the person's achievement of success in his or her work life (Sternberg, 1997). Success should be according to the concept of talent and how this success defined in his or her sociocultural environment. The intelligence helps talented persons employ their strengths and compensate for their weaknesses to adapt to the surroundings by shaping, modifying or changing in cooperation with their creative and analytical abilities (Sternberg, 1997).

According to Sternberg's theory, successful intelligence consists of four components or includes three types of intelligence: (1) The analytical Intelligence, which is the ability to analyse, judge, criticise, compare, evaluate, and observe. (2) Creative Intelligence, the ability to innovate, discover, visualise, and develop assumptions. (3) Practical Intelligence, which is the ability to solve well-defined non-school life problems that can have several solutions and several ways to reach these solutions and (4) Balanced Gifted is the person with proper levels of the analytical, creative and practical intelligences(Grigorenko, E.L. and Sternberg, R.J., 2001)- (Sternberg, R.J., 2005)

Table 4. Sternberg's classification of giftedness

Analytically Gifted	Creatively Gifted	Practically Gifted	Balanced Gifted	
The talented person in this	The talented of this class	The talented person in this	Is the person who has	
class usually performs	do not require intelligence	category knows what he	proper levels of analytical,	
well in school duties as	tests, but they need tasks	needs to succeed in his	creative, and practical	
well as in traditional	or tests that require	environment and reveals	abilities, and knows at	
intelligence tests.	generating new and	his intelligence in	what time to use these	
	authentic ideas such as	situations and situations	skills?	
	writing short stories and	with specific content.		
	graphics and solving			
	unusual mathematical			
	problems.			

(Grigorenko, E.L. and Sternberg, R.J., 2001)- (Sternberg, R.J., 2005)

The theory suggests that traditional intelligence tests, school tests, and school curricula are primarily analytical intelligence not related to the individual's experience of life, nor his creative experiences. Thus, those well-known intelligence tests mainly relate to academic skills and not to measure the applied and creative intelligence that is essential to success in the individual's working life. The theory emphasises that the adequacy in dealing with real-life situations depends on the implicit knowledge of the practical nature and not the academic one, not taught directly or systematically (Sternberg, R.J., 2005).

Sternberg and Grigerenko (2002) have reviewed the relationship between the gifted education model and other well-known models such as Renzulli's, Gardner's and Gagne's models that distinguish between talent and excellence (Sternberg, R.J. and Grigorenko, E.L., 2002). The theory of the general factor of intelligence has concluded that there is no contradiction between the theories of successful intelligence. Nor the theories and other models as the theory of successful intelligence may fill the deficiency in some of these models and may integrate with them. Sternberg (1985) presented a new trilogy of human intelligence that interprets mental talent more comprehensively than traditional intelligence theories and goes beyond what intelligence or achievement tests measure.

2.8.2 Criticising Robert Sternberg's Theory of Intelligence

Sternberg's theory not only concentrated on the skills of the people used in real life and stepped away from the high academic achievement, but also focused on the concept of intelligence domains and contrasted with the abstracting testing techniques used in the IQ tests. This theory supported the changes in the look at talents and giftedness. It boosted the importance of performance and the other abilities that constitute the concept of talent and how we should view it. Despite all the praise, the theory faced several criticisms, such as being difficult when it comes to applying it in the schools within an educational setting. Gottfredson (2003) criticised Sternberg's methodology, saying that there is insufficient data and that it is unsafe. Correspondingly, and despite the importance of this theory in the Western psychological and educational research, this theory does not seem necessary for the Arab world researchers except concisely and briefly. Accordingly, it makes the theory looks ambiguous and not familiar.

2.8.3 Howard Gardner. The Theory of Multiple Intelligence

In (1983), Howard Gardner wrote extensively about specific domains of talents and intelligence. He said intelligence makes humans able to respond to problems and value their cultural aspects and elements (Gardner& Hatch, 1989). He noted that IQ tests only support those who are familiar with tests and can answer them. He indicated that there are "different concepts of what intelligence means in different cultures" (Smith, C., 2006). In his theory of multiple intelligence, he wanted to blend a mixture of factors such as cognitive knowledge, biological, and cultural in intelligence. Gardner then came out with a different approach that distinguished him from the rest of the researchers by trying to explain the nature of intelligence. His theory, derived from his observations of individuals with exceptional mental abilities in some fields, who got low grades in the intelligence test or were below intermediate levels. Gardner thought that intelligence is composed of many separate skills, each of which operates independently of the other. The theory holds that people have unique patterns of strengths and weaknesses in their mental capacity (Gardner, 2011).

This theory founded on several bases:

(1) Intelligence is a group of intelligence developed to progress. (2) Each person has a unique configuration of multiple intelligences. (3) The intelligence differs in its growth within the individual or between the individuals. (4) Multiple intelligences can develop to varying degrees if available. (5) The various intelligence and mental cognitive abilities behind each type can measure and identify.

Therefore, the theory known as the (MI) Multiple Intelligence theory was born to define human intelligence by involving several abilities used to fix issues and to create and find questions to

solve them. Gardner's theory believed in the geniuses of the learner, his ability to give, produce, innovate and create, solve severe problems, and confront complex situations. It is why academic psychology and many educators widely accepted and embraced it (Gardner, 1993, 2000). It supports researchers' agreement regarding the ability of children to acquire competence at an earlier age than usual (Radford, J., 1991). It also helped the differences in talent abilities and their needs of the practice and their need for social and emotional development according to a group of scholars and researchers including as an example (Ericsson, K.A., Krampe.R.T. & Tesch-Romer, C., 1993). (Ehrlich, P.R., 2000), (Terman& Oden, 1959, Whybra, 2000), (Neihart, M., Reis, S.M., Robinson & Moon, S., 2002). This theory supports the identification of talents and highlights the importance of serving their needs for development by deliberate practice after identification (Tirri, K., Cho, S., Ahn, D. & Campbell, J.R., 2017). In his theory, Gardner devised a set of intelligence, which he did not claim to be comprehensive. The intelligence comprises (1) Linguistic intelligence, talents in the domain of languages that includes new languages and the ability of expression by speaking the language or writing in it (Gardner, 2011). (2) Musical talent: that means the skills of performance and appreciation of music (Gardner, 2011). (3) Spatial talent: The skills of realising shapes and the relationships of space objects (Gardner, 2011). (4) mathematical talent: which means having a gift in math and any other activity that requires logical thinking (Gardner, 2011). (5) Bodily-kinesthetic talent: This means the ability to solve physical problems (Gardner, 2011). 6) Interpersonal talent: This means understanding the other's motivations and desires and work with them in a way that makes it beneficial (Gardner, 2011). (7) Naturalistic talents are the talents of understanding living things and reading nature (Gardner, 2011). (8) Existential talent: Is how the individual tackles the questions of (Why we live and why we die) (Gardner, 2011), and (9) Intrapersonal talent: Self-understanding meaning asking

ourselves what we think, how we feel and what we want(Gardner,2011). Howard Gardner radically reconsidered intelligence and its effects on the learning process by introducing his theory title:" multiple intelligences" in his book "Frames of Mind" (Gardner,2011). He rejected the idea of one intellect and asserts the existence of many mental abilities, which is relatively independent of each individual called "human intelligence", each having its characteristics and features. This theory has forever changed the concept of human intelligence and its impact on the educational system.

2.8.4 Criticising Howard Gardner's Theory of Multiple Intelligence

Howard Gardner's theory of (MI) (Gardner,2011) was an attractive theory for the educators, not only for being a dynamic theory but also because it allowed educators and teachers to estimate and distinguish more abilities in more children in classes and schools at any time. The theory supported the assessment of skills away from the traditional way that has always emphasized on solutions and answers to the questions. Gardner said that each intelligence from the nine bits of intelligence that we have must have a distinct tool to measure it. Consequently, it focuses more on the way students will reach all solutions or answers. Unfortunately, there are no specific tests to measure all the multiple intelligences differently.

This theory, thus, offers a variety of differentiated practices and tools for every student with high ability. It explained how children differ according to their different type of intelligence, their culture and their abilities (Gardner,2011). By offering every child and assessing each of them with different types of processes and methods, these differentiated tools enable the students to solve various issues that may occur and need multiple intelligence to solve them. They also support the theory's principle regarding intelligence development and culture. Several studies conducted on the application of this theory indicated that the theory of multiple intelligences had been used

recently to support the identification and development of the gifted and talented in schools (Fasko Jr, D., 2001). It is supportive of highly abled students because it helps educators to see how different methods complement rather than contradict. It also helps to organize the usage of tools and strategies that would support all children and provide each of them a place and an opportunity to succeed (Armstrong, 2003).

- (1) MI points out that intelligence is not a phenomenon. It consists of different abilities (Armstrong, 2003).
- (2) Multiple intelligences correspond to the recent studies of the brain based on the segmentation and classification of brain capacities and their variation from person to person (Armstrong,2003).
- (3) The theory of (MI) is also supportive in cases when teachers want to deal with many students using different strategies(cornerstone.edu).
- (4) This theory encourages a differentiated educational environment in which students can achieve their potentials by unique aspects for each of them (Fierros, E.G., 2004). It also explains for teachers how to design new approaches in teaching for different students and how to provide them with a framework that helps them submit any educational content for many children with different abilities and from different cultures in different ways. It helps the teacher divers their teaching methods to consider the diversity of these learners and to help them reveal their abilities in their ways. It accordingly can eliminate and decrease student's violence towards their peers, school's environment and the society as well. It helps educators nurture sensible students and the teaching of intellectual skills. This theory as well contributes to students' grouping and determine their needs for scientific and psychological services (Fierros, E.G., 2004). The theory of multiple intelligences reduces the number of processes performed by teachers in schools. One of the

processes that benefit from this theory is the process of relocating students who have learning disabilities and special needs to different classes. Solving this problem helps teachers and students find a way to settle and feel secure in their places. Besides its role in increasing their selfappreciation and helping them to achieve well among students instead of being underachievers while they have hidden undiscovered abilities. Moreover, it should be easier to apply the theory of multiple intelligences and not to complicate any procedures according to it. This theory does not mean that the teachers should present the lesson in various ways and using many procedures or tools. Teachers do not have to try to develop all kinds of intelligence through one subject, but they need to be able to know what to apply and when. According to (Gardner, 1997) such thing is a misunderstanding of his theory and why it is there. The (MI) base that there are different intelligence and each of the intelligence responds to different content. It means that the capable teacher should be able to select the right content out of a wide range of them to teach the students accordingly. This content should include the suitable intelligence that matches the student's ability and help him or her progress through it, as well as the methods of teaching, and the proper educational activities (Gardner, 1997).

Other scientists criticised the theory by saying that the (MI) theory lacks support from scientific research; therefore, it lacks measurability according to them. (Waterhouse, 2006). Contrary to Gardner's assertions, in theory, there were strong correlations found between the intelligences, and this supports the theory of general intelligence, not the multiple intelligences, which means if someone is intelligent in one area, then he or she tends to be useful in the other field. Other scientists' arguments are about the solid relationships between diverse intelligence and how is this relationship. They say that although there are strong relationships between the diverse intelligence, however, each intelligence is different from the other. Again, the question is which test can develop

to measure which intelligent and how can any test measure which intelligence is dominant in the individual? Unfortunately, Gardner himself admits that he has no empirical evidence for this theory. Furthermore, several theorists have criticised the association of multiple intelligences with multiple learning styles, which Gardner asserts are lacking in empirical evidence. However, this theory evolved and become popular within the field of education, and its success came from its equity and giving all children their opportunities for success (Hoerr, T.R., 2000).

A significant criticism of this theory is the definition of intelligence. Objectors criticized Gardner for using the concept of intelligence in a way that converses with the intelligence recognized concept in the field, (Davis, Christodoulou, Seider & Gardner,2011) Others say that this theory does not acknowledge the existence of intelligence, but the concept of intelligence to what is traditionally known as capabilities or competencies. (Sternberg, 1983). Defenders of the theory, on the other hand, claim that the traditional definition known for intelligence is inadequate, constricted, and narrow-minded that cannot use it within an area that is supposed to be flexible and intelligent. They added that the concept of intelligence must develop in a way to match the developing reality (Nikolova& Taneve-Shopova, 2007). The theory was also criticised by (Gottfredson, 2005) who as well criticised the Triarchic Theory on the same grounds.

Gottfredson Said that Gardner, in his Multiple Intelligences Theory, used untrustworthy data and interpreted this data in a way that is not challenging or even supporting the (g) factor (Gottfredson, 2005).

Conversely, researchers like (Visser, Ashton & Vernon, 2006) tested eight of Howard's intelligence. They came with the result that the (g) factor exists in most of the intelligence and not as what Gottfredson said. They concluded by indicating that "The multiple intelligence is secondary to the g factor". Another criticism was about how each learner has his style of learning,

and this means the teacher must correspond to this diversity by implementing a different teaching style. A criticism for this statement was about the absence of evidence about how different students learn in different ways and what makes this accurate. However, and on the other hand, no evidence that demonstrates how the customisation of the "teaching methods" to suit the "learning method" can be useful and is an approved way. Gardner personally had his doubts regarding the misconceptions related to learning styles, as previously mentioned.

Despite all criticisms that accuse this multiple intelligence theory of being on shaky grounds, the theory is prevalent and finds its way in the educational systems. The question if this means we should go back to the general intelligence theories and one size of education that fits all children.

An optimistic proposition indicates that there is enough evidence that proves how learning

outcomes can enhance if children knew how to access the content using multiple differentiated ways and various methods and tools to learn and then to demonstrate their knowledge and skills. This theory can also benefit the educational field when the teaching differentiated methods would be able to provide students with their differentiated needs and help their development and the growth of each of them. It is an ethical theory that supports the identification of students who appear more talented in different areas and not in one only. It would be very beneficial to measure their abilities according to this different intelligence, as suggested by Gardner and to enhance their contributions by reflecting more on the prominent talent he or she has.

2.8.5 Gagne's Differentiated Model (DMGT)

Gagne believes that learning is a semi-permanent change in the behaviour of the individual because of the experiences or exercises in an educational position. He focuses on the content of the learning subjects as well as how to organise and present this material to the learners. Gagne emphasises the

quantitative side and the organisation of the information in the educational process. Therefore, he suggested that the learner's readiness to learn any new knowledge depends on the amount of this learner's preliminary information; and to what extent is this information necessary to support his or her experience. He believes that learning involves eight hierarchically arranged patterns from simple to complex, and each higher education in the pyramid depends on mastering the lower. The task analysis, according to Gagne, is fundamental to understand the structure and the sequence of information presented to the learner, which is what Gagne called the hierarchical construction of learning (Gagne's Differentiated Model of Giftedness and Talent (DMGT) (Gagne, 2009) indicates that talent is an innate readiness, and to help talents excel is the result of this readiness development(Gagne, 2009). He identified many aspects related to talent and put them in a theoretical context that determines the relationship between them and the aspirations of society. He said that the social factors, educational system, and other "inner and outer factors", are what will help to invest in talent and achieve remarkable achievements. Gagne sees talent as a natural ability, that can turn into a high-level performance with incentives, motivation, and support. The abilities are raw at the beginning; however, by receiving positive impacts that support the growth and development of the individuals' talents, this individual will excel. Gagne indicated that the individual could not be superior if he or she is not initially talented; therefore, if the factors influencing the development of the talents are not available, then even the initially talented person will not be able to become superior or to classify as talented. Not every talented person, according to Gagne, is excellent, but the opposite is accurate, and every superior is talented. The talents may remain invisible, and this will lead the talent to become a potential talent with ability or readiness. Gagne explained how the cause of many school dropouts, that may occur within high skilled people

in drawing or poetry, or other domains of talent are due to the lack of influential intermediary factors that would support identifying and refining these dropouts' talents(Gagne, 2009).

Gagne clearly distinguished the differences between talent and giftedness in his recognised model, that was first developed in the year 1985 and later re-drafted in the year 2000. He named the model (The Differentiated Model of Giftedness and Talent (DMGT). This model consists of six factors; the first two are Talent and giftedness, while the other four interpreted as the factors affecting the talent, drive talents and giftedness to achieve excellence include:" the natural abilities, environment, intrapersonal developmental process and the developed competencies" (Gagne, 2009).

Gagne's model differentiates between giftedness and talents by saying that" giftedness" is what a person possesses and uses one domain of his or her outstanding abilities (gifts). The use of this ability should help this individual to be one of the top 10% of peers in his age with skilled and unique abilities and immense knowledge in one domain or more of talent (Gagne, 1985). Gagne's model changed the common understanding regarding giftedness and talents by indicating that giftedness meant academic achievement and talent is the ability to excel in the fields of arts, sports, and other performing talents. Gagne (2005) model presented four different components. These components exhibit the complications accompanied with talent development and what happens during this development (1) (the intrapersonal catalysts: physical and mental characteristics: Personal qualities and self-management), (2) (Developmental Process: Informal, formal learning and practising), (3) (environmental catalysts: physical, parental or teacher, provisions, events involvement in the life of the talented individual), (4) (learning and practising: systematically developed skills) and chance.

2.8.6 Criticizing Gagne's Differentiated Model (DMGT)

The criticism of Gagne's model was around its recurrences and that it repeats the themes of Sternberg', (1985, 2000, 2002, 2004), Renzulli (2005), and Ericcson (1993 themes). They meant by their criticism to say that the learning and practising part in Gagne's model is repeated and is not providing any new thing. However, Gagne rejected that because he believed that his work was different and is not repetition for any other. He focused on his theory on the discrepancy between giftedness and talents besides introducing the prevalence rates (top 10%) and separating between the natural abilities and the developed skills. According to Gagne, this makes his model different and not repeating any other models.

This theory can be complicated according to researchers who thought the first five types of learning depend on the behavioural theory, which emphasizes that students need to be stimulated to receive response and reinforcement besides the difficulty the teachers face when they have to arrange for some subjects learning to be in hierarchical systems such as mathematics and languages. Another complication that occurs in this theory is about model complications. Such complexity in Gagne's model does not support teachers in classrooms, but on the contrary, it makes the usage of this model more challenging and complicated for teachers. The use of Gagne's model in classrooms requires the use of psychometric tests to help teachers assess the learners' potential. However, these tests should implement in the presence of a psychologist and while no educational psychologists would be available in classes when those tests are applied therefore it will be difficult for teachers to assess their learners as they should (Wellisch, 2016. (Wellisch, (2016) Studied and examined Gagne's model (DMGT) and indicated that (DMGT) mainly based on behavioural/biological research. It ignores genuine environmental factors, and its application relates specifically to gifted achievers who are already succeeding. The model, according to

(Wellisch, 2016), does not include the gifted underachievers such as those who have problems caused by environmental reasons without a legitimate claim for the identification or an appropriate educational pathway. (Wellisch, 2016) then proposed an expanded conception of underachievement and recommended that the Australian education department, guided by Gagne's model should switch to use a revised Model "Inclusive Gifted Identification and Progression" as a replacement for Gagne's (DMGT) (Wellisch & Brown, 2012).

In conclusion, Gagne's model, according to several theorists and scholars, is a widespread wideranging model. It helped to arrange learning subtasks that lead to final performance learning. The
theory is cumulative learning, directly applied in schools as learning, and affects the rate of
cognitive growth of students. It also helped in setting goals and identifying students to stimulate
their motivation to learn. The theory also links previous and new learning to help students and
make it easier for them to adjust their learning with their experiences. It is useful as well in building
diagnostic tests used at the beginning of the teaching to clarify the learner's skills and their
proficient. According to Smith & Ragan, Gagne's differentiated process by following new
researches to refine and improve his model made him introduce a differentiated model from other
theorists and not repetition as researchers claimed. He usually extends, develops and updates his
theory based upon new research, and this differentiated him from others (Smith & Ragan, 1996).

2.8.7 Three-Ring Conception of Giftedness

Renzulli's theory was born after lengthy studies of large numbers of individuals who had extensive contributions to society. This theory, called the three-ring model, is a qualitative leap in the field of gifted distinction and a shift in the type of programs offered to them. This theory assumes that gifted behaviour is the result of three characteristics found in an individual. These characteristics

are high levels of (1) ability, above average (2) creativity, and (3) determination and commitment to perform specific work. Task Commitment and Motivation are basics in giftedness (Renzulli,2012). Individuals who exhibit gifted behaviour can usually combine these three characteristics and activate them to produce a spectacular outcome in one of the areas of benefit to humankind. Renzulli's perception of the nature of talent conveys the understanding of it as a mental or physical gift given to limited individuals who are able, because of that talent, to achieve the appropriate successes by using their abilities, that the talent of mind or body needs special care and attention invested as appropriately as possible. The presence of high ability in a domain of talent is not enough to achieve excellence, but there must be creative capabilities established through the development programs that develop their thinking and creativity. Also, a well-trained educator can keep the motivation for work and achievement high. Perhaps one of the essential things that can come out of this theory is that there are students in classrooms and children at homes seen as having limited abilities or achievements while they possess multiple talents. They only need a professional educator to get these talents to exist to start their countless successes further.

Renzulli' used different definitions to describe his model

- Model Triad Enrichment
- School-Wide Enrichment Model.
- Revolving Door Model.
- Multiple Menu Model

Renzulli indicated that there are two types of "giftedness" (1)-Schoolhouse Giftedness demonstrated through test results and (2) Creative Productive Giftedness. He identified three phenomena's: Brilliant, Personally Creative, and Creative. (1) Brilliant refers to unusual and

stimulating thoughts (2). Personally, creative means people who run-through diverse new experiences in the world in different ways; however, they may not share their experiences with others. (3) Creative refers to those who have changed the culture.

He as well defined the highly able students as those who possess some qualities or can develop and use a mixture of qualities in any valuable field of human performance. Children who appear to be interactive and can build interaction between the three groups require different educational services and other opportunities that the mainstream educational programs do not usually provide them.

2.8.8 Criticising the Three Rings Model of Giftedness

Renzulli's contributions were remarkable and supportive to the field of education. To propose motivation as a factor in giftedness was supportive and clarifying what it needs to make gifted and talented students thrive. He also looked at the interactivity of the three factors: ability, creativity, and motivation in a way that made them very useful and clear to adapt in classrooms. Teachers view his model of giftedness as useful and accessible to them in schoolrooms. The model explains the concept of under-achievement, and it is simple enough for teachers to understand it and use it to identify giftedness in their classrooms. However, (Gagne, 1985) argued this model and discussed it critically because he saw this interactivity as challenging. He could not as well understand why do we have to consider a student with an IQ of (130) as not gifted because he underachieves in the classroom due to not being sufficiently motivated or encouraged by any means to succeed (Gagne, 1985). Gagne said that the creativity element in this theory might need more interpretation as well to make it more transparent and understood. Gagne thought that Renzulli's model is limited to academic achievement and does not refer to other talents that need

to have the teacher's attention as well. Other critics assumed that even though Renzulli's definition combines some characteristics of educational-oriented definitions and definitions of features, it still suffered some deficiencies such as (1) ignoring the mentally gifted children with low achievement. It is implicit in the requirement for the effectiveness of each of the three components of talent and excellence, while studies have proved the existence of mentally gifted children at different levels of school whose low level in school achievement was a result of their lack of motivation to learn (Rimm, 1987). (2) The model does not explicitly indicate the level of performance required for each of the three components of talent and excellence. Saying "The overall ability is "above the average "or that the creative and motivated abilities are "high levels" do not solve the problem. However, and despite the criticisms of Renzuli's definition, he provided many additions to this field, including broadening the concept of talent and excellence, highlighting the role played by motivation in achievement and the importance of interaction between general, creative and motivational abilities for planning special programs for gifted and talented. Renzulli, as well, emphasised looking at talent and excellence as a developing progressive situation.

In summary, some of the models are complex and can be difficult and not useful if the intentions are for them to provide proper use in schools. However, the benefit of reviewing them and their criticism is to make identifying talented individuals reasonable and to stop the bias of the IQ tests. They also show how intelligence can improve and how they are helpful to decide which one to select for talented identification that will be the central issue in this study.

2.9 Employing the Theory of Multiple Intelligences

Despite the long extensive debate over the meaning and the concept of intelligence and the differences between gifted and talents, the field has not yet focused on a definition that conceptualises intelligence or talent as well as highlighting the importance of identification and development process. Theorists in this area are still not stable about a description that explains talented individuals' differentiated traits, such as being non-committed, in favour of loneliness and other emotional attitudes that make them seem strange from others. The main problem in being unable to achieve high grades in their academic careers is another issue that they should address too.

Therefore, and for the sake of this study, despite all the criticism imposed on Howard Gardner's Theory (MI), this theory is widely thought to be one of the essential theories of psychology and contemporary education.

The question put here is to what extent Gardner's theory can provide a useful framework for the educational institutions to plan, design and implement the identification and development programs to discover more highly able students in the UAE?

In the next section, a detailed description will try and identify how can this theory apply to gifted and talented identification and development in the UAE. Looking at Gardner's (9) intelligence, they all promise to be an effective way to identify different students abilities and to support multiple modes of identification and development accordingly. According to Gardner, the future technology will help in more development of intelligence (Gardner,2011), and this reflects on the ability of this theory to be a framework that can infuse the efforts in these institutions and the UAE's vision towards the future.

The theory of (MI) can serve the purpose of this study and support its aim towards implementing identification and development for different highly able students. The development process should differentiate to suit their differences and types of talents. The theory of (MI) believes in the existence of multiple, diverse and independent intelligence, therefore, this support the research aim to explore the implemented identification and development programs and to what extent they meet the differentiated needs of the gifted and talented students in the UAE. It is a fact that gifts and talents' thrive once they get the appropriate practice, encouragement, motivation, education, and training. Applying the concept of creating multiple initiatives to nurture different intelligence will help in satisfying the needs of a wide range of different talents if not all of them. Employing the Theory of Multiple Intelligences within talent development initiatives is crucial to set the educational and instructional goals to support teachers and other educators (Armstrong, 2003). These goals are achievable and limited to the following objectives: (1) the theory believes that learners have multiple intelligences and it is possible to look at them in a way that allows more talents to exhibit. That intelligence includes a variety of talents that helps and supports the institutions' include more challenges and differentiated tasks in all their development programs. The variety of talents allows more identification of a variety of highly able students in the areas of linguistics and languages, logical-mathematical, natural intelligence, interactive intelligence, selfintelligence, existential intelligence, musical, motor, and visual-spatial intelligence that constitute the main domains of performing talents(Gardner,2011).

- (2) This theory contributes to solving problems related to individual differences, and it seeks to develop genius, talent, production abilities, innovation, and creativity.
- (3) The theory as well reveals the strengths and weaknesses of the learner, primarily as they address the obstacles and learning difficulties. Thus, if traditional intelligence theories focus on the

learner's flaws, the theory of multiple intelligences takes care of what is positive in the learner, explores other abilities of intelligence, ensures the development of smart skills, develops the capabilities of the learner to solve problems and work on his or her innovation, production and creativity. It invests in the intellectual abilities of the learners in their academic education through educational activities in the fields of this intelligence, and its validity represents itself in teaching programs for ordinary children and people with special needs.

The (MI theory), therefore, focuses on different abilities to practice various activities so that each child can benefit from any activity that suits his or her intelligence, and it allows each learner to achieve equality and to distinguish their unique aspects. Therefore, this theory of multiple intelligences will support the study by its most critical theoretical components, educationally and ideologically represented to identify more individuals that are talented and to make them better served and allocated to multi development programs with more planned strategic methods and tools.

The methodological mechanisms in dealing with multiple intelligences are as follows:

What are the objectives of the gifted development programs? What are the best methods to implement them? What are the mental competencies of the learners to whom the program aims?

How can each program provide different abilities with different development in different ways, considering the multiple intelligences of different talented learners? How are the goals of the talented student determined in each application to ensure that each program contributes directly to achieving that desired goals?

The methodology used to apply the multiple intelligences theory in the initiative must come to be planned while preparing the programs to include as many possible intelligences as possible. The

plan should consider the learners' different intelligence during the preparation of the plans. The content should consider how skilled individuals would familiarise themselves with the process of tutoring and how they can select the appropriate program to develop their intelligence. Trainers and talented individuals should be involved in the program's preparation and to discuss their views as well. It is not essential to introduce all intelligence in one program, but other intelligence can include within the next program.

Table 5. Summary of Theories of Intelligence

N	Theory-Name	Theorist	Year	Summary	
1	Differentiated Model of Giftedness and Talent-DMGT	Gagne	1985	Stated that talent is an innate readiness, and to help talents excel is the result of how they develop this readiness. Talent development in this model presented by transforming natural abilities to become developed skills. Three compounds can support or hinder the process: 1) Interpersonal, 2) Environmental, 3) Opportunity.	
2	Theory of Giftedness- Triarchic	Sternberg	1985	looked at talent from the perspective of leadership and its three harmonised elements: Intelligence, creativity, and wisdom	
3	Multiple Intelligences	Howard Gardner	1989	Intelligence makes humans able to respond to problems, value their cultural aspects and elements, and become able to give, produce, innovate, and create, solve severe issues, and confront complex situations.	
4	Renzulli's Three- Ring model	Renzulli	1997	Proved that talent is the integration of three fields of high intelligence, high creativity, and high ability. The ability to achieve highly in implementing ideas and solving problems.	

Chapter 3. Institutions related to supporting gifted and talented programs in the UAE Introduction

This chapter outlines the background of five educational institutions in the UAE. It explores the development of `programs provided for highly able students in UAE schools. This chapter provides a specific background and context to the primary research study around the educational institutions in charge of education in the UAE.

3.1 Rationale and Purpose

The aim of these five specific institutions selected for this study is that they are the leading institutions established to look after education in the entire UAE. The primary responsibility is to ensure making education available and accessed by every citizen and residence in school-age in the UAE. These institutions were the most suitable subjects for the current study. They could mirror the educational system and position of education in the UAE, including gifted education as well. This chapter focus is on how these institutions are providing the gifted and talented development initiatives and if these programs provided equally for all highly able students, knowing that such institutions located in different emirates, work independently. However, they follow the regulations of the government in the UAE.

The launch of the National Agenda in October 2016 within the UAE Vision 2021 coinciding as well with the UAE 50th National Day was to focus on transforming the educational system to become an excellent system in the entire UAE (Vision2021.ae). It requires a full revolution that changes the current educational system and the teaching methods from their roots. This National Agenda, as interpreted and explained in the (Vision2021.ae) website aims to build a smart system that adapts the new technology and uses the (AI) to allow schools, universities, and all regular and

highly able students use the updated technology and devices as the basis for their learning. It also aims to improve the teaching methods and execution of researches and projects by adapting this technology to support them and enhance their achievement. Another aim for this National Agenda is to promote and reinforce more enrolment in preschools in order to start investing in human talents and abilities from an early age. It plays a significant role in impacting children's personalities, support their plans and sharpening their skills with deliberate practising and motivation from early ages. Moreover, the UAE National Agenda targets students' achievements by making them ready to rank high among the best students in the world in various subjects and domains of talents such as artistically and scientifically talents. Students should be able to compete in several areas, such as the area of reading, mathematics, and science. They should be knowledgeable and fluent in the Arabic Language as well. The Agenda's objective, furthermore, is to raise the number of graduates from secondary schools and to reach the international standards in this aim. It also aims that all schools would have exceptional educators, with high-quality leadership qualifications as well as a globally accredited teaching staff. (vision2021.ae)

The concern in the National Agenda is on how to implement the best methods and structures to accelerate, activate, and update gifted and talented learners in schools to become ready 'for a competitive knowledge economy. The significant changes in the global economy were behind the UAE inspiration to launch this National agenda to help in putting the country in the heart of these coming economic changes. To main objectives of this agenda is to transit the economy from a regular economy dependable on oil to a knowledge-based economy that relies on the global advancements and goes with them. The UAE is urged to focus on various economic, touristic and commercial factors that will support the UAE as a capital for innovation. This Agenda views the importance of research as the support of the country's continuous growth. The Agenda would

support the strength of the framework of the regulations and raise legal awareness in the main sectors. It would encourage and stimulate the quality sectors to work for more improvements in the country by making it more appealing for investors. The National Agenda as well would put the UAE amongst the best countries in the world in the area of entrepreneurship(vision2021.ae). It requires encouraging the nationals to multiply their efforts and increase their skills by enabling them with different tools to become the economic driving force in the UAE.

Moreover, the Agenda's efforts aim to encourage and grow a culture of work success in schools and universities to enable the new generations with different skills and ambitions to thrive in the area of entrepreneurship. This Agenda would support the UAE's ambitions as well to become a leading country in the development of gifted and talented in several areas including leadership and entrepreneurship by implementing various processes and methods to support them and to make their skills easier for progressing. It targets high levels of national participation and involvement in the private sector workforce to increase their productivity in the creative industries sectors as well. The development of quality processes and procedure, as well as launching other supporting initiatives specifically for "gifted and talented" individuals, is an integral part of this Agenda's objectives and UAE's Vision 2021 (vision2021.ae). Therefore, the institutions' role is crucial to oversee this transmission. This transmission should start by reforming the educational system to achieve the required objectives. The institutions' role is associated with their responsibilities to design and launch innovative programs and initiatives to help the UAE identify and develop more skilled and very well-equipped talents. (National Agenda's Vision 2021)

3.2 The Role of Educational Institutions in the UAE

(MOE) Ministry of Education, (MCKD) Ministry of Culture & Knowledge Development, (ADEK) Abu Dhabi Department of Education and Knowledge, (KHDA) Knowledge & Human Development Authority in Dubai, and (HA) Hamdan Award are five main channels that extend from Government and private level to initiatives level. They implement gifted and talented identification and development programs. The primary research intended to look at how these institutions are functioning in the area of gifts and talents. How do they implement the gifted development programs and to whom this development is applied? The study will provide, for the first time, to the international literature, a local context on these institutions' roles and achievements. Noting while searching for more information and data for this research that the little available literature has focused on the implementation of development programs at schools' level and not at their evaluation or assessment of their achievements. The apparent shortage of research in this area might justify the growing number of similar initiatives and programs. This continuous establishment of initiatives might be assumed as the solution to fill unaddressed gaps. However, these gaps resulted are there due to ineffective planning, researching, lack of evaluation procedures and assessment and missing a policy that regulates the establishment of institutions or initiatives in this regard. This study is a limited exploratory study; therefore, it is an effort to redress the shortage of research in this area in the UAE while providing further illumination to the findings within it.

Table 6. List of UAE Educational Institutions

Institution	Abbreviation	Inception	Emirate	Type
Ministry of Education	MOE	1972	Abu Dhabi	Federal
Hamdan Bin Rashid Foundation	НА	1998	Dubai	Private
Abu Dhabi Department of Education & Knowledge	ADEK	2005	Abu Dhabi	Local
Ministry of Culture & Knowledge Development	MCKD	2006	Abu Dhabi	Federal
Dubai Knowledge & Human Development Authority	KHDA	2006	Dubai	Local

3.3 Sources of Information

Chapter 4 is a framework for the methodology used to support this study. Data used in this study came from different sources. These sources were: (1) library search: Reviewing the institutions' published documents, journals, other researches, books, official papers, e-booklets, and reports. (2) Interviews with stakeholders like experts, teachers, managers, parents of the gifted and talented, and highly able students. The acquired secondary data included policy documents, services, and programs offered online, published annual reports, evaluative documents, and all useful material available from various reports, newspapers, websites, and other UAE official platforms. Fortunately, "HH Sheikh Mohammed bin Rashid Al Maktoum, Vice President and Prime Minister of the UAE and Ruler of Dubai, adopted the Dubai Data Publishing and Exchange Act in 2015, (Law No. (26) Of 2015)" (ogp.dubai.gov.ae) which provides confidential data to UAE's public through an integrated platform. Researchers, investors, and service developers have access to this platform continuously. The law sets forth clear rules and mechanisms that obligate all departments in the UAE to open their data and share them with other departments in an integrated manner. The aim of unifying data is to provide an integrated service to the public as well as decision-makers. Open data is available free of charge and can be used, reused, distributed, or shared with others without restriction. According to the United Nations e-Government study of 2018, open government data means the data disclosed and deployed proactively over the Internet without restrictions(government.ae). Official websites, interviews, document review all generated a rich source of information, despite their lack of different viewpoints of the participants. Data analysis resulted in four subjects that were apparent in the literature, the interviews, the focus group discussions, and the interviews. These themes analysis and discussions in depth exist in chapters five& six of this study.

3.4 An Overview

Before exploring the educational institutions, subject of the study, there will be an overview to have a look at the education in the UAE. It would be an excellent start, to begin with, a brief from the history of education in UAE order to pave the road towards the educational system's current position and its efforts for transmission. The concrete educational progress in the United Arab Emirates occurred in 1962. It started with some (20) schools, fewer than 4,000 students, mostly males. However, and according to (Wekipedia.org 2019-Education in the United Arab Emirates), education in the UAE started in the early 1900s when wealthy pearl merchants in Dubai, Abu Dhabi, and Sharjah, established the first local three schools. These schools staffed with foreign teachers who came from the entire Arab World to teach subjects like reading, writing and Islamic studies. The economic crisis of the 1920s and 1930s forced schools to close. The Japanese competition in the pearl industry incepted this crisis.

Different British schools reopened later in Sharjah in 1953 and Ras Al Khaimah and Abu Dhabi in 1955. Abu Dhabi began making significant profits from the oil in 1960(encyclopedia.com). The UAE emerged after that to transmit from a deprived country to become one of the richest and

vastly growing countries in the world (encyclopedia.com). Many essential social services were not available in its infrastructure in 1962. There were no hospitals, housing, and airports, and there was a significant shortage of workforce. However, the UAE was able to develop its public national educational system within thirty years. It is like what Western countries have established over a hundred years (encyclopedia.com). Few countries in history have experienced the UAE leap and its development shift that happened in less than four decades (encyclopedia.com).

The following summary of the historical systematical growth of the UAE educational system:

- (1) Al-Mutawa's teaching based on memorising the Holy Quran and Hadith, as well as writing and knowing the pillars of Islam.
- (2) The teaching scholars had a comprehensive knowledge of the fundamentals of doctrine, jurisprudence, interpretation, grammar, dictation, history. They held their various religious lessons in the corner of a mosque or the house of the jurist himself or the house of a merchant.
- (3) New or semi-formal education appeared within the period from (1907 to 1953 AD).
- (4) Pearl merchants opened Enlightenment schools in the cities. They brought scientists to manage these schools and supervise the organisation of the lessons and the progress of education.
- (5) The first school in Dubai incepted in 1912.
- (6) The establishment of the first school in Abu Dhabi was in 1930 as well as several schools of similar styles emerged in other UAE cities.
- (7) Evolutionary education continued until the late 1940s, as many factors led to the suspension of these schools. The spread of the economic recession and the emergence of artificial pearls and the impact of the Second World War on trade reflected negatively on these pilot schools.

- (8) The first school in Sharjah founded in (1935). It has developed into a system of formal education.
- (9) By 1964-1965, Abu Dhabi schools were six including (390) boys and (138) girls, (33) teachers staffed them. In Dubai (3572), students in (10) schools and (137) teachers.
- (10) During 2000-2001, 314,217 students, 710 schools, and 27,493 teachers existed in UAE schools.
- (11) The development of formal education occurred in two phases. (a) The first based on the local governments and the knowledge departments during the 1960s. (b) The breakthrough of education took place on December 2, 1971, the day of the UAE's inception.
- (12) The (MOEY) Ministry of Education and Youth incepted in (1972). It oversaw the supervision of education. During that period, government public schools, state-of-the-art equipment, and educational missions from various Arab countries came to help in the development of modern education that is infused by the ongoing advancements in the UAE.
- (13) The current educational system established in the 1970s in the frame of unity. It consisted of four-levels covering the early education for (KG-4-5 years old), primary education for ages (6-11), intermediate education that includes ages of (12-14), and secondary that include teenagers (15-17) levels. They formulate 14 years in total(uaecd.org).

3.5 Gifted Programs

Highly able students need additional gifted programs to satisfy their needs, match their abilities, and contribute to the development of their thinking skills. (NAGC) "organises advocacy for the highly able students at the federal, state and local levels" (www.nag.org). It appeals to other

advocacies who work at home, schools, and communities to support the highly able students and support its efforts to fulfil their requirements. (NAGC) Advocates for these gifted student's "need for day-day-quality programming, a supportive learning environment and effective laws and policies to promote programs and services in which these highly able students thrive" (www.nag.org).

Advocates for the needs of gifted students base their argument on the positive impact of schools and universities when all students, including gifted students, have opportunities to make continuous progress and reach their potential. The impact also reflects in imposing changes on policies, practices and laws that affect the education of gifted students(Purcell, J.H. and Eckert, R.D., 2006). Consequently, a guidebook to inform the educators about gifted education and how to design services and programs for high-ability learners composed by (Purcell, J.H. and Eckert, R.D., 2006) included different authors who shared their articles and their experiences and knowledge on the effective gifted programs and how to effectively advocate for gifted students' requirements. A story provided in one of the guidebook articles, cited an example of effective advocacy reflected in the attitude of a mother of a talented young girl who moved recently to a poorly serviced district that has no gifted programs in their school. This mother, who was concerned about her daughter's needs was aware that she had to advocate and prove that gifted students, including her daughter, need additional programs to thrive. This mother succeeded in launching a campaign after spending one year studying and searching in the subjects related to gifted children and school. Her first success was in building a new playground for the school children. However, she joined a group of advocates for gifted education programs and the need to implement them in schools. Through this group, they were able to educate school and district administrators about the need to create a suitable program for gifted students. These defenders succeeded in reaching their goal (Purcell, J.H. and Eckert, R.D., 2006).

3.6 Definition of Gifted Programs

Gifted programs are tools used by teachers or educators in schools or community centres to educate those students with high abilities and gifts. These programs are supposed to be challenging to these students and enable them to progress in schools and life continually. These programs can be in different types and forms such as enrichment or accelerated. Highly able students need gifted programs because general education is not able to provide them with their needs to thrive. These programs influence students' futures by effecting their plans according to many studies that reported the impact of these programs on the gifted students and their achievements. Students who participated in gifted programs could maintain their interests and talents over time, and they continued motivated within their creative, productive work after school and university graduation.

3.7 Gifted Programs Requirments

"High-quality gifted programming should be planned, maintained and evaluated carefully (www.nag.org). According to (NAGC) practical programming standards, there are seven criteria to consider when planning and maintaining and evaluating the gifted programming and they are: "Program Administration and Management, Design, Evaluation, Socio-emotional Guidance and Counseling, Professional Development and Student Identification(nag.org).

According to (VanTassel-Baska, J., 2009) The services and education programs of highly able students should be consistent with the tools and the assessment methods used to screen students in stage one. The provided gifted programs should be regulated. The regulations include having an

accurate definition for each program and the service components and observing the following controlling guidelines:

- (1) Gifted programs must include both the grouping and the resource room, transferring classes and individual classes. At least to provide one of these options in order to ensure the delivery of appropriate service to the gifted student.
- (2) Programs and services duration should be at least (150) minutes per week with at least one hour per week for teacher planning.
- (3) To modify the curriculum in a way that makes every subject in it related to gifted students. They must include pace, complexity, depth, challenge and creativity based on specific criteria that require the development or use of a curriculum tailored to the educational needs of gifted students.
- (4) The measurement procedures used in gifted programs are matching the specific curriculum objectives including Standardized tests, performance-based tasks and personal portfolio datasheets.
- (5) Develop an expanded program to accommodate gifted and at-risk students.
- (6) Accelerated education for gifted students should be a permanent option in the gifted education program.
- (7) Include emotional and social support for highly able students' development as part of the service delivery plan.
- (8) Provide academic guidance and career guidance for ages from six to twelve with the need to participate in advanced courses and the use of student assessment data to guide them in terms of options for college and post-graduate.

- (9) Establish a local advisory council that provides guidance on the plan for the provision of local or other services at the state level and is approved by the local educational institutions or councils.
- (10) Develop complementary policies related to gifted services and program delivery (VanTassel-Baska, J., 2009).

3.8 Gifted programs in the UAE

Ever since 1979, the educational system in the UAE depended on the (MOE) to oversee the different educational needs. UAE unity infused a unique collaboration among the seven Emirates by employing regulations and various plans and strategies (AlGhawi, M.A., 2017). It managed that with the aid of national planning and policy integrations in many areas. However, and according to what the unified UAE School Inspection Framework indicated; each emirate advanced its educational reform in its approach. The inspection framework supported the inspection of all schools in the UAE that provide education for students within (KG-12) (moe.gov.ae). The structuring of school inspections depends on six performance standards:(1) achievement ;(2) personal development;(3) Teaching;(4) Curriculum;(5) Protection, guidance and support;(6) management and Leadership (moe.gov.ae). The definition of special Education is according to this framework, interprets it as the education provided for people with a disability or any recognised disorder named in the UAE as the "people with determination". (moe.gov.ae) "Therefore, the presentation of special education programs in the UAE only accepts and applies to individuals with disabilities. However, gifted and Talented individuals 'education might take place if the (MOE) develops a specific national policy to unify the gifted and talented definition, identification, and development services. Special education viewpoint according to the (School for all), that every student is a unique individual and needs to be surrounded by a thriving environment that keeps

him or her safe, takes care of his or her abilities and support his or her needs to thrive. This gifted education should look at the highly able students as the most valuable investment who need all types of support. It should also aim to provide them with the kind of learning that helps them study in a challenging environment to thrive. It aims to provide them with regular constant practice and equitable development, too (moe.gov.ae). The UAE undertakes using all available sites and sources to plan and implement the development programs that ensure each student has an Individual Education Program (IEP) or an Advanced Learning Plan (ALP) according to (the General Rules for the Provision of Special Education Programs and Services in private & public schools). Consequently, the role of the UAE educational institutions is to satisfy the country's potential and requirement. They should ensure implementing a proper education for all and to apply the appropriate identification and development programs and procedures to identify and nurture gifts and talents.

The development of gifted education is not only a new process in the UAE, but it is a relatively new strategy in the globe (nagc.org). However, the UAE is witnessing an increasing public awareness by the power of continuous exchanged debates in cultural and educational forums which resulted in a corresponding proliferation of gifted programming in the UAE. The definition of such programs is enrichment activities provided after school, in summer camps, at government educational initiatives, and other offered programs. These varieties of programs have sparked controversies and debates regarding what are they providing and if they are benefiting the highly able students'; how are they nurturing their potentials and to which category they should provide services. According to an international study, various facets contribute to the success of these programs such as optimistic expectations, focusing on mastery, cooperation, positive relationships amongst students and teachers, and embedding school elements within them (Martin,2005).

Therefore, this requires establishing policies and strategies that regulate these enrichment programs, or any other programs aim to form part of the gifted education.

Finally, gifted education has much to offer the UAE National Agenda. It helps the country face new challenges, meet their requirements for advancement in the 21st century and boost emeriti youth's future ambitions(vision2021.ae). Therefore, all controversies and underlying issues need thorough examination, such as the debates regarding the definition, identification, evaluation, and assessment of the identification and the development programs. Learning more about the UAE educational institutions and their role in the development of this category helps understanding how many efforts are collaborating to support this field and what are their methods in doing so.

3.9 Identification of the gifted and talented in the UAE

According to the General Rules for the Provision of Special Education Programs and Services in private & public schools and the Inspection Framework (moe.gov.ae), identifying and distinguishing between the highly able students and the other learners is not an isolated activity, but it drives through a range of methods. These methods are represented in a variety of tools and styles such as an accurate observation of the students; conducting parents' and students' interviews; using teachers 'screening checklists; applying student's surveys; using self-reports; standardised tests and assessments of cognitive development and ability that are conducted explicitly by psychologists. Lately, the (MOE), Ministry of Education, has ratified a cooperation agreement with the Hamdan Foundation in 2018 (moe.gov.ae). This agreement was intended to implement the Identification of talented students in many schools across the UAE. The identification is limited to only students in grades 4-8. (moe.gov.ae, 2019). Accordingly, highly able students will enrol in specialised development programs created for them. These programs

are supposed to provide the identified students with services that will be able to meet their requirements in line with the UAE National Agenda. (MOE, gov.ae, 2019).

According to (ha, 2019), the identification stage is the essential stage that guides and controls the highly able students' enrollment in the appropriate development programs(ha.ae). Once determining the type and level of talent, they will enrol in the suitable programs to nurture their needs. Giving them advice and guidance will help them receive the type of education and development that most suit their talents and gifts (ha.ae). The followed criteria in the identification process include the IQ tests and academic achievement as well. Quantitative tools used by (NPGP) include (1) measurement of the mental abilities, (2) Identification Kit created to identify gifts, (3) Achievement Tests used to estimate their academic abilities, and (4) Testing how candidates creatively solve the problems. Qualitative tools used by (NPGP) include (1) Portfolio, (2) Note Cards, (3) Behavioral Statements, (4) Personal Interviews, and (5) List of professional interests. The (NPGP) in (HA) has developed an Identification Kit to identify the academically gifted students by measuring their diverse abilities. This identification kit was developed in collaboration with international experts and designed to commensurate with the UAE environment.

Table 7. Three stages of the identification process by (NPGP)

NPGP	Screening	Narrowing	Inclusion	Quantitative	Qualitative
	It depends on	Identification of	Interview the selected	IQ test -Academic	Portfolio
	teachers,	nominated	students after the	achievement test	Note Card
	parents, peers,	students.	narrowing stage to		Behavioural
	or self-		assign them to		Statements
	nomination.		educational programs.		Interviews
					List of professional
					interests and preferences

(ha.ae)

The advanced identification portfolio recently developed in 2019 is to care and look after each gifted student. A scientist is responsible for monitoring the gifted child throughout his educational

progress. It will tailor a development plan for the gifted student through his life journey according to (ha.ae).

3.10 Assessment of progress

The (MOE) inspection framework describes the assessment of the highly able students as an ongoing Pre and post-assessments to help teachers determine what students already know in order to provide them with more challenging programs. (moe.gov.ae) However, the post-assessment is not clear in the inspection framework, and it is not evident if educators should use the assessment before or after the programs in order to differentiate the curriculum, the teaching instructions, and to monitor the gifted and talented student's advance performance and development. The results of both assessments should enable the teachers to evaluate the programs and be able to amend and improve them as required.

3.11 UAE Educational Institutions

Background information regarding the five institutions chosen for this study is included in the next section to provide a context for gifted education and programs provided by these authorities. According to the latest update posted by the UAE Government on December 16, 2018, on the (government.ae) official website, there are two levels of authorities concerned on education in the UAE, (1) Federal level and (2) local level.

At the federal level

3.11.1 The Ministry of Education (MOE)

The Ministry of Education (MOE) oversees education through all phases and stages in public and some private schools in the country, including schools and colleges. Higher education is also within its field of supervision since 2016. The Ministry aims to build and manage an educational and innovative system for a highly competitive knowledge society that includes all age levels and meets the needs of the labour market(moe.gov.ae).

At the local level. Emirate of Abu Dhabi

3.11.2 The (Abu Dhabi Department of Education and Knowledge) (ADEK)

The establishment of (ADEK) in 2005 to make it responsible for overseeing development initiatives in all sectors of education from the foundation and secondary levels to higher education, technical, vocational, and education for the older students in Abu Dhabi.

Emirate of Dubai.

3.11.3 Knowledge & Human Development Authority (KHDA)

The (KHDA) established in 2006 to become responsible for monitoring private education and schools in Dubai (government.ae). KHDA, on the other hand, it inspects the quality of teaching, training, and other elements of schools' inspection in Dubai. It also works to provide different types of support for students, schools, universities, and relevant parties in this area. It includes several institutions, involving the school education institution and the school supervision system in Dubai.

Emirate of Sharjah.

3.11.4 Sharjah Special Education Authority (SSEA)

In November 2018, the Government of Sharjah issued The Emir Decree No. 45 of 2018 to establish Sharjah Special Education Authority in the Emirate. The Authority follows the Central Government of the Emirate of Sharjah and enjoys the legal and civil personality necessary to carry out all acts and legal acts to ensure the achievement of its objectives.

3.11.5 In the Northern Emirates

The (MOE) oversees schools in the northern Emirates as well. Schools in Sharjah, Ajman, Umm Al Quwain, Ras Al Khaimah, and Fujairah are all under supervision of the Ministry of Education (MOE). Each district in each Emirate is responsible for supervising schools within their area (government.ae).

3.12 Education in the UAE

Present & Future

3.12.1 The UAE National Agenda

The UAE National Agenda launched to achieve the (UAE Vision 2021) when it is the UAE golden jubilee celebration in 2021 (vision 2021.ae). Education is a priority for the UAE as one of its essential focuses and objectives. The (MOE) according to this vision, has developed the Education 2020 Strategy that includes five plans for the intended educational improvement in the country. It aims mainly to improve the educational system and the style of teaching and learning (government.ae). The (MOE) strategy of (2017-2021) for all age groups is to "develop an innovative educational system to support the building of a knowledgeable and globally competitive

society" (government.ae). This strategy based on values such as equality and justice, science technology and innovation, commitment and transparency (government.ae).

Curriculum development planned from (2015-2021). The (MOE) implemented significant improvements and reforms in the curriculum to prepare the students for the future. This curriculum focuses on developing the student's skills and strengthening their abilities along with their national identity and innovation. Education will be according to (MOE) in two tracks, general education and advanced education. Such reforms would allow the students to meet the needs of the job market and facilitate the direct enrolment of the high school graduates in the international universities without the need for a foundation year (government.ae).

Therefore, eight objectives determined to be on this agenda as the primary factors and the sectors that would lead the UAE to be among the most prosperous countries in the globe. Providing an excellent education is one of the main objectives. These purposes target essential areas in education related to the development of students, teachers, and school leaders" (www.moe. gov. ae).

Currently, numerous educational authorities in the UAE oversee, support, and elevate the UAE's intended strategies for an excellent education. The leading institutions, the focus of this study, are responsible for setting and developing education policies and strategies for schools in the country. They are responsible for the gifted and talented development as well.

Therefore, this chapter will look deep in these institutions concerning their role in implementing the identification and development programs for the highly able students. The exploring approach will be in case studies methodology. All information obtained from and about these institutions will follow a form that makes it possible at the end of the chapter to discuss and compare their roles towards these extraordinary students in the UAE.

3.13 First: (MOE) Ministry of Education

(Information accessed through the open data provided by (government. ae)

Ministry of Education (MOE) responsibility is to elevate and implement development, improvement, and radical changes in the educational frameworks in the UAE. Its philosophy is in line with modern educational philosophy. It builds an educational system for the whole country based on establishing the skills of the 21st century through the curriculum, interactive activities, and scientific exploratory trips. It enhances innovation in UAE schools. The Ministry furthermore works to adapt its curricula according to the requirements of the labour markets in order to keep up with the country's plans. The Ministry has provided a path for vocational education through the establishment of vocational and technical institutes.

The (MOE) establishment reflected the values of the UAE that appears in the strategic plans, organisational structures, and development initiatives that achieve the indicators of the national agenda. The ministry accordingly has exerted enormous efforts to sharpen and strengthen the youth skills from an early age. The (MOE) vision that links with the UAE's vision was to provide these gifted and talented with the required skills to enter the future labour markets. It provides the best systems for intelligent learning by working on two tracks. The first track is to deliver the necessary infrastructure, rehabilitation, and development of youth skills through the training portal. The second track is to enable students through specialised curricula and Pioneering activities that mimic the future and equip young people with the skills of their time.

Table 8. MOE Vision & Mission- Strategic Objectives 2017-2021 (government.ae)

MOE Vision	MOE Mission		
Innovative education, a pioneer in knowledge and a	Developing an innovative Education System. Enhancing		
leading global society.	knowledge and globally competitive society, including		
	all age groups, to meet labour market demands.		
	Ensuring the quality of the ministry of education		
	outputs. Ensuring the quality of the (MOE) provision of		
	best services for all internal and external customers.		
Ministry of Education Strat	regic Objectives 2017-2021		
- Ensure implementing equal education, including pre-	-Attracting and qualifying students to enrol in higher		
school level.	education institutions, commensurate their skills with		
	the needs of the labour market		
- Achieve outstanding efficiency of the leadership and	-To foster a culture of innovation in the institutional		
educational bodies.	work environment		
- Ensure implementing a quality, efficiency, and	- Enhance scientific research and innovation capabilities		
governance of educational and institutional performance	according to globally competitive standards		
- Ensure safe, supportive and stimulating learning	- Ensure the provision of management services		
environments.	following all standards of quality, efficiency, and		
	transparency		

Self-education is the legitimate form of education practised in the emirates of the United Arab Emirates. It started long before the existence of unity, while educations are the means of transferring to a skill-based knowledge. The transmission of education from a traditional style and method to a further modernized style undertook four iterations of development including the "Mutawwa" who is the old name of the teacher who used to teach in the traditional classes named "Katateeb", (2) The improved educational Circles, (3) The organized systematical Education, and lastly (4) The Modernized Educational System (Alnabah, 1996). In (2008), the (MOE) implemented the primary education system in the country to organise public education. This system involved the 14 years of education starting from (KG-12)(Alhebsi, Pettaway &Waller, 2015). Currently, UAE established a wide range of public and private schools for pre-university education that consists of: (639) public schools with the total of (287,725) students, (23,146) teachers, (580) private schools with a total of (793295) student, (46,870) teachers from (187) different nationalities. Higher education becomes within the (MOE) field of supervision since 2016. It aims to build and manage an educational and innovative system for a highly competitive

knowledge society that fulfils the needs of the graduates and open the doors for them towards the labour market (gov.ae.ae-open data).

3.13.1 Emirates Foundation for School Education

The year 2016, witnessed the establishment of the Emirates Foundation for School Education as an independent organisation to facilitate public-private funded initiatives that aim to improve the welfare of youth across the UAE. The foundation, following independent powers under the Council of Ministers Decision No. 8. Established to promotes the education sector and its efficiency. It provides schools with the required education and implements policies, strategies, standards, and controls related to the education sector, including vocational, technical, and continuing education(emirates foundation.ae).

3.13.2 Development of Gifted and talented

Special education in the (MOE) is only concerned about providing the disabled individuals a unique environment that helps them grow and mature socially and emotionally. It comes within the (MOE) objectives listed in the School Inspection Framework and the General Rules for the Provision of Special Education programs and services in the UAE. The (MOE) role is to provide highly able students with services in all public and private schools. It is also responsible for providing schools with qualified professionals to educate these unique students. The (MOE) as well must exert efforts by implementing an ongoing professional development for the teachers. They need consistent training by the best internationally acknowledged practices and education in the areas of special education (moe.gov.ae).

The (MOE) according delivers various services to these highly able students, including development programs and any other enrichment courses or training given to them inside or outside the regular classes. Some other services, assigned to the (MOE) involves giving students assignments as individual projects and studies to challenge their abilities. Another method involves providing them with specific educational plans and lectures, along with debates to sharpen their knowledge and skills. Enriched Activities involve independent studies and employment guidance (moe.gov.ae).

The gifted student that needs gifted education according to (MOE) is the one "who exhibits willingness, ability, or performance in the fields of (STEAM). Performing arts like painting, music, and sports comprise the general framework for highly able students and what identifies them according to the (MOE) standards and requirements. Development cognitive skills include" critical and creative thinking, decision-making, and problem-solving, and personal and emotional development skills". These skills develop through personal self-skills, learning skills, listening, observation, the use of research skills, reference materials, research projects, the use of a desk and electronic references, and the use of community resources(moe.gov.ae).

3.13.3 MOE Definition & Identification of Gifted and Talented

Gifted students, according to the "Differentiation Model of Giftedness and Talent" refers to the "student who is untrained and spontaneously expresses exceptional natural ability in various areas of human's capacity"." Giftedness includes "intellectual, creative, social and physical ability" (moe.gov.ae). These students' display exceptional potential, though, they might underachieve. The talented student is "the able student to convert his or her "giftedness" into performance and utilise it." Talented students usually demonstrate exceptional levels of competence that shines in the

particular domains of human ability." (Inspection Framework (moe.gov.ae). The students' identification is an isolated activity that aims to define students with extraordinary abilities. However, the identification stage is the first stage. A range of other following provisions will provide these highly able students with customised learning programs in the next stages. Certain aspects are taken into consideration when preparing these developmental programs. They should follow an individual plan to provide these gifted and talented students with their equal education according to the gifted education international standards (moe.gov.ae). These aspects comprise of diversity in methods and programs, pace, ongoing assessment, groupings, the level of work, and enrichment programs.

3.13.4 Example of a development program at (MOE)

(1) Masterpieces Program

In 2017, the Ministry of Education collaborated with local and international partners to work on the "Masterpieces initiative". It aims to evolve and improve students' talents and capabilities in specific fields of arts. Its goal is to uncover students' talents in six fields of arts and culture such as Music; Theatre; Visual Arts; Culture; Cinema, and Poetry. This initiative, according to (moe.gov.ae), provides specialised workshops and training seminars to help develop and nourish students' talents and skills through local and international professionals. The Masterpieces received the number of (1300) applications for registration in 2017 that marked its first launch. However, the jury ended up identifying 183 students as talented. (al bayan. ae, 2017)

(2) Our Ambassadors Program

The establishment of this initiative was to employ active, talented students who can represent the UAE talents in the areas of art, music, theatre, and heritage. In 2018 it launched its third round. It

targeted (1080) students from both public and private schools, universities and teachers. The (MOE) developed eight programs included within this program: Ambassadors for the Future, Innovation and Giving programs, the Education ambassadors Program, Ambassadors of Diplomacy, Sports Excellence, Masterpieces, and Higher Education Ambassadors Program for university students. Participants in these eight programs will travel to visit community service centres in 10 destinations in the world within the US, Europe and Canada(moe.gov.ae).

Talents Identification for Masterpieces

To enrol in "Masterpieces" programs, talented students must register through the (MOE) website and wait for the jury decision. This jury is responsible for identifying the eligible participants in this program. There is little information regarding the sequence of the identification process or the development programs displayed on the (MOE) official website.

3.14 Second: (MCKD) Ministry of Culture & Knowledge Development

(Information accessed through the open data provided by (government. ae)

Table 9. MCKD Vision & Mission

MCKD Vision					
The UAE should be a knowledge hub and a pioneering model that cherishes its national identity and culture around					
the globe.					
MCKD Mission (mckd.gov.ae)					
To promote the UAE culture and to preserve its national identity through the empowerment of creativity and					
innovation.					
Persevere the national cultural identity. Enable the UAE cultural sector.					
Enhance the contribution of the cultural and creative	Strengthen the UAE's position on the global cultural				
sector to support developing a sustainable knowledge- map.					
based economy.					

(mckd.gov.ae)

The Ministry first incepted in 1972 in the name of the Ministry of Information and Culture. It was under the Interim Constitution of the United Arab Emirates in Article (58, the Federal Decree No. (1) for the year 1972 on" the terms of reference of ministries and powers of ministers." In 2006, Decree No. 1 of 2006 amended some provisions of Federal Law No. (1) For the year 1972. The Ministry of Culture, Youth & Community Development created to replace the Ministry of Information and Culture. In February 2016, a decision issued by the Council of Ministers to add some competencies and change the name to the Ministry of Culture and Knowledge Development (mckd.gov.ae).

3.14.1 Gifted and talented identification and development

Policy for Talented

The UAE Government assigned to the (MCKD) the development of a national policy for talents' identification and development in the UAE. The seven main themes adopted within its tasks intended to start the achievements of the first cultural agenda of its kind. It represents the strategic framework for the development of the cultural sector in the UAE (2018 to 2031). The policy aims to support the development of talented people in cooperation with competent authorities basis on the differentiated needs of talented individuals in the fields of Arts & Culture. The policy is a regulator for discovering, supporting, training, and motivating this category to produce creative industries. Such a policy is a crucial support for talented people in the cultural sector in the UAE. The (MCKD) held several workshops and brainstorming labs in collaboration with the concerned institutions, authorities, and other stakeholders to oversee the policy needs and support its formulation (mckd.gov.ae).

Table.10. (MCKD) Talents' Nurturing Policy Themes

Identification	Support	Train	Motivate	
Talents' unified definition	Talented individuals need	The policy develops a plan	To motivate talented	
that reflects the concept	that reflects the concept a differentiated supporting		persons in the sectors of	
of talent in the cultural	system. This policy	and sharpen their skills. It	Arts and Culture.	
sector, including all	regulates and facilitates all	provides diverse and	To create a thriving	
categories of talents,	types of support required	sustainable quality	environment.	
develop a plan for	and needed by these	training programs within		
discovering and	talents.	and outside the UAE for		
implementing		all talents in all domains of		
development programs		Arts and Culture.		
and establish a gifted and				
talented comprehensive				
database.				

3.14.2 Example of an (MCKD) Development program

Nurturing Talented Authors

This program is a practical application for The "National Reading Strategy "launched and adopted by the Ministry of Cabinet Affairs in the UAE. The program responses to the issue of weakness of reading in the Arab world and the UAE. (O'Sullivan, 2004). The strategy launched in May 2016 - 2026. The launch of the program was in 2017. It aimed to support a promising generation of talented writers and authors in various fields of literature. Talented writers need support and cannot make it by themselves, according to the (MCKD). Talents' support within this program takes a systematic and thoughtful form. The program aimed at enhancing the value of reading. It motivates young people to write and produce literature. The Ministry seeks to involve all talented people in the program. It, therefore, built the program in cooperation with universities, schools, and local authorities in relation. This program is an integrated curriculum that includes several authors, academics, and supervisors from different domains of talents. It includes a set of basics of literacy curriculum, writing theories, workshops, training, re-reading, and drafting of writings in the presence and supervision of a group of illustrious writers who will reveal their writing methods

and secrets. The program also provides an opportunity to translate the book into several languages and to promote the winner internationally (mckd.gov.ae)

The Nurturing Talented Authors program is for the young writers who display talents in all types of literature. The program has two levels: Beginners and advanced writers. It constitutes an integrated literacy curriculum prepared with essential experiences of training and education. It targets school and university students and creative writers from the age group 15 to 35 years. The cultural centres, traditional media, social networking sites, schools, and universities announce the program and call for participants. Applicants must be UAE nationals. The submission of the filling will be through the MCKD website. The application is subject to approval by a jury. A committee forms to investigate all applications that will nominate the suitable and eligible participants according to the sample extracted from their writings. Another way of identification will apply during the «Cultural Summer Camp." The formulated committee by (MCKD) will observe the participants during the program through all the emirates. The committee identifies several talents, listens to the views of their peers, and gets recommendations from the camp supervisors. The committee discusses the possibility of developing these nominated talents during their attendance in the camp (MCKD.gov.ae); (al-Ittihad. ae, 2017)

The development process constitutes a series of practical writing workshops.

(1) Training on publishing and distribution. (2) Training on communication and marketing. (3) Acquiring valuable critics' opinions. (4) Communicate with skilled writers. (5) Communicate with auditing and proofreading. (6) Communicate with nature and learn how to reflect. (7) Get support to become a member of literacy associations. (8) Getting the translation privilege to be able to communicate with the world. Furthermore, the program paused currently for essential

improvements. It will undergo a thorough transformation to convert to a digital online development program.

3.15 Third: (ADEK) Abu Dhabi Department of Education and Knowledge

Table 11. ADEK Vision & Mission-Strategic Objectives (adek.gov.ae)

ADEK Vision					
They are providing a world-class education system that will thrust all students to take full advantage of their					
potential and abilities to qualify them to compete globally	y.				
ADEK Mission					
Graduates of world-class students, possessing a keen cult	tural and heritage awareness, and qualified to meet global				
challenges.					
The core values of the Board at ADEK are the develop	ped rules and principles that will ensure the continuous				
improvement of performance levels.					
ADEK Strategic Objectives					
To develop the strategic and executive plans for Supervision of the education sector following internal					
education in Abu Dhabi. The approvals received from	regulations, legislation, and conventions in force locally				
the Executive Council that supervises their	and internationally, in cooperation with the concerned				
implementation as well.	authorities.				
Monitor the commitment of the entities in the	To license all private educational institutions in Abu				
Department of Education and Knowledge in	Dhabi and monitor their commitment to providing high-				
implementing strategic and operational plans for the	quality educational services following the existing				
education sector. regulations. Collaborate with the relevant					
Supervising all the departments of the Education and	Provide recommendations and suggestions regarding				
Knowledge Department concerning their revenues and	fees and approval by the Executive Council following				
expenses, and the extent of their commitment to their established procedures.					
terms of reference and their approval of the strategic and					
operational plans.					
To regulate the education sector through the Prepare periodic reports and analytical studies related to					
development of policies, standards, regulations, and the competences of ADEK and submit them to the Abu					
executive and operational decisions. Dhabi Executive Office.					
Other functions or functions assigned to it by the Executive Council.					

On September 10, 2005, His Highness Sheikh Khalifa bin Zayed Al Nahyan, UAE Supreme Commander and Commander-in-Chief of the UAE Armed Forces issued Law No. 24 of 2005 establishing the Abu Dhabi Education Council (ADEC) (taleem.ae).

In September 2017, H.H Sheikh Khalifa bin Zayed Al Nahyan, President of the UAE, issued an Amiri Decree to change the name of the "Abu Dhabi Education Council" to the "Department of Education and Knowledge". Accordingly, Law No. (9) Of 2018 on establishing the "Department of Education and Knowledge", to replace the "Abu Dhabi Education Council" (taleem.ae).

Under the law, the primary role of the Department is to develop the education system in the Emirate of Abu Dhabi, to foster a culture of creativity, excellence, and sustainability and to focus on humans' development, and their social, and economic capabilities as well(taleem.ae).

3.15.1 The role of (ADEK) in Education

The Department of Education and Knowledge has been responsible for the management, direction, adoption, and implementation of developments in the education strategies in the Emirate of Abu Dhabi(adek.gov.ae).

3.15.2 (ADEK) Gifted & Talented Development Initiatives

3.15.2.1 Definition of gifted and talented

A talented student is an individual who possesses unique skills and abilities in areas such as speech, poetry, painting, handicrafts, sports, or theatre(adek.gov.ae) The individual's high ability to possess or have significant abilities in one of the areas that highlight intelligence, creativity, or academic excellence. The talented student usually performs exceptionally and needs circumstances to enable him to meet his educational needs in the regular classroom. Teachers as well provide these highly able students with support and their essential needs from education(adek.gov.ae).

3.15.2.2 Development Programs & Initiatives

ADEK development services include enrichment programs in regular classrooms or educational resource rooms, assignment of students to individual projects and studies, lectures, discussions, or seminars and to participate in educational competitions. Consequently, (ADEK) has launched several initiatives to develop the educational system in Abu Dhabi and help talented and gifted students obtain the necessary skills and abilities to participate in the socio-economic development of Abu Dhabi. (ADEK) Ratified a joint agreement with the "Special Education Department" at the Faculty of Education at (UAEU) University of the United Arab Emirates (cedu.uauu.ac.ae). The program includes a professional development program for teachers who work in Abu Dhabi Public Schools. This program is in cooperation with (MOE) in order to identify the highly able students and nurture them. The program targeted (55) male and female teachers in various disciplines. This program aims to develop the capacity of Abu Dhabi schoolteachers to teach these students with high abilities. The program enables them to identify these extraordinary students in order to provide them with alternatives and the unique experiences that respond to the particular requirements of the country. The program enhances teamwork skills and spirit through various activities and workshops. Teachers participating in the program get educational film samples, guidance on how to implement the strategy, educational activities, assignments, duties and tasks, evaluation criteria for lesson plans, final projects and papers, articles of impressions, and personal opinions, as well as additional sources(Adek.gov.ae).

3.15.3 An example of a development program at (ADEK)

3.15.3.1 Mawhibaty Programs

(ADEK) Launched Mawhibaty program in collaboration with "The Department of Culture and Tourism in Abu Dhabi" (DCT). This program aims to nurture talented students who exhibit abilities in the area of visual and performing arts. This program is to all talented students in arts from grades 1 to 12 in public and private schools in Abu Dhabi. The program aims to discover and develop these students who are outstanding and possess talent and desire in the fields of visual and performing arts. Talented students get support from specialised technical centres throughout the academic year. Talented students fill out an online application form Parents or teachers of talented students may register the talented student too. The application will not go through unless the electronic form contains samples of the student's work (artboards, audio recordings, and video clips). A team of experts and specialists will arbitrate these applications and nominate all students who are eligible to join the program. The program includes a clear plan of action that contains all the expected outputs, such as the timeframe, required resources, and the development of the identification tools. The design of the program follows a model for evaluating potential candidates. (ADEK) Electronic platform for the registration of talented students in English and Arabic from grades 1 through 12. The electronic platform is open for the enrollment of any talented student. After filling the electronic application form, a team of experts will evaluate all applications and communicate with eligible students for further evaluation and interviews. Art experts will perform the selection procedure. The two phases of evaluation and selection are (1) the arbitration of students' works of art (visual and performance) by a group of specialists from each of the "Department of Education and Knowledge" and the "Department of Culture and Tourism" in Abu Dhabi. (2) Interviewing the nominated students to allocate them in workshops conducted by local

artists and specialists. (ADEK) has already organised workshops for (600) teachers in Abu Dhabi schools to develop their professional abilities and skills to identify talented students and design talent development programs in specific domains of talents(Adek.gov.ae).

3.16 Fourth: (KHDA) Knowledge & Human Development Authority

Table 12. KHDA Vision & Mission-Strategic Objectives

TZTTT	^	T 7.		
KHI	JΑ	V 15	sio	n

An integrated educational system.

KHDA Mission

To improve the quality of education and access to knowledge and human development through community partnership(khda.gov.ae).

Strategic Objectives

To improve education and its outputs in all its stages and forms in line with the economic and social development witnessed by Dubai and the UAE, thus contributing to the transition to the knowledge economy.

To launch a range of projects and initiatives that will benefit all stakeholders in the education and human development sectors, from students, parents, teachers, and institutions of both" public and private sectors".

Access to knowledge, education, learning, and human development through community partnership.

The (KHDA) was established by (Decision 38 of The Dubai Executive Council in the Government of Dubai in 2007) to be responsible for elevating the quality of education in private schools in the Emirate of Dubai. Its responsibility is to provide support to all stakeholders and sectors related to education in Dubai. These include "schools, universities, students, parents, educators, investors and government partners". Its role is to collaborate with them to develop a high-quality education sector that promotes happiness and good living. A Supervision Agency established in the (KHDA) issued by the (Dubai Executive Council following Decision No. 38 of 2007), to provide comprehensive information on education standards in Dubai and to support planning for school-level development and policies. Several methods, including annual school inspections, meetings

with parents, and dialogue groups, as well as questionnaires aimed at teachers, parents, and students, will help (KHDA) to achieve that(khda.gov.ae).

3.16.1 The Role of (KHDA) in Education

(KHDA) Provides a comprehensive and reliable School Monitoring System to assess the standards and the performance of private schools in the Emirate of Dubai (khda.gov.ae). The educational standards are developed and regularly reviewed by the (KHDA) in collaboration with qualified educators and supervisors for this mission. These educators represent a broad spectrum of the different educational curricular in Dubai. (KHDA) Seeks to participate in the international assessments with the support of the school supervisory body and through school inspections. It collaborates with schools to provide Government of Dubai, and many other stakeholders a detailed and accurate study of the quality of Dubai schools' education compared to the performance of education in the world-leading countries(khda.gov.ae).

3.16.2 KHDA definition & Identification of gifted and talented

(KHDA) Adopts the (MOE) definition for gifted and talented. The definition considers the "Differentiation Model of Giftedness and Talent (moe.gov.ae)." The definition of giftedness refers to a student with new and spontaneously exceptional natural abilities. These abilities are displayed and expressed in one or more domains of human ability. They include the intellectual, creative, social, and physical abilities(moe.gov.ae).

A talented student refers to" a student that could transform his or her giftedness into exceptional performance through hard work and within the appropriate environment." The identification methods depend on a range of tools indicated and specified by the (MOE) Rules and provisions

handbook. The methods administered by psychologists include accurate observations, interviews with parents, and interviews with students with higher abilities, teachers' checklists; student interest surveys; self-reports, in-depth student's interviews, standardised tests, standardised assessments, evaluation, and assessment of cognitive development, and ability (khda.gov.ae).

3.16.3 (KHDA) Gifted & Talented Developmental programs

3.16.3.1 Examples of (KHDA) development program

(1) Human Talent Map

(KHDA) Works currently on creating a human talent map in Dubai. The first phase of the project named "It is me," targets government employees in the public sector. This smart platform expected to launch by the end of 2019. It aims to enable every employee in Dubai to document his/her talents following specific government accreditation standards. The participants will document all their accomplishments, such as certificates, experiences, and accreditations, or any other appreciation or honours received. It will also include any lessons, courses, and conferences attended, and the development of any plans commensurate with health and physical condition. The map includes documentation of skills and experience gained, training programs, conferences, courses attended, and recognition. Through this platform, it is possible to identify the skills of the employees and their needs in the future. It aims to become the employees' lifelong educational support. The second phase of this project will include students and retirees to make use of their talents, keep them up-to-date, and guide them to strengthen the areas where they need more training and skill-building.

(2) The National Institute for Vocational Education (NIVE)

This "National Institute for Vocational Education" (NIVE) established in (2006) by the guidance of H.H. Sheikh Mohammed bin Rashid Al Maktoum, the Vice President, Prime Minister, and Ruler of Dubai. Its responsibility is to provide students with skills in the field of vocational education. It provides them with the opportunity to develop specific vocational qualifications that enhance their access to the right jobs in the future(nive.gov.ae). The institution that is an affiliation to the (KHDA) offers a range of programs that comply with the latest local and international standards. All these programs, locally accredited by the (NQA) the national qualifications' authority in the UAE and internationally from Pearson-BTEC in the UK represent a new and clear vision for education, which powerfully highlights the importance of vocational education to lead the future of the economy in the UAE(nive.gov.ae). This vision links to providing a knowledge base, creating a competitive spirit, and supporting economic development that can keep the UAE economy progress with the global economy. This vision also aims at promoting vocational education and providing real opportunities for UAE nationals to be able to develop their skills and match them with the required jobs in the market(nive.gov.ae). To serve this objective, the National Institute (NIVE) adopted a clear structure for vocational education and its future, through which it provides a gradual and clear pathway. The various levels in this education accept and qualify different levels of students who would contribute effectively to support the new economy in the UAE.

The Institute aims to build a highly qualified and flexible workforce that is the key to innovation, productivity and competitiveness amongst the levels of the Arab nations and the globe. Its basis on (1) Providing the labour market with qualified labours. (2) Achieving the ambitions of UAE talented nationals. The basis for the selection of knowledge and skills acquired by the students at the Institute was to enhance their abilities and career development and to meet their needs.

(3) Young editors

(KHDA) has collaborated with Gulf News to give students in private schools in Dubai the opportunity to publish their newsletters in a national newspaper.

This program aims to identify talented students in the fields of writing, photography, and design skills. The newspaper will publish the best 16 entries every Thursday for 16 weeks. Selection criteria depend on creativity and originality, content, consistent in style, attractive visuals, and explicit videos. Submission is through the email, and judgment will be upon the determined criteria.

3.17 Fifth: (HA) Hamdan Foundation for Distinguished Academic Performance

Table 13: (HA) Hamdan Foundation Vision & Mission-Strategic Objectives. (www.ha.ae)

The Hamdan's Foundation Vision				
To pioneer the leadership of Academic Performance Excellence, Gifted Education, and innovation.				
The Hamdan's Foundation Mission				
To develop the academic performance in gifted, innovative, and talented education through best national and				
international programs that lead to the distinguished academic society.				
The Award Strategic Objectives				
To upgrade performance in the education domain To encourage initiatives, innovation, and excellence				
To support and nurture the gifted and innovators. To highlight and strengthen the status of the foundation				
nationally and internationally.				
To develop performance, institutional excellence, and customer satisfaction. (ha.ae)				

3.17.1 The role of (HA) in Education

March (1998) witnessed the issuance of a decree to employ this award to support the educational sectors and enhance their performances in the entire UAE. It aimed to work under the supervision of the Ministry of Education (MOE) in order to support the educational levels and sectors within the regulations of the Ministry of Education in the UAE. The council is the supreme body responsible for drawing up the organisation's policies, organising them, and approving their needs,

programs, and projects. The first version of the Hamdan Bin Rashid Al Maktoum Award included nine categories, but it became fully qualified in 2000 to act following its purposes and statute. In 2001, His Highness "Sheikh Hamdan bin Rashid Al Maktoum" ordered the establishment of a "Centre for the Care of Gifted and Distinguished People" to become affiliated within the award and to provide it with a practical lab that supports practices and development of all students who joined the award. The Centre obtained membership at the World Council for Gifted Education (WCGTC)(ha.ae). The award expanded its services later to include the entire UAE and to include the Gulf States as well as. The aim was to develop a prize that nurtures the educational research sector in the UAE at the level of the Arab world. In 2010, they announced another exceptional award for Higher Education. Hamdan award is currently targeting 21 categories. (Dubai. ae, 2019). The foundation announced on (April 22, 2018 (Decree No. 16)of 2018) as a non-profit organisation with a legal personality and a legal capacity to conduct business within limited to its objectives, statute, and provisions(ha.ae). The Foundation aimed to strengthen governmental and community efforts to highlight the importance of education. It also worked to promote the culture of talent, innovation, excellence, and quality in education. It contributes to enhancing the quality and level of performance and creativity in educational institutions. It also aimed to establish and support centres of innovation and scientific research. It supports highly able students in the fields of academic and science. It works to develop their capacities by providing an incubator environment for them. The (HA) develops specialised programs for talented students and encourages the promotion of educational and scientific research. It exerts efforts to promote the field of excellence and talent. It encourages and motivates creators, innovators, and distinguished students in the field of science and academic achievement and the educational as well. (dmi.gov.ae)

3.17.2 The Role of (HA) in implementing Gifted & Talented Development

Definition & identification of the gifted

The definition of highly able students at (HA) is" those who have extraordinary abilities and who perform more distinctly than their peers in mental abilities, academic achievement, and leadership, or any of them which the community appreciates."

The process of gifted identification at (HA) is a critical stage before preparing any development programs. In the initial phase, they develop a field survey. Gifted students will take tests that measure their motivation, abilities, learning, and the learning environment. The selection process will build forms and a personal interview guide for gifted students. The design of the development programs intends to fulfil their essential needs of development, challenge their abilities by smart programs and tools, and enhance their development as well. This process helps to adjust the enrollment system in the programs of gifted students. The programs provide high ability students who deserve these services with access to their educational services. They will be identified to determine the type and level of their abilities and to support them with their different needs(www.ha.ae).

3.17.2.1 Stages of identification

Nomination: This stage aims to accommodate as many students as possible into the screening procedures. Different methods of identification applications are by teachers' nominations, parents and peers or self-nomination, Test results, achievement files, teacher's reports, and background information(www.ha.ae).

Selection: This stage aims to decide the inclusion of the screened students in the second stage of the program. This stage tools include personal interviews with the students, lists of tendencies and list of interests. Used criteria are Intelligence standards, creativity criterion, and academic achievement criterion.

Table 14: Tools and Metrics for Gifted Students at (HA)

Measurements of mental abilities
(intelligence).

Battery Detection Gifted Identification Kit.

Educational achievement tests.

Test problems in creative ways.

Second: Specific tools

Record of achievement Portfolio.

Note-cards to record the behavioural aspects - leadership).

Behavioural lists (self - paired - teachers - parents).

Interviews.

Professional interests and trends.

Discovery bag for gifted students

3.17.2.2 National Plan for Gifted programs

The gifted education is a significant educational priority in the UAE. However, the interest in educating talented individuals by a differentiated education and programs is a new trend in the country. Consequently, this required establishing a framework to organise the gifted education and development programs. Hamdan Awards accordingly developed the (NPGP) to provide the ideal framework for gifted programs and to facilitate their planning and implementation in the UAE. Designing the National Plan depended basically on the latest practices adopted in the field of gifted education. It aimed to make it a reference and a systematic framework that organises all practices and enhance quality by promoting the standardised programs. This plan comes from the belief of

the Hamdan Foundation regarding the importance of caring for gifted people. It aimed to provide the necessary support to the national and social trends in raising the level of human talents (ha.ae).

Table 15. (NPGP) Programs (www.ha.ae)

Discover	Sponsorship	Develop		Awareness	Part	tnership	Center		Schools	
Gifted	Development	Cadres	and	Scientific	All	relevant	Center	for	Hamdan	
students	programs	teachers		dissemination	insti	tutions.	talent	and	schools	for
							creativity	7	gifted	
									education	

3.17.3 Examples of (HA) development programs

(1) The Professional Diploma Program in Gifted Education

This program incepted in 2010 as the first professional diploma program for teachers of gifted and talented students in the Arab world(ha.ae). It aims to develop and qualify the teachers by encouraging professionality in the field of gifted education(ha.ae).

Table 16: (PDPG) Objectives (www.ha.ae)

To prepare cadres of specialists who possess the knowledge, abilities, and necessary skills to implement and teach programs for different types of gifted learners.

To build a scientific reference advisory that provides educational and training systems for schools and institutions with expertise.

To provide on-the-job training alternatives for general education teachers and teachers of special education, enabling them to acquire practical teaching skills by carefully examining carefully selected courses that provide theoretical and practical frameworks for classroom instruction for gifted learners.

To provide diverse models of creative work in the field of distinguished teaching for private groups, benefiting all stakeholders and gifted and talented students in the UAE.

To prepare professionals to lead the educational development improvement to promote and develop the creative child in public and private education in the United Arab Emirates and the GCC countries

The (PDPG) duration lasts for about six months. It targets teachers and mentors of special education and other educational constituents. Enrollment in this program is limited to UAE nationals who hold a bachelor's degree in any areas of study and teaches in public or private schools

who have three years of experience The teacher who is nominated by (MOE) or any official employer in the UAE should have an annual performance assessment that ranked (2) for the last three years and possess a good computer and some e-learning software skills with distance learning. The candidate should pass a personal interview and be ready to commit to (undertake) training and all requirements for graduation from the program(ha.ae).

(2) The Hamdan Center for Talent and Creativity

The establishment of the (HCTC) was to cater the gifted students from Grade 4 to 12 who pass the gifted identification tests. It provides students, teachers, and parents with specialised scientific programs, training workshops, and counselling.

Table 17: Hamdan Center for Talent and Creativity -Objectives

To meet the educational, psychological, and	Increase learning opportunities for gifted
social needs of students in order to fully	students outside the classroom.
develop their personalities.	

Provide specialised programs aimed at supporting students' abilities and supporting their interests.

Provide the gifted students with an attractive learning environment

Spread the culture of talent and creativity in society.

3.18 The relevance of the Institutions Focused Study

The purpose of this chapter was to provide a framework for the research. This research aimed to explore five of the educational institutions responsible for implementing and developing gifted education and programs in the UAE. This chapter had the opportunity to review (5) educational institutions that were launched to look after the highly able students' programs in the UAE, in addition to their primary role as in control of educational reforms and improvement in the UAE. These issues and the information obtained from their portals were to investigate the process of identification and development, and the effect of these processes and procedures on the

stakeholders, which is the focal point of the Case Study. However, and due to little information available in this regard, this research aims to recommend conducting more investigations on this topic for future researches.

The chapter explored the role of each institution, its vision, and mission, its leading development programs, processes and procedures of these programs, and their provisions.

Chapter Four: Methodology

4 Introduction

This research is exploring the role of five educational institutions in the UAE in charge of implementing gifts and talents development programs. This chapter describes the methodological issues relating to the present research. It involves research objectives, its design, the choice of the qualitative approach, the choice of the case studies, issues of validity, reliability, trustworthiness, limitations, data collection and data analysis methodologies.

4.1 Research Approach

The question of what we mean by methodology is a constant significant question and can be challenging to answer. It is the reason researchers give on why they want to conduct their research in a way or another. (Clough and Nutbrown (2012) indicated in this regard that the methods used in doing research, for instance, questionnaires, interviews, focus groups, and any other ways, are like the ingredients in a recipe, whereas methodology is the way that makes the final product. (Robson, 2002) Formerly said that beforehand researching, the researcher should consider several cautions. These cautions, according to Robson, including a lack of theory, the use of a specific method, how to start the research, and if it starts quickly and for what aim (Robson, 2002). Things like these can make the research unsuccessful if not considered before starting the research.

On the other hand, and according to Robson as well, the successful research usually develops from the involvement of different factors such as field participation, merging activities with interests, and having an insight with a concern for theory (Robson,2002). Therefore, this chapter outlines the methods, and all efforts made by the researcher to make the study successful. Choices of methodology in this research originated from the availability of information and to what extent the

researcher can gather the required data. According to (Scott & Usher, 2010), research is not just to follow the right procedure, but as a social practice, too (Scott & Usher, 2010). The research will be complicated if it is applied or explored by international methods plus that the educational field research is not like the other types of research in natural science as (Robson, 2002) claims.

The researched in the educational type of researches have their ideas and perceptions about the world around them(Mertens, 2014). The researcher who adopts a constructivist/ interpretive approach in educational research integrates these beliefs into the research methods with the idea that people build the knowledge, and this should make the researcher exert efforts to understand people and their ways to view the world(Mertens, 2014). The researcher in the educational research looks at ends of value and not ends of facts according to (Scott & Usher 2010). The influence of this research is by what Mertens indicated about the reality being a social construction and the ability to hold multiple realities together even if they conflict with each other. In this method data, the collection is more social and interactive than the methods used by a positivist researcher.

The approach criticism was for being not reliable due to its reliance on the honesty of the participants. Although the researchers try to ensure the participants are honest, what if participants are not honest, and they did not want to reveal their actual views? It is a limitation of this approach as known. However, some researchers see that this type of research, regardless of the participants' defects, may help policymakers find vital ideas that will help them drive forward initiatives. The incompleteness of this type of research may be challenging, too according to (Scott & Usher, 2010). As this approach's objective is to help the researchers understand and not to change anything, it is a weak approach. However, this research aims to understand in a hope to contribute to the literature on a subject that is not yet explored and in need of more researching that might lead to a change, if appropriate, in the future.

This research will adopt a descriptive-analytic approach as the most suitable method. It will support the researcher to address the issues related to identification and development of the highly able students in the UAE and to ensure this research has sound underpinnings. Many researchers like (Bassey, 1999) believe that educational research should improve and influence the educational system in a direct aim. However, educational research is educational because it intends to inform. It critically informs educational judgments and decisions that will improve the educational activities. (Clough & Nutbrown, 2012) Added that educational research should be able to persuade the readers, and its purpose should be clear in addition to being positional, and political as well (Clough & Nutbrown, 2012). They additionally indicated that the excellent methodology is not the confined one within a brief chapter, but it is more a critical design that will always exist at work throughout the study.

4.2 Research Strategy & Design

This chapter constitutes of two sections. The first is the research objective and research strategy.

This section describes the research approach. The second section deals with the research design.

It discusses the reasons for selecting five institutions: data sources and data analysis methods discussed in this section.

The research (1) explores the role of five educational institutions who are responsible for implementing programs to nurture the highly able students from the perspective of experts, teachers, parents, and students.

(2) Determines the position of these five educational institutions and if they act as facilitators of identification and development programs besides their educational role.

- (3) Examines the views of experts and leaders who oversee the identification and developmental programs and the factors that may promote or hinder them.
- (5) Collects and analyzes data about the development programs in these institutions.
- (6)Interviews selected participants to obtain more data regarding the initiatives and programs designed and provided for the highly able students in public schools in the UAE.
- (7) Interviews parents, their gifted and talented children, and to get more data regarding the effect of implementing these nurturing gifted programs.
- (8) Traces challenges and places suggestions that might help in supporting decision-makers in their future decisions regarding Current gifted and talents' provisions; Identification and development programs; Challenges; and social awareness.

To achieve this study's objectives, it attempted to answer the following research questions:

1. What is the role of development programs implemented by governmental, educational institutions to develop gifted and talented students in the UAE?

Sub Questions:

- 2. How are gifted and talented students defined and identified in the UAE?
- 3 How is the development of gifted and talented students implemented in the UAE?
- 4. What are the main challenges facing the gifted and talented identification and development in the UAE?

This research design followed the qualitative loose structured design with data that is not prestructured. The field interviews conducted at different sites where applicable. The primary data collection techniques used in this research study came from "individual interviews, focus groups, documentation analysis, and library search".

4.3 The Case Study Strategy

The Case study method, according to (Denscombe, 2003), gives the researcher a chance to focus and investigate in-depth in the natural setting of the subject. It also provides insights into how things are processing in the subject of the study. He added that it is a comprehensive approach to social research (Denscombe, 2003). Boundaries of the case study should be known and understood according to (Stake, 1994), even though the case is simple or complex. The researcher should set the boundaries at the beginning because the case, according to (Miles and Huberman (1994), is in effect with the unit of analysis, and there should be a focus that helps the researcher builds outwards. Moreover, a case study has several standard features with social research illustrated as (1) Attention is focusing on one instance, (2) it is an in-depth study, (3) focuses on relationship and process, (4) using multiple resources and methods and (5) The research takes place in a natural setting. However, only some case studies are explanatory. They may be an inquiry tool by employing a positivist ontology (Miles and Huberman, 1994). Yin (1994) advises not to confuse case studies with qualitative methods. He said that using the ethnographic method needs a more extended period to conduct and very detailed evidence, while case studies are within a limited period, plus they do not necessarily imply the use of ethnographic techniques (Yin, 1994). The case study, according to (Schwartman, 1993), helps the researchers obtain more knowledge and identify actions and instruments utilised daily by the participants. The most supporting thing for the researchers conducting case studies that they do not have to visit the institution under study, but they can collect their data from secondary sources such as documents and websites or by interviewing participants from these institutions by a telephone call or emailing them the

questions(Schwartman, 1993). The case study approach is useful in critical situations where the researcher can not control the contextual conditions of the studied events, and they require specific techniques for data collection. However, sources of collected data should be diverse to ensure their honesty and truthfulness. The different types of case studies according to (Stake, 1993) are (1) Intrinsic case study which is unique, (2) Instrumental that provides insights to existing theory (3) Collective case study that encompasses to more than one example. (Stake, 1993). (Yin, 1993), as well, defined three types of case studies;

(1) Exploratory case study's first stage is to define the researched issues. In this type of case, the allocation of data is before theories or before formulating the research questions. This type of case leads to a systemic approach. (2) The casual case studies for the relation between cause-and-effect (3) The descriptive case study that requires a speculative theory that helps to guide the collection of data. Case studies can be multiple or single, and they can be embedded or holistic. (Yin, 1994). This study involves five educational institutions' boundaries. Therefore, it adopts a case study method to collect more data regarding their role in the identification and development of highly able students in the UAE. This research uses one of (Stake's, 1994) explanations, the Instrumental Case Study. It provided an insight into the role of each institution, how it implements developmental programs and the effectiveness of the implementation of these programs within and outside the institution.

Table 18: Denscombe's Elements of a case study

Denscombe's Elements of a case study Denscombe (2010)

How are they addressed in this study

It focuses on a few instances of the phenomenon.

It provides an in-depth description of events, relationships, experiences, or processes occurring in the instance.

A detailed study.

Five educational institutions are the subject of the study. The collected data allowed a greater depth of the study.

Interviewing different participants from different categories generated more information about the institutions.

4.3.1 Case Study advantages & Limitations

The collected data for this study were from multiple sources that vary between interviews, focus groups, documentation, and digital resources. The use of the data was to build up the roles of each institution in implementing gifted and talented development programs. (Cohen & Manion 2011) Thought that the Case Study approach is an advantage for the accessibility of the language, and this makes different viewers utilise the research (Cohen & Manion 2011). Another advantage is how case studies are aiding the researcher to explore more about the definition of gifted, talented, and different types of identification used in these educational institutions. It makes the study useful and beneficial to the educational institutions subject to the research. It helps them find more about their programs and use the collected information for their future improvements. This study, on the other hand, is a contribution to the area of gifted and talented in the UAE.

However, and despite the above, according to some researchers, finding an appropriate case study to suit all subjects could be a problematic approach. The Case Study approach contains several disadvantages, as well. The major criticism of this approach was by (Bassey, 1999) who claimed the generalizability of the findings in the case study method. (Denscombe, 2003) As well mentioned the reflexivity in a case study, which allows the researchers criticise and reflect their thoughts and biases in addition to the observations and perceptions of the researcher that can completely miss other aspects while presenting it in one manner. Case Studies, in general, are time-consuming if compared to other instruments, and they may not be suitable for studies that require

shorted time. However, the advantages of the Case Study approach far outweigh the disadvantages by knowing a case study is a great tool that supports many studies and researches.

Table 19. Advantages & Disadvantages of a Case Study

Case study advantages

The case study approach allows an indepth analysis of data.

- The case study approach explores the issue in its natural setting.
- It allows answering questions like (how and why).
- It allows defining the complexities and more exploration.
- The case study approach is flexible.
- Methods of data collection, data analysis, are not limited, and it allows for the usage of multiple sources of data.
- It can use either quantitative or qualitative methods.

Case Study Disadvantages

- The major criticism of this approach was by (Bassey, 1999) about the generalizability of the findings.
- (Denscombe, 2003) talked about the reflexivity in a case study.
- According to some researchers, finding an appropriate case study to suit all subjects could be a problematic approach.
- The Case Study is the study of observations and perceptions of the researcher.
- Case Studies, in general, are timeconsuming if compared to other instruments.
- They may not be suitable for studies that require shorted time.

4.4 Data Collection Methodology

The data collection methodology included different tools and techniques to collect the required data that will benefit the purpose of the research. The research methods differed for diverse types of inquiry (Walliman, 2017). The selection of the accurate method ensemble the research objectives and guides the researcher to answer the questions accurately. This section explains how the researcher divided the components into two: Data collection and data analysis. Several sources of evidence can be used in qualitative research to collect data, including documentation, interviews, observation, archival records, and physical artefacts (Yin, 2009). However, researchers

who usually use the case study approach can differ their methods of data collection according to (Bassey, 1999).

The researcher in this study applied three methods for data collection discussed in detail in the following section:

4.4.1 Library research and document analysis

Organizational and institutional document analysis has been an essential tool used in qualitative research for many years. Documents mean Public Records that include any organization's activities; Personal Documents that mean the first-person accounts of an individual's actions or experiences; and Physical Evidence, which means any physical objects found within the study (flyers, posters, agendas). The researcher interprets the documents to clarify an assessment topic, and many researchers have used this method in recent years. (Bowen, 2009)

Furthermore, and despite the frequent use of this method, some indications argue that document analysis has not always been successful because of the ineffective use of it in the research process, even by experienced researchers. The literature related to this study is the primary source of documentation, too (Denscombe, 2003). Documents analysis as a method of data collection has advantages and disadvantages (Yin, 2009). The illustration of strengths and weaknesses displayed in Table (24) below.

Table 20. Strengths and weaknesses in document analysis

Strengths	Weaknesses
Documents analysis is an efficient method.	Not suitable to evaluate users' values or needs.
It is less time consuming and more efficient	Many documents are not publicly available
than other research methods.	and can be challenging to find.
It requires the researcher to select data than to	They may have any bias if they reflect the
gather it.	unknown based on the author.
Stable and can be repeatedly reviewed.	Some documents were difficult to access.

They cover a more comprehensive range of topics, settings, and events.

They contain exact names and correct information.

The documentation in these educational institutions might have strengths and weaknesses as well. A significant weakness in the UAE is the absence of researches and studies that discusses the highly able students' needs and issues. Most of the published documents on the internet include decrees, laws and regulating documents, inspection reports, private school policies, and the institutions' launched initiatives or plans, registration forms, applications or other news.

The quality of the gathered data needs an assessment before using it (Robson, 2002). It means the researcher must check up on the assembly of this data or documents and how each institution gathers data and select which data to publish, plus, to check on this data and make sure they are updated.

Table 21. Some of the documents reviewed for the sake of this study

The essential documents reviewed for the sake of this study included and not limited to

• School for All. (MOE)	 United Arab Emirates World Data on Education.
The KHDA's Private Schools Inspection Reports.	United Arab Emirates Constitution of 1971.
• NPGP.	The Current State of UAE Education.
 Professional Diploma in Education of the Gifted Module and Requirements. 	• KHDA Inspection Key Findings-2018-2019.
 MCKD Draft of the Talents' Nurturing Policy. 	 Dubai well-being –better.
 UAE Model for Government Leadership 	 UAE National Innovation Strategy
UAE Vision 2021	 Government Achievements
 Government performance management framework 	MOE Schools statistics 2017-2018
MOE international Tests Management	 MOE National & International Assessments
 Licensing and Accreditation-ADEK 	Qualification-ADEK
Abu Dhabi Vision 2021	Abu Dhabi Plan
Talent Policy-MCKD	Cultural Agenda-MCKD
Education Journal Middle East	UAE's Fourth Industrial Revolution Strategy

• Global Economic Outlook

 The Role of International Assessments and School Inspections in the Reform of Education in Dubai.
 (Cambridgeassessment.org.uk)

Dealing with documents required the researchers to be very cautious about their trustworthiness and the quality of data and their purpose. According to some researchers, this data needs evaluation and assessment, especially if this data is for a promotional purpose. Such worries would make any researcher very careful in selecting the documents for the research. However, the researcher in this study was familiar with the way these institutions compile their documentations. The documentation protocol in the government, as mentioned previously, is encouraged to be transparent and useful to allow people or any researchers to benefit from them. The researcher downloaded the documentation used in this study from official websites, and they are trusted resources. These resources include the UAE government portal, the institutions' official websites and e-libraries, the official accredited newspapers or media websites, and the articles or interviews with UAE officials regarding gifted and talented issues. The researcher considered each document used in this study to guarantee its appropriateness.

Library research deals as well with the analysis of historical records. The required data obtained from books and other library materials, both published and unpublished such as the academic dissertations, journals, articles, conference proceedings, and theses. Library research includes both real and virtual online libraries. Researchers usually read books, conduct an internet search for articles, journals, and other dissertations and thesis that have discussed the same topic. Web search can substitute challenging to access populations (Cantrell & Lupinacci, 2007).

On the other hand, researchers should be cautious and consider how they will collect the data from the web, which websites are guaranteed and approved and the quality of the data that will be the required to validate the findings. However, library research provides data that is available for everyone who needs it. It is the utilised data by any researcher or students or anyone who requires information regarding the selected populations. The researcher collected the secondary data from different sources such as online published documents and paper published. While this secondary data is not what the researcher aims for, but they are useful for the base of the study and to understand the history of the institutions under exploration. The secondary data formulated the literature review in chapter two. The literature review includes the secondary data gathered from several journals, articles, books, and any other documents regarding gifted and talented issues from all over the world. Such data is required to clarify the processes and procedures followed by the educational system to nurture the highly able individuals in the UAE. The secondary data in chapter two was to describe and explain the theories of intelligence, the definition of talents, and the description of gifted programs implemented in and by these UAE educational institutions.

4.4.2 Field research: Focus groups and interviews

Field research contained two methods to help answer the research questions. These were focus groups and interviews. (Walliman, 2017). The researcher selected the participants for each of the interviews and the focus groups. The selection of the participants, according to (Hackett, 1998), supports the study with the required information. The collected data focused on the extent of knowledge the participants have regarding the definition that describes gifts and talents, the identification process and the role of the development initiatives and programs in these students progress. The interview method helped the researcher to obtain the primary data. The employment of the interviews and the focus groups contributed to the researcher's aim to obtain primary information with prepared written questions.

4.4.3 Designing the interview

The interview questions were bilingual. This bilingualism considered the importance of making the interview easier and accessible to everyone.

The designing of the interview questions depended on the following:

-Analyses of the literature and all available documents related to identification and development processes and programs used in these institutions; Referring to the research questions; Self-reflection by identifying assumptions, and Preparation of a provisional guide for that.

The interview questions consisted of four parts:

- The first part dealt with preliminary data to know how the interviewed define gifted and talented people, and why they think that definition was the proper definition. What identification methods they used or how they identify highly able students. All participants answered the starting question about the definition of these highly able students and how do they define them.
- The second part was about the role of the development programs towards these individuals' progress. How do they come to know about these programs and how they evaluate them?
- The third and last part consisted of the questions about the challenges faced by this category from the perspective of the interviewed stakeholders and to poll suggestions regarding the responsibility to support talents' development in the UAE.
- The fourth part constituted the recommendations on how to overcome any challenges and by whom.

4.4.3.1 Participants in the interviews

- Different groups participated in the interviews such as teachers, experts, managers, highly able students, and their parents in case they were minors. These stakeholders' categories helped the research obtain real data through verbal communication and by stimulating them to give information about the situation of the gifted programs and to what extent do they serve the highly able students.
- The design of the interview questions was in a way that helped the researcher obtain more information regarding the development programs, how do they define and identify the highly able students, what were the challenges and how did they overcome them, if they did.
- The researcher defined the objectives of the interviews with the participants before conducting the interview.

Pre-preparation of the interview include,

- A) Identifying and preparing the questions that the participants can answer according to their category and personal experience.
- B) The researcher listed the intended questions and queries for the interviews and sent them to some of the participants by email before interviewing to give the participants an idea about the subject considering the clarity of questions, and by making, they direct with precise wording.
- (c) The selection of the timing of the interview was according to the participants as the plan of the interviews was to be in proportion to the circumstances of the interviewees and their obligations to do so.

(d) Conducting some of the interviews in the researcher's place of work reflected a feeling of satisfaction among the interviewees who felt more relaxed away from their workplaces.

4.4.3.2 Conducting the interview:

In order to make the interview useful, the researcher took the following procedures while conducting the interview:

- A) Created an appropriate atmosphere for the dialogue by the first impression with an appropriate appearance and by choosing the appropriate phrases for the interview. The conducted interviews in either Arabic or English were according to the interviewee's satisfaction and desire. This bilingualism was helpful and welcomed by the participants who were able to choose which language is more satisfactory and helps them express more. English language interviews were beneficial with those who were part of private schools and institutions, while the Arabic language helped candidates from the public schools and the institutions.
- B) The research topic created an atmosphere of cooperation with the respondents who felt it is addressing their real challenges for the first time. The setting away from the office most of the time ease the environment for the interview and to allow the conversation to run spontaneously without making the interviewees feel that the interview is an interrogation.
- (C) The researcher gave a clear description to explain the questions and to reform them if some of the participants did not understand them.
- (D) The researcher obliged with the time required to gather data and finished them in a timely fashion.

- E) The researcher conducted the individual interviews separately in isolation from the rest of the staff, and this was very helpful, especially in an environment where offices crowd with more than one employee.
- F) Respondents did not see any impression that their answers are not correct in cases when some of the participants gave unneeded answers. They could complete the answers or add to them, and then the researcher tends to ask them to explain their answers and to give examples for what they mean.

4.4.3.3 Registration and recording of information

- A) While conducting the interview information, the researcher tended to record the answers immediately on papers and sometimes printed the answers directly on the computer. The distribution of questions came in the form of sections in an organised way to make them easier to add before or after each question, as well as the need to take additional notes if required (in cases the participants illustrate or tell a story about what happened to them). It was better (however, not welcomed by the interviewees) to use a recorder.
- B) The researcher recorded the information using the same words and language, as said by the interviewees and printed the handwritten information.
- (C) The researcher did not interpret the answers provided by the interviewee, and nothing added to them; however, some of the statements were subject to reinterpretation by the researcher when that was necessary. The researcher, however, had to translate the interviews conducted in Arabic in a way that tried to keep them as they said.
- D) There was a balance between dialogue and comment and recording and writing answers, and this helped the researcher obtain the required data.

E) The researcher's colleagues were conducting interviews with different participants, which helped to get different views and participation.

(F)The researcher revised the answers and observations with some participants and sometimes sent them to the other colleagues' researchers after the final draft to review them.

4.5 Research Design-Structure

Research design means the formation of the plan on how to execute the research in order to get the needed data that will help the researcher answer the research questions. (Punch, 2013). Therefore, the research design is the most significant part of the research. It includes the four principal contemplations such as the" strategy, conceptual framework, identification of subjects, and the tools and procedures for collecting and analysing the required data". The division of research design can be into different types, such as qualitative, quantitative, and mixed methods research. (Holmes, Dahan, Ashari, 2005). The development of the qualitative method in social sciences enabled researchers to study social and cultural phenomena. The qualitative method helps them observe feelings, thoughts, and behaviours and believes of people in societies (Babue, 2008). The qualitative method as well includes different examples such as "action research, case study, and grounded theory". The sources for collecting data in the qualitative approach include observation, interviews, questionnaires and documents reviews (Burns, Bursn, 2000).

The usage of a qualitative method in this study relied on the nature of the educational research that entails research in human phenomena and the interpretation of results through researchers' perceptions. As the researcher needs to interpret the field of human activity and examine people's

experiences and search for the answers to questions that concern why, what, how, and which method to explore and understand the effect of implementing the developmental programs, therefore, qualitative research is the most capable of understanding and interpreting problems related to humans and searching for the causes and beliefs of the research problem and aims to understand and interpret the contexts related to the problem. In addition to the fact that qualitative research is the best source of information and data that we urgently need in the UAE in the absence and lack of research, studies and clear data. The qualitative data that can be collected includes the qualitative empirical materials that comprises the different types of data resources and how the qualitative research can benefit from as many as resources such as Interview transcripts, documents, records, audio-visual materials, and personal experience materials such as diaries, journals and narratives, emails and instant messaging conversions.

This research employed the method of individual interviews and focus groups. These methods were always essential means to obtain more in-depth insights from the participant, therefore, they helped the researcher to determine the extent of awareness of stakeholders regarding the development of the highly able students and the services provided for them through the educational institutions, subject of the case studies in the present study (Frankel, Wallen, Hyun, 2012). The involved categories included teachers, experts, parents of the students, and highly able students.

This study, henceforth, aimed to gather data from stakeholders' interviews and obtain more information regarding their awareness concerning the fundamental factors and criteria's that surround and support the successful implementation of such programs.

Additionally, the researcher also explored how these institutions collaborated. Whether they provide all students with equalised treatment and services through the entire UAE, and if their programs nurture all types of gifts and talents from different ages and domains. The researcher,

also, inserted the practices and procedures on how the identification procedures are used to select these students and nominate them for development programs. The interviews explored the extent of social awareness about the availability of these programs and the extent of involvement of the teachers, experts, parents of the students, and the highly able students as well.

According to the researcher, most of the past researches conducted in this area and on the gifted education, and development programs adopted all forms of methodologies in researching, and all the approaches contributed to the issue of development of these students. Therefore, this research will use the case study to enable the readers to understand how these institutions are fulfilling their roles and how they create and implement identification and development programs for the highly able students in the UAE. The researcher's intention to use a case study technique was to collect all the available data and to view the similarities and differences of these institutions' roles regarding the development of gifted and talented. (Coleman, 2007).

Several researchers and educators strongly supported the idea of conducting researches using qualitative methods such as interviews and observation in addition to case study tools when the research is dealing with humans' interaction such as (Coleman, 2001, Babu, 2008, Reise, 1995, Zabloski & Melacci, 2012). Therefore, these methods supported the research accuracy and helped it to deliver more truthful precise findings.

4.6 Research paradigm

All primarily collected data for this research came from individual interviews, focus group documents, and open data published on their official websites to understand the issue and to identify which information would help formulate the discussion. The obtained information helped the researcher write the results and the final recommendations in chapters (5 & 6). Analyzing data

was in an inductive way that emphasised on the meanings mentioned by the participants of the interviews and the participants in the focus groups and described the process in a convincing expressive language as intended by the researcher to interpret how the interviews and focus groups participants reacted to the questions. Such method helped the researcher understand the extent of knowledge and awareness people have about the importance of gifts and talents and regarding the development initiatives and programs offered through these educational institutions. It also provided a naturalistic description of provisions from these programs, validated and generalised the results.

4.7 Population, research sample and sampling methods

Reid (2013) describes and defines the population as "all units who possess specific characteristics." These features are of the interest of the researcher's study (Reid, 2013). This definition interprets what the researcher target out of the population by saying that they are the community or group of people who are essential and useful to add valuable information for the study. When in the hunt for the required population for this study, two main types were to consider. The use of probability sampling in a quantitative method is to support this type of research, while non-probability helps to collect views from groups of people. Therefore, and as this research is adopting the qualitative type, only individuals who form the essential stakeholders in the lives of the highly able individuals and are involved with the issue of their development were the appropriate sample that must constitute the population in this study. The questions suggested by (Punch (2009) were used in this research as the leading questions while trying to decide the population of the research: (1) How big will the sample be, and why? (2) How the researcher will choose the sample, and why? (3) What will the researcher claim?

Consequently, and for the sake of this study, the researcher selected the population from teachers, managers, experts and parents who are familiar with these educational institutions and related to the highly able students.

In this study, the researcher employed a purposive sampling method to select the participants required for obtaining specific information. These participants constitute of teachers, experts, parents of these students and the highly able students who are involved in development programs in these institutions. Purposive sampling according to researchers refers to the method that the researcher intentionally chose to select the research sample according to their knowledge about the study and its requirements(Palinkas, L.A., Horwitz, S.M., Green, Wisdom, J.P., Duan, N, and Hoagwood, 2015). The researcher deliberately selects some participants to suite the research aim, and because of their willingness to participate in the study as well. The study strategy focuses on obtaining relevant and useful information that will accurately answer the research questions.

4.7.1 Purposeful sampling

Selecting these five leading educational institutions in the UAE was a careful selection that depends on the following criteria:

- Their position as the regulating bodies for education in the UAE besides providing development programs and initiatives to nurture different types of highly able students in schools in the UAE.
- The conducted individual interviews were face-to-face. Focus groups were through groups divided into categories. However, some of the necessary information came from documents, newspaper articles, and interviews, the institution's websites, and anywhere in

which there are reliable and validate available documents and open data concerning the same issue.

 The sample included gifted and talented from several schools, programs, initiatives, domains of giftedness and talents, and age groups.

Participants

- Responsible for the implementation of development programs like teachers, psychologists, supervisors, and sometimes parents.
- Users (talents, experts, program managers, and sometimes parents)

• Site

The conducted interviews were at different sites in the UAE.

The study community consists of:

A selection of educational institutions and Stakeholders according to their direct relation to the subject.

4.8 Ethical Considerations

When a researcher uses a qualitative method, according to (Scott & Usher, 1999) and other writers, ethical issues usually occur. It means the researcher must distinguish between what is right and what is wrong while conducting the research. Ethics also refers to thinking about rules and what ought to do, what is acceptable, and what is not. Researchers face several ethical necessities, as they must meet professional and institutional standards for conducting research that involves human participants. Research ethics involve informed consents, beneficence, respect of anonymity and confidentiality, respect of privacy. However, the researcher has chosen to declare the names of the institutions as the research topic focuses on exploring the methods and tools used by these

institutions in implementing development programs for highly able students. The research depended on their official open data and published documents information. These publications are downloadable and free for use, reuse, and sharing by anyone, including researchers according to the government's open government data policy (government.ae). The UAE government has a transparency policy to support improvements for all its institutions. It is accordingly exerting efforts to implement all requirements to benefit the society and the development of these students too.

Moreover, these institutions, launched by the government to oversee their role, need to estimate and evaluate the extent of their achievement and how much did the targeted population benefit out of their efforts. Therefore, it is the researchers 'role, to support these institutions' development by the findings and results obtained from various methods such as document reviews, interviews and focus groups. Recommendations as well would help them implement their educational development system in a better manner. The researcher's role is to help decision-makers become aware of what makes educational improvements more beneficial and fruitful in the future. The findings would provide these institutions with an in-depth view of what are the stakeholders' needs or how they view their efforts. The given opportunity to the researcher to interview and obtain data throughout stakeholder's perspectives or expectations formulates massive support for these institutions and decision-makers. It is the research role to evaluate, explore and transfer the accurate information to the concerned entities in order to make improvements and modifications wherever required.

Ethical issues appear in all stages of a study. According to several researchers, the inclusion of anonymity, informed consent, and confidentiality impact participants and participant's potential impacts on the researcher (Sanjari, Bahramnezhad, Fomani, Shogi, and Cheraghi, 2014). Since the

nature of the qualitative research, forces the researcher to deal with human participants, the research needs more guidelines, and the researchers must take the necessary steps to clarify their role in the research process. However, ethical issues occur in three stages: The recruitment stage, the fieldwork stage and reporting, and the four layers such as utilitarian, relational, ecological, and deontological (Flinder, 1992). Other models named the four layers in different names such as external, consequential, deontological, and individual; others like (Mertens, 1998) looked at it as beneficence, respect, and justice. This ethical study framework will refer to (Flinders' model, 1992).

4.8.1 Utilitarian

Utilitarian means: If the action makes the highest good for the most significant number, then it is ethical. The action includes obtaining the informed Consent, avoidance of harming any participant and assuring confidentiality for them (Flinder 1992). Informed consent means that participants understood that the conducted interviews are for the sake of research that seeks to explore their knowledge regarding talented programs and what kind of challenges they face when it comes to benefiting from these programs. They were open and ready to answer the interview questions. Avoidance of harm means to avoid harming any participants who are involved in the research interviews, (Flinder 1992) stressed on the responsibility of the researcher to protect participants from any harm that may occur because of that. In the present study, it expects no harm to occur due to its nature. However, the researcher ensured that all participants felt relieved and prepared for the interviews in addition to protecting their anonymity.

4.8.2 Confidentiality

The researcher addressed this issue by making the interviewed anonymous and by using the available already published open data within the institution's websites or in, their documents and they include most of the required information and data. Besides, the researcher did not discuss most of the interviews conducted with any personals who work within these institutions, although sometimes debating some of the general themes that had come from the discussions in other meetings with later participants.

4.8.3 Relational Ethics

Relational Ethics means deriving moral behaviour from people's attachments and regards for others and not from rules and obligations(Flinder, 1992). The researcher used Flinders' ethical grid to evaluate ethical processes like others, such as (Stutchbury & Fox, 2009).

Ecological Considerations: Flinders' definition of environmental ethics is a set of moral principles that govern human attitudes towards the environment. The researcher in this study interviewed the different stakeholders to learn more about the talent's community and their opinions regarding talent as a definition and a concept.

Deontological Ethics: This is about how to make decisions regarding whom to interview and when to interview them. The availability of participants related to the subject studied in this research.

Reciprocity: The researcher made some verbal agreements with the interviewed about periods and sharing their interview information in research conducted to explore the reality of talented' identification and development in the UAE.

Avoidance of wrong: This research has used some information derived from the conducted interview data, focus group discussions for the (DOT), and the information downloaded or searched in the documents regarding the institution's activities in this regard.

The researcher fairly reported the findings at the end of the research. They derived from the answers and the discussions obtained through the interviews and the focus groups. Fairness means treating the institutions in a balanced way in this research. As the researcher is familiar with these institutions and works in one of them, possible bias, however, may be unintentionally there. To avoid this, the researcher also used the interviews and focused groups obtained for the (DOT) as documents review in researching for answers.

Various guidelines used to set out ground rules to help inform the practice of research. These guidelines were,

- 1- Responsibility to the research profession: The researcher avoided fanaticism or bias in the selection of the study society and was accurately represent and reliable in generalizing the results. The research findings were not false, fabricated, or misrepresented. No exploited personal benefits from these institutions. The researcher did not compel others to participate in the research as a sample.
- 2-Responsibility to the participants: The participants were aware of the interview objectives and aims. The researcher ensured they understand the objective of the study and the importance of their part in it and the participants understood the aims and purposes of the interviews. Participants had the opportunity and knew that they were free to withdraw from the interview at any time they wish or when feeling unable to continue. Anonymity and confidentiality to participants ensured too. The

researcher took into consideration the policies and guidelines of the research conducted within government institutions that have their own rules and require special attention and care.

3-Responsibility to the public: The researcher reported the findings in a clear and accessible way relevant to all stakeholders.

4.8.3.1 Validity, Trustworthiness, and Reliability

Validity and reliability are among the most critical topics of interest to researchers in terms of their profound impact on the importance of research results and their ability to disseminate results. Reliability and consistency relate to the instruments used in the research, their ability to measure, and the accuracy of the readings from those instruments. (Campbell and Stanley (2015).

To elaborate, validity, for example, in community, research relies on the use of questionnaires; research may be looking at the prevalence of a definition for talents. In order to ensure that the tools used in this research are credible and reliable as well, they must meet several conditions or criteria. The simplest of these criteria is the sincerity of the arbitrators because the arbitrator is a competent person in this field and has the scientific and practical background to qualify and to assess the questions set in the questionnaire. They must ensure that they measure what they have to measure.

4.8.3.2 Types of validity

Apparent honesty: The appearance of the tool indicates its ability to measure what it should measure.

The validity of the content: If the content of the tool questions of various forms measures the scopes of the study.

The validity of the concept: Relates to conceptual construction, which means that the tool measures the concept of the subject of the study.

Factor validity: Concerned with the analysis of the adjective measured, in the sense that the subelements of each adjective pour all in measuring the attribute.

Test validity: This means the extent of the meter to a specific criterion (test); the test is successful if the test is sincere in disclosing the test.

Predictive honesty: It depends on the extent to which the scale predicted by reality.

Telemedicine honesty: The agreement of the results of two measures assessing the same capacity while one is honest and consistency.

Validity then refers to the extent of accuracy of the phenomena represented by the research.

4.8.3.3 Reliability

The use of reliability and validity is usual and familiar in quantitative research. However, the idea of reliability and validity is common in all kinds of research. Reliability is misleading in qualitative research according to (Golafshani, 2003) who indicated that it is essential to redefine reliability, validity, and triangulation to reflect and suit the ways that the qualitative research is reliable and valid and telling the truth. (Golafshani, 2003). The way of obtaining information requires testing the quality of this information, and to which extent it generates understanding while the quality in the quantitative research reliability is its purpose of explaining (Golafshani, 2003).

Therefore, reliability is the instrument's ability to give the same results in the case of repeating the measurement on the same person multiple times within the same conditions (Carmen's and Zeller, 1979). In most cases, it is a correlation coefficient and is the extent to which readings of repeated

measurement results correlate. In many types of research where the use of a measuring instrument is for the first time, the test will be first on specific people, and then the same people retested and then calculate the correlation coefficient of the measurement results the first time with the next time. If the instrument is very reliable, the results of the first or following times will be identical. More arguments discussed the concern with validity and reliability to contain some relevance to qualitative research and not only in the tradition of quantitative research. The concept of reliability is providing a degree of confidence related to the possibility of a research procedure for repetition and its ability to produce identical or similar results (Coleman and Briggs (2007). Two types of errors may occur with validity and may occur at the data analysis stage too, according to (Silverman, 2006). The two types are (1) to believe that a statement is true when it is not valid and (2) To reject a statement that is not true (Miles and Huberman, 1994). The use of the concept of validity as a judgment is to find out if the research has described the phenomenon in an accurate way intended by the researcher. All the research design, the methodology, and the conclusion need to have regard to the validity of the process. Validity has two main types: Internal validity concerned about the accuracy of findings representation and external related to the ability to generalise to a broader population.

4.8.3.4 Construct validity

Construct validity refers to how well data collection methods were able to measure the phenomena that are supposed to be measured. It verifies by comparing the data to other data that measure similar qualities to see how these two measures are connected. There is a need for multiple proofs to ensure their validity (Yin, 2009). One way is by lining the outcomes of the interviews with the other results obtained by another method of the instrument.

This study has interviewed (161) participants who have different perspectives and the emerged themes accordingly based on more than a point of view.

4.8.3.5 Other measures of validity

- The researcher allowed the participants to evaluate the interpretation of the data by being open about the values and revising them.
- DOS feedback granted. Modifications made based on the remarks and opinion of the DOS.
- The researcher conducted a sample interview to verify the genuineness of the instrument and to improve them through feedback after identifying the existence of some problems in them, such as the wording of questions, the length of the interview, and its legality.

4.9 Reliability of the study instrument

Reliability depends on the ability of the replicated study to reproduce the same results under a similar methodology (Joppe, 2000). It assesses the instrument and the procedure used for data collection. The tool must be reliable as one of the fundamentals in the evaluation of an instrument that measures concepts, psychomotor skills, or practical values. Validity is the instrument's ability to measure (Tavakil, Dennick, 2012)

4.10 Reliability in the Cases Studies

Reliability can be difficult in the Case Study method, because in a case study approach what is right in one setting may not be accurate in another as the case study settings are unique (Stake,1994),. However, while this does not invalidate the research, it may consider it as less useful.

To establish reliability in a Case Study, the researcher should follow its protocol that includes an overview of the project (Yin, 2009). In this research, case studies follow the same strategy by writing them all in the same method, collecting data by the same tool and using the flexible methods intentionally to ensure consistency between the institutions through their presentation. Moreover, the researcher paid attention to data collected over the months. The data collection was through interviews, library search, analysis of documents, official websites, and prominent, well-known journals and publications. This data allowed the researcher to review these five institutions' processes, as well as the development programs and the individuals involved with them. They gave the researcher and the research more trustworthiness and reliability.

4.11 Triangulation

The research used multiple methods of data collection in this study to ensure the validity of the findings and results. The researcher did not implement the triangulation process in the case study method because it is inappropriate for that method. Case studies in this research were only to demonstrate the roles of the five educational institutions and to compare their systems, processes, and programs. The use of triangulation in research is to ensure that the collected data to answer the research questions are valid and reliable. The different types of methods allowed the researcher to ensure, crosscheck and compare data. The interviews, focus groups, and document review ensured that. Respondent triangulation achieved by asking many different participants the same question to ensure its validity. Researcher triangulation achieved by getting support from the researcher's colleagues to collect and analyse the data that ensures comparability. Discussing the answers and clustering them by the used digital software (Mural) invested in analysing data and obtaining results.

The researcher as well used the themes obtained from the interviews to form the focus groups' scenarios.

Table 22. Summary of advantages of Triangulation

Advantage of Triangulation	Different methods are used to ensure the reliability and validity of data.
More data	Data obtained from individual interviews focus groups
	and documentations.
Different perspectives and opinions	Different methods helped to find out different perspectives and to hear different opinions about one issue.
Allows corroboration	Matching perspectives of different stakeholders regarding the same issues (themes).
Enhances validity	Gathering data across multiple methods enhances the validity of the data and the results.

Using different types of methods was a beneficial method in this study to develop the converging lines of inquiry. The individual interviews were constructive to explore as many as possible perspectives and unexplored community attitudes and to include as many as possible participants. Moreover, the advantages of multiple methods in the research to gather more data at a particular time. (Denscombe, 2003). The second benefit was the ability to corroborate by comparing data from these different methods.

Table 23. Key themes addressed by the interviews

Participants'' Groups	Key themes addressed by the interviews
Managers, experts, and teachers	The definition of talent, factors influence talent
	development, community awareness and knowledge of
	talents and giftedness, Identification tools and programs,
	development tools, and programs, challenges faced by
	talented individuals, program challenges, funding
	challenges, enablers challenges, and recommendation.
Parents	Definition of talent, knowledge, and awareness
	regarding talents' identification and development
	programs in the UAE, identification of child's talent,
	Development of talents, Challenge faced by parents -
	child, suggestions, and recommendations.
Younger gifted and talented students	Definition of talents, identifying self as talented,
Matured gifted and talented	knowledge about identification and how they got to
	know about the development programs, how they want

to invest in their talent and what did they achieve, participating in different programs and functions inside and outside the UAE, who encouraged and supported their talent, challenges faced by them, suggestions and recommendations.

4.12 Focus Groups & Protocol

The second method of data collection was the focus groups. Focus groups were to provide supportive data and information to add to the complementary data (Robson, 2002). The primary purpose of the focus groups approach was to gain in-depth information and understanding of social issues (O.Nyumba, T., Wilson, K., Derrick, C.J. and Mukherjee, N.,2018). The attitudes, feelings, expressions, stories told about the experiences, and the spontaneous reactions of participants during the discussions were more helpful in a way that was not feasible in the interviews. Playing the scenario games while discussing the major issues, helped a lot in generating an original spontaneous type of data.

A focus group usually refers to a who gather to discuss an idea. For this research, the selection of focus groups was from the categories considered as the main people surrounding the highly able students and who have the most impact on their lives and talents. Those participants came to share their ideas and views around the development of this category in the UAE, and the impact of this development on their progression. The topics of the scenarios allowed the researcher to know how much they know about the development programs and how much these students benefit from them. The note-takers recorded their responses in order to measure the likely reaction of the more extensive conducted interviews. The purpose of the focus groups was to see what people share and what they would say or reflect upon the introduced situation that they had to plan.

4.13.1 Facilitation & Note Taking

A Note Taker in each focus group had to record the whole scene and the brainstormed discussion. Those note-takers who are the trained teachers managed the registration of the discussions as well as capturing the generated ideas through small and large group discussions. They recorded the interactions and the outcomes for the group on a notebook and a flipchart. They were timekeepers too. Note-keepers should moderate their notes with the facilitator's notes to ensure full coverage for the generated ideas.

A Facilitator in each focus group had built rapport; be direct; Use reflection; Probe with questions, be comfortable with silence or conversational lulls. Check-in with participants and Remain neutral. During the Focus Groups Ideation sessions, each facilitator had to:

- 1. Welcome the group & share the strategy they are using, such as the theme/ story/ challenge and apply the rules, including active listening, one talks at a time, respect, learn from each other, positive, criticise the idea and not the person.
- 2. Propose the small groups and amend if required
- 3. Manage the discussions (ensure it a dialogue and Not a Q-A session),
- 4. Observe body language.
- 5. Empathies with others, show understanding for every contribution, concerns with the participant's feelings, views, and attitudes, but should avoid answering or giving directions, or showing any personal feelings towards the ideas.
- 6. Record the ideas and appreciate every contribution.
- 7. Plan Wrap-Up, support the group to summaries the ideas and priorities and assign people tasks (when needed)

8. Designate who will be giving feedback to the facilitator.

After the focus group sessions were over, the facilitators worked to define the Problem Statement, which involves synthesising and making sense of all the available information the note-takers gathered during the focus groups by discovering connections and patterns.

The focus group scenarios were short and not multi-pronged or overly complicated; concise and concrete and did not use technical terms or jargon; conversational and open.

The advantages of the Focus groups that they make it easy to measure the people's reactions, and it usually provides immediate ideas and helps identify and provide insights into the current position. Its condensed nature makes it possible for any issue to solicit several opinions and feedback on multiple aspects. It allows researchers to expedite their researches and the findings (O.Nyumba, T., Wilson, K., Derrick, C.J. and Mukherjee, N.,2018).

The disadvantages of the focus groups that they did not cover determined depth compared to individual interviews. The members of the focus group may hide their honest opinions when they are in a group. The group may make them not able to oppose the views of the other participants. Besides, if we compare the focus group responds with surveys or questionnaires, it will appear that focus groups might not express as much as they do within an individual interview, and they need a longer time for setting the place and the logistics. On the other hand, facilitators might affect the results from the focus group if they insert their personal biases intentionally or inadvertently while the participants are discussing their ideas. It might result in getting the unneeded data and the wrong information that will not support the research in any way. The facilitator can also guide the focus group towards personal assumptions, and the participants may follow the facilitator's opinion by not disclosing their genuine and honest opinions (O.Nyumba, T., Wilson, K., Derrick, C.J. and Mukherjee, N., 2018).

Table 24. Summary of Focus Groups (advantages & disadvantages)

Disadvantages of Focus Groups
Not efficient in covering maximum depth
compared to individual interviews.
Members of the focus group may not be honest
in a group.
The group may make participants hesitant to
express their thoughts.
Compared with surveys or questionnaires,
focus groups are more complicated.
Facilitators' biases might affect the
discussions.
The bias can result in inaccurate data and wrong information
The facilitator can guide the focus group to
reach his assumptions.
The participants may follow the facilitator's opinion.

In this research, the researcher participated as a facilitator, and the focus group participants already knew from this that the researcher was gathering data and asking them to discuss and express their opinions regarding the development of the highly able individuals and the challenges they face. Moreover, the design of the focus groups was to gain a more comprehensive view of the opinions discussed during the discussions. The researcher analysed the collected data by coding to avoid note-taking bias.

For this reason, note-takers took notes while observing the session moving on, and not to leave out any comment or remark said by the people attending the groups.

These notes added to the facilitators' notes and remarks. The type of recording used was a descriptive observation of the discussions. They were in the form of notes and visuals.

Descriptively written notes helped to make the notes look like observations and allowed for an overview of what happened during the discussions while recording the notes in a structured way.

4.13.2 Data Collection Challenges

Four different categories describe data collection challenges; Researcher, data collection, participants and instruments design (Rimando, Brace, Namageyo, Parr, Sealy, Davis, Martinez, & Christina, 2015). Therefore, in this research, the most expected challenges were:

• The Researcher

-As employed in one of the researched institutions and the other researching (DOT) this would make it difficult to criticise the process without biases.

Participants

- -Some participants refrained from the interviews.
- -Participants did not disclose preliminary data.
- -Participants did not know about development programs and were not aware of how they work.

Instruments design

- -Some questions required explaining sometimes.
- -Some questions gave different meanings and allowed multiple answers.

• The nature of the subject

This subject is the first of its kind in the UAE. Investigating five educational institutions programs is challenging. This type of research faced different obstacles at the beginning, such as ignoring the researcher's request for interviews or delayed responses. Furthermore, and due to the researcher is involved in establishing the (DOT) at the (MOP), the similarity of the topic and the research approach made the researcher exert efforts to split the interviewing timing and types to fulfil both projects equally.

4.14 Data Analysis

Each qualitative researcher interprets data analysis differently according to (Punch, 2013), and this makes data analysis conducted in different methods. Punch believes that analysing qualitative data can be right in any way. However; this depends on the research purpose to decide the appropriate usage of the methodological framework. The methods of analysis should follow a systematic discipline, describing, and transparent way to allow the reader see how the researcher concluded from the data and to represent the participants' views objectively and impartially as much as possible (Punch, 2013). This method will make the reader understand the undertaken process in a transparent way. Therefore, data analysis section is a significant part in the qualitative type of research, knowing that data analysis carries more negative implications than any other single part of the research process according to (Drew, Hardman & Hosp, 2007). Data analysis described the steps that take the research from being specific to become general (Creswell, 2017). It ascribes that to the fact that in the qualitative type of researches, data is words, opinions, explanations, expressions and feelings and not numbers like the quantitative research method. (Drew, 2007).

Moreover, data analysis is what most differentiates quantitative research from qualitative research (Donald et al. 2006). Researchers arrange their collections of data in order to present what they obtained, found and learned through them to others (James, 2004). Researchers also arrange the data to separate each theme and to unit them later into operative units. There are different types of data analysis, and there is no model that has common characteristics for this process (Miles and Huberman, 1994).

4.14.1 Choice of data analysis

To analyse the obtained data from both interviews and focus groups, this required deciding upon the approach at first. The data collected from the interviews followed the same questions. They were accordingly easier to sort and code. On the contrary, the data collected from the focus groups. It happened because different note-takers performed the note-taking and the varsity of facilitators. The different approaches used by each of the note-takers and the facilitators resulted in obtaining more unnecessary data and information. Moreover, as the participants in each group were from different categories, this supported the differences in the approach and the quality of the collected data. The current research adopted the grounded theory analysis approach with its open coding in analysing the interviewees' outputs. Codes arise from data, and another secondary coding of the data processed. The data arrangement was in a way that helps the researcher create explanations and generate new findings, but without summarising them (Miles and Huberman, 1994). The researcher consciously revised the collected data and worked more on reducing them by illuminating the repeated answers. Grouping of the data was by (Mural) application to focus on the main themes, find the patterns in answers, and select the responses that help to answer the research questions. (Miles& Huberman, 1994).

At the first stage of coding, the data coding was according to the categories of the participants. It helped at a particular time to know if they could answer the research questions. It also helped to clarify which type needs more interviews. However, in the second stage, the coding changed to be according to the themes. The themes become more explicit and more focused on the answers required for the research questions. It allowed the researcher to include any new ideas that came from the data of the analysed interviews. Codes were the coloured labels that assist the researcher in assigning units of meanings.

In this study, the researcher used the Mural digital program to aid the analysis of data. Computer software packages make data analysis easier and support the storing and locating of qualitative data in an organised way (Creswell, 2017). The ability to store and analyse the obtained data helped in discovering their relationships within the required themes. The usage of Mural technology to explain and relate the data was a helpful method. The organisation of the collected data and having them in one place helped to organise and manage the study in a more natural way. It gets things running very smoothly and creatively.

There were (5) types of head codes under the themes created from a list prepared before the fieldwork. These codes were identification, development, awareness, challenges, and participants' recommendations.

The researcher divided the page into four sections according to the main categories used in the interviews. Notes types differentiated by colours. Through data analysis, the main theme code was the purple label that centred on the heading page. Other subdivisions were the yellow labels that represented the quotes, or the issues derived from the interviews' dialogues and focus groups' discussions. The orange label was the pattern, which refers to answers or opinions repetitions during interviews and focuses groups. The blue label contained the questions derived from the interviews. For example, the Identification label explained the issues related to identifications and any quotes derived from the interviews. The purpose of this data was to verify the findings.

Table25. Summary of stages for data analysis

Data reduction	Data Display	Verification	Data Coding
Collected data revised, reduced, and coded. The reduced data in a separate folder	Tables and charts used to display the data in an organised and transparent way.	Continuously examining the implemented process and referring to the notes and interview papers to develop conclusions.	Themes identified and organised sequentially. Each theme assigned a code that distinguishes them by a category or a label.

4.15 Conclusion

In conclusion, this chapter discussed the methodology of this study and outlined the constructivist approach that was the philosophical position of the research. The researcher gave a rationale for the qualitative methods; ethical considerations explored, with a description of the processes used to collect data and to analyse it.

Chapter Five. Presentation of the data

Introduction

The purpose of this study was to answer its questions that wanted to explore the impact of the development of the highly able students by five Governmental Educational Institutions' in the UAE. It aimed to suggest the applicable recommendations for improving their services in the future. The study also intended to investigate the beliefs and perspectives of different stakeholders who are involved in the study. These stakeholders are the people around the gifted and talented students, and they are influencing them. Stakeholders in this study include program developers, experts, teachers, and parents. The investigation was to know how gifted and talented are defined, and the identification and development methods and processes applied to achieve the goals of education in the UAE. The researcher used multiple methods to draw data and obtain the required results, including an interview, focus groups, and document review.

Table 26. Alignment of data sources, participants and research questions

S	RESEARCH QUESTION NUMBER	SOURCES OF DATA	PARTICIPANT GROUPS WHO COMMENTED ON THIS QUESTION
1	What is the role of development programs implemented by governmental, educational institutions to develop gifted and talented students in the UAE?	Interviews Documents Focus group	Gifted and talent experts -Managers- Leaders-parents-gifted and talented students /adults.
2	How are gifted and talented students defined and identified in the UAE?		Experts in gifted and talented. Managers-Leaders-parents- gifted and talented students /adults.
3	How is the development of gifted and talented students implemented in the UAE?		Experts in gifted and talentedManagers-Leaders.
4	What are the main challenges facing the gifted and talented identification and development in the UAE?		Experts in gifted and talented.

5.1 Results from the Interviews

As discussed in the methodology chapter (3), the interviews intended to add to the findings by document review and the other method of focus groups. The conducted interviews on six different categories using four types of interview questions to guide the discussions. The discussions covered some (96) participant who constitutes stakeholders in the life cycle of these highly able students. Following is a summary of the interviews' participants.

This study will not disclose the participants,' and it protects their anonymity.

Table 27. Participants in the interviews

Participant Group	Code	Number of participants
Experts including psychologists	E	6
Teachers	T	6
Managers	PM	9
Parents	P	14
Young talents	YT	26
Adolescence talents	AT	35

As shown in table (27), (96) participants contributed to the interviews that aimed to obtain the opinions and answers from different categories in the ecosystem around the highly able students. The selection of categories included several individuals related to this area, and the choice depended on their closest relation with the gifted and talented; therefore, they were a mix of experts (6), teachers (6), managers of programs or institutions (9), parents of gifted and talented students (14), and gifted and talented students and adults (61). The considered choice was to obtain a broader scope of opinions as the participants in the interviews are from different backgrounds and different educational institutions. They had long experiences in the field of gifted and talented in and outside the UAE, and they are familiar with the UAE educational system and environment.

Some of the selected participants are experts and teacher who perform the identification process and implement the development programs for the highly able students. Parents are the most significant stakeholders in the identification and development of these students. Their role is essential in the first stage and at a very early age. They are the first to realise their child's giftedness. Therefore, the most critical data and information come from parents regarding their children's behaviour and development. Parents of these highly able students are those who decide on whether to allow their children to participate in development programs or not (Colangelo & Dettmann, 1983). Other participants comprised experts, including psychologists, teachers, and managers aware of the development programs implemented in the UAE by the educational institutions. The study aimed to investigate these educators' experiences on discussing how these governmental, educational institutions implement their various gifted programs in the schools of the UAE. The study also wanted to explore the teachers' professional development and how do they train to implement development programs for this category. Participants in the interview were highly able students selected from different ages and stages. The interviews intended to solicit their views and to obtain an informed perspective and a broader view. Their participation was essential to know to what extent these institutions' development programs are influencing and motivating them. These participants could answer the most critical questions regarding recent emerging trends and concerns in the development programs because they represented different educational entities across the UAE. The most significant number of participants were from the direct stakeholders to give trustworthiness to the study as their responses will shed light on the actual status of development in the UAE.

The interview questions consisted of a structured approach designed to explore more responses and in-depth information concerning the study topic. The differentiated questions guided the conversations with the different categories to answer the research questions. The data obtained from the interviews were within the themes, which in return helped to guide the researcher in the findings and recommendation sections.

5.2 Findings from the interviews

For consistency and clarity, the following section will display the main themes obtained from the interviews' analysis. The themes provide a more general view of the results and ensure drawing the main findings from the collected data.

5.3 The Interviews Analysis

Four main themes emerged from the interviews and focused groups.

Summary of the findings from the interviews according to the themes:

(1) Knowledge and awareness

Research question (1) -what is the role of development programs implemented by governmental, educational institutions to develop gifted and talented students in the UAE?

Table 28. Interview questions in alignment with the RQ(1)

	Do you think the talented youth in the UAE are ready to adapt to	Experts-
	the postgraduate requirements and the job market needs?	Teachers-
		Managers
_	What opportunities are there to promote talent investment in the	Experts-
	UAE?	Teachers-
		Managers
Knowledge &	How well do you know about gifted programs and talent care	Experts-
Awareness	programs in the UAE?	Teachers-
		Managers
_	Have implementation of the development programs contributed to	Experts-
	the development of talents in the UAE?	Teachers-
		Managers
_	To what extent do you know about talents program in the UAE?	Parents
_	To what extent you think the UAE student is ready to travel and	Parents
	study abroad?	

	How did you know you are talented?	Gifted &
		talented
	How do you want to benefit from your talent?	Gifted &
		Talented
_	In your opinion, what qualities of talent that facilitate their	Experts-
	identification?	Teachers-
		Managers
_	What are the characteristics that made you recognise your child's	Parents
	talent?	
_	How can talented people be supported in terms of developing their	Experts-
	talent / connecting them to the labour market/universities?	Teachers-
		Managers
_	How your child's talented abilities are invested?	Parents
_	To what extent did the UAE support your talent development?	Gifted &
	,	Talented

Table.29. Interviews' results in alignment with (RQ1)

Participants

Knowledge and awareness

č	1
The family & community do not have adequate knowledge about the importance of giftedness and talents, which negatively affects their support for their children.	Parents
In several interviews, parents, gifted and talented students are not aware of the existence or the importance of a policy that regulates the identification and development programs.	Parents, gifted and talented students
Parents prefer and encourage their children to look after their academic programs	Parents
over artistic talent programs due to the insufficient understanding of the significance of performing talents for the UAE national and international profile.	Experts, teachers, and managers.
The reflection of society's perspective was apparent in the interviews. The interviewed from all categories and different ages or types of gifts ad talents agreed that the community still finds art talents as a complementary component, and it is just an entertainment category. There is no investment directed towards artistic talents, according to them.	Parents, experts, teachers, managers, Gifted and talented students and adults.
There is a lack of adequate guiding and mentoring given to the students on aligning their career options with the needs of the workforce. It is also another challenge in awareness reflected in the interviews by all categories and several partners.	Parents, experts, teachers, managers, Gifted and talented students and adults.
In some interviews, there was consent among gifted and talented females regarding males are still dominating their parents' concerns. However, there is no appreciation for their artistic talent.	Gifted and talented females.

The question aimed to explore the role of these five selected educational institutions in the UAE, in implementing development programs that have influenced the emotional, cognitive and skills development of the highly able learners.

It also explores the extent of knowledge of people in the UAE regarding the availability of different gifted and talented development programs, how these various programs would be able to nurture different domains of talent as well as their ability to evaluate and assess their achievement. A general question in the interviews was about the definition of the gifted and talented from the perspective of the participant. The information in this regard gained from the experts and specialists who are familiar with development programs and how they are implemented were beneficial and informing. However, it was almost challenging to obtain such information from the students or their parents who are not able to understand the development process in-depth and do not usually give their insights as the experts and specialists do in these issues. The interviewed experts, teachers, and some managers were generally familiar with these developmental programs, relying on their overall experience in this field. They pointed at the lack of definition that leads to missing awareness among families and people, including many educators as well. They indicated other problems in development programs that make them unqualified to meet the needs of these highly able students. Some problems referred to the lack of configuration and the inability of some imported programs to support the differences in the environment or the different needs of these learners. Some managers were unwilling to participate or disclose any information. However, many other experts and teachers expressed their theories about what they think regarding educating and the development of these students along with their parents who should be more knowledgeable and aware of many issues related to their children's development such as the definition, identification, and developmental programs and the issues of funding. For example, an expert at one of the institutions' said:

"Many parents of gifted and talented students are not concerned, not even worried about having a law or a policy that regulates the identification and development of gifted and talented

students. They do not view the policy as an essential tool or an instrument that protects their children's rights to receive their needs.

According to the expert, there is no awareness regarding the importance of having such a policy because many of parents and sometimes teachers assume that:

(1) Gifted and talented are not in need of any support, (2) gifted and talented can help themselves if they are gifted and talented (3) gifts and talents for them constitutes academic achievement only. (4) their children need more education and homework, and they do not need development or enrichment programs. They usually point at people who succeeded on their own, and they view

Another finding that emerged from the interview with a program manager:

them as role models to follow."

"Many schoolteachers, including some managers, showed little awareness about the content of a gifted and talented policy. They even do not realise its capacity to regulate the gifted and talented needs such as development programs' or any other initiatives that aim to nurture the gifted and the talented. The behaviour of teachers in this issue widens the gap of awareness and knowledge among parents and the gifted and talented students as well. However, some of them thought it was easy to guess what it entails based on their understanding of the policy of the special needs."

The interviewed experts were very much aware of the impact of not having a national policy that would organise the gifted and talented definition, identification, and development programs and enable them, as well as impose training and development on those teachers who oversee implementing such programs and dealing with this category. A few participants understood the national concern about the extent of the developmental achievement of these programs and

assumed that having a national policy would help to unify the definition and the other institutions' activities." It will regulate the identification and development programs and enable them to achieve better and higher results. "

Considering the gaps in the programs requires identifying the enablers. The enablers are the strategies, methods and professional development used to make these programs more beneficial and efficient for the end-users. The problems in some of these programs referred by several participants to the absence of a regulating policy; some participants expressed disenchantment with the existence of multiple institutions that oversee implementing gifted and talented developmental programs. This dissatisfaction, according to them, refers to the negativity of the non-collaboration that would reflect on all of them. The idea of supporting a non-collaborative system enhances undesired systematic competing and lessens the benefits desired.

A talent expert, who is also programs' developer, commented to that by saying:

"I know that every institution in the UAE has implemented different programs and initiatives to support gifts and talents' development in the different emirate, but I know that there is a need to equalise and regulate these programs under one national policy and within the context of the UAE. There are no enablers to empower these programs and allow them to work well in a sustainable manner, we are blessed to have several types of the latest trends in gifts and talents' developmental programs, all of these are able to support them, and they receive fund from the government, but we do not follow programs assessment to ensure their continuity and improvement, we don't have either many qualified experts or teachers to empower these programs or conduct their evaluation".

Participants in the interviews were mostly unsatisfied with the misrepresentation of the programs in the UAE. They were also unaware of the existence of many of these programs and any benefit from them. The parents, gifted and talented students, were uninformed about what to do in order to overcome the challenges faced by them regarding skills training and development issues. One expert talked about a new policy intended to launch by the (MOE) as the first of its kind. It will be the first national policy in the UAE to regulate gifted education, identification, and development. The expert said:

"An international team of specialists experienced in gifted and talented issues revised the policy along with many other educational institutions in the UAE. They all were involved in the review sessions to finalise this policy.

However, and because the (MOE) did not announce or publish this alleged policy, it is not for discussion in this study.

Another expert who works on the development of educational strategies and policies said that the national underdevelopment policy would only focus on talents in the field of culture.

"Unfortunately, and in my opinion, this is not the federal policy that has been long waited to cover all types of gifts and talents and regulate their ecosystem in the country."

Another expert explained her view regarding having the importance of having the policy to regulate the gifted education and development programs in the UAE.

"Our main concern is about having a policy that mandates having an ongoing training strategy to qualify not only teachers and enable them to perform the proper implementation of gifted and talented identification and development programs, but in addition to parents, who need to be familiar with such programs and should be a solid part in their gifted and talented children's

development. Once we make parents aware of the methods and we give them the directions and how to identify and develop their children, we save much time, and we stop the leaking of gifted and talented students due to this ignorance. We want to regulate the functionality of this training. We do not want to have a policy that will not include all types of enablers such as parents and accredited packages of comprehensive programs with tools related to all types of gifts and talents as well as teachers and skills."

Several teachers use their limited knowledge obtained from their sole experience in regular teaching and general training, but this was not enough and had an effect of transferring poor awareness and understanding that is disadvantaging the more recently qualified teachers and preventing them from practising and applying their skills. It is demotivating gifted and talented individuals and reducing their enthusiasm as well.

An expert who was involved in the development of several educational policies more than once said:

"Gifted and talented programs cannot achieve their objectives towards spreading awareness and changing the attitudes of people in the community regarding the importance of gifts and talents and their development if teachers who are in charge of the implementation of these programs are not qualified enough and do not have enough literacy for the task. What will the teachers do if they do not have the basic required knowledge on how the program function and what they should expect from it in the long run? How will they implement any program if they do not know that they have to assess its functionality and progress to measure its success and achievement? How can they provide the gifted and talented with their needs for development if they do not have any background about the basic gifted and talented needs? How can a program

fulfil that need and nurture it if the teacher is not aware of many aspects of development or not fully convinced about the importance of development for gifted and talented students?"

Finally, and from the above said data, extracted from the interviews, not only teachers need professional training to implement development programs and gifted education, but parents also need intensive training to understand the concept of identification and why their gifted or talented children need development. They must attend continuous and regular courses to exercise and know the process of identification and development for their highly able children and to be online with the occurring changes in fields of education and development. Parents should help teachers as well to identify their highly able children and to select the type of program that relates to the gifts or talents and suits their abilities. Teachers' awareness and knowledge regarding identification and development processes is a challenge that can harm the entire ecosystem surrounding the gifted and talented if they were not sufficient and adequate for their jobs.

(2) Definitions and identification of Giftedness & Talents Research question 2, How are gifted and talented students defined and identified in the UAE?

Table 30. Interview questions in alignment with RQ (2)

	What is the talents' definition from your perspective?	Experts-
		Teachers-
		Managers
		Parents
		Gifted &
Definition &		Talented
Identification	How did you discover your child's talent	Parents
	What are the identification tools that are used to detect talent in your	Experts-
	institution?	Teachers-
		Managers
	How did you come to know about the talents' identification	Parents
	programs in the UAE?	

Table 31.Interview results in alignment with (RQ2)

Definitions and identification of Giftedness & Talents

The participants, in general, differentiate their definitions regarding (giftedness Parents, experts, teachers, and talents), and they do not know if there is a unified certified meaning for these managers, categories from any of the (5) institutions. Gifted and talented students and adults. b) The majority of participants (parents, gifted and talented) were confused Parents. regarding how to define talents and differentiate them from hobbies. Gifted and talented students. c) The other confusion among (parents, some gifted and talented students) was in Parents. differentiating between academic giftedness and artistic talents. This confusion Gifted and talented students. was due to the absence of an official policy that provides them with a definition that identifies each of them differently. d) The absence of policy was also one of the main issues mentioned by all experts, Experts, teachers, managers, and gifted and talented adults. The policy would provide the Teachers, managers, and

gifted and talented adults.

institutions with the appropriate unified definition of gifted and talented

This question explores the description of the gifted and talented and the identification programs implemented for this purpose in the UAE. This analysis indicates that the definitions of the talented and the gifted in the educational institutions differ, and accordingly, this assumes that the identification programs vary to suit the definitions descriptions.

A-Definition

individuals.

One of the interviewed experts indicated:

"To derive a definition of gifted and talented, we researched more than 21 definitions and derived our own from them. However, declaring this definition is not possible yet because it is in the new policy, which should undergo more intensive reviews before publishing it or discussing its context in this interview."

All participants, in general, answered a unified question in the interviews on how they define a gifted or a talented individual, and where from did they learn about this definition. Experts, teachers, and managers said they represent their definition following their institutions that derived the description with the cooperation of researchers and experts who visited the theories in the field

of gifted and talented education and development, the values and various concepts applied in the UAE such as innovation and creativity to obtain their unique, specialized definition.

According to this research, giftedness is the skill and the abilities in the academic and scientifically domains, while talents are the performing skills in the cultural and artistic realms.

The mixed views regarding the differences between talents and giftedness reflect the amount of ambiguity in them. An expert who develops identification and development programs explained his opinion in the interview by saying "

"Some teachers do not see or believe in the differences between the two terminologies because they do not have enough knowledge. In his opinion, this was not a surprise because teachers require intensive continuous training and more in-depth readings and researching to grasp some knowledge about this complicated area."

Another manager explained his opinion in this regard by saying:

"Some teachers and even some parents do not like this differentiation because it puts more effort into them when asked to nurture or support their children in any way. However, and regardless of the manager's opinion, some interviewed parents for this study were in favour of this differentiation because this helps them understand why their child is underachieving academically while they are showing other amazing abilities in arts or sports. However, this does not change the parents' perspective on how they view gifts and talents. They accept it if their children are talented in music or arts, for example, if they score high marks in their academic subjects."

Another expert with four years of experience in this field defined talents as:

"The gifted person is the one who can think outside the box and think in an unfamiliar way. The characteristics of a gifted that he/she relies on self and does not return to any official (parents, teacher, coach). The gifted often introduces a new and a right decision that differentiates between what is right or what is wrong, even in the provision of solutions."

An expert who oversees strategic planning of development programs said:

"A talented individual, in my opinion, is the one who enjoys extraordinary abilities that develop and become an investment.

When it comes to defining gifts and talents, parents had their definition that reflected what they culturally know and experience from their tradition, community, or children's behaviour.

A parent of a gifted child said:"

His mother and I could see his talents because we know that a child who makes something different from his brothers or peers is a talent. The talented children's work will be different, and it will shine by itself."

A teacher said, "For me, talent is the ability to perform and deliver a mission to make people happy and satisfied, especially if this favour affects the underprivileged people. The talent is what makes our work unique, and people love it."

A talented musician defined talent as:

"It is what I am passionate about doing all the time, and what I want to be more skilled in doing it every day."

Accordingly, there is a consensus among experts, managers, parents, and gifted and talented that the definitions at different institutions are very narrow and do not reflect the future's needs. There

is a need for a more broadened and flexible definition adopted by all institutions and at the same time used to interpret all types of talents. Another consensus among the interviewed participants that there should be a more clarified definition, which differentiates between giftedness and talents. These differences should be more transparent for the regular people who are not experts and do not prefer scientifically complicated explanations to use each definition.

Another concern is about how the different definitions do not reflect the reality about talented and gifted abilities away from their skills. The confusion of definitions has made some of them define the skills of the students while others described the competencies and not the talents.

Additionally, some students do not find what describes their abilities included in the definition. Although such a shortage does not mean that they are not gifted or talented, it might confuse them and hinder their development that requires planning according to their talents or gifts needs besides, the definitions of the gifted and talented cannot be a sole decision taken by any person. It is the role of a committee that consists of teachers, experts, and managers in educational institutions.

B-Identification

An expert who worked as (Head of the gifted programs department) for ten years indicated:

"The talented person is identified through several methods. First, parents, friends, teachers, schools, and the ministry of education identify him/her through the forms of talent screening (nomination, direct submission), and objective achievement tests. These tests will recognise the identification of the tested giftedness. There is no doubt that there must be an objective sequential program aimed at the talent to be refined and highlighted correctly and to receive support through the mentor who may be able to turn any gifted student into a scientist, thinker, researcher or an expert."

Another expert who worked as (Administrative of tests and measurements) for more than ten years said:

"Talents need nurturing after identification. Using appropriate test tools to discover their talent, helps them to go through varied training programs. These programs include (lectures, games, issues, activities, trips.) The differentiation of methods is beneficial, so the talented individual does not feel bored while he/she wants to practice things out of the ordinary. It is necessary to provide a development training program for all of the parties around talented individuals and a rewarding system to provide a motivating environment that suits his/her development needs."

An expert who is a psychologist pointed at the identification programs or the used methods to identify the gifted and talented students in schools and even at home. She said,

"These tools or techniques might be sometimes misleading or do not prove to work well. What makes her say that is due to her experience in the field, and what she has been facing for a long time? She noted that some children might suffer other problems such as hidden physical problems in hearing, sight, or even psychological issues that impair their achievement. Their physical or mental difficulties prevent them from declaring their talents or gifts if they have any of them. They usually feel restless, confused, and sometimes bullied by others. Such problems would make them look like underachievers'. They hide their gifts or talents if they were gifted or talented.

A manager who is in charge of monitoring the identification said:

"The identification tools might not be accurate if most of them are very easy to the extent that they do not reflect the real abilities of highly talented students. Some problems in the identification cause wrong enrollments for those who are untalented. They may enrol in a talented program, but

they would not continue in them. The other problem here is parents. They will not understand that the identification was wrong, and they will misunderstand the issue or the institutions' intentions. The focus on (Math and Science) as a measurement criterion is not fair, and it can neglect other hidden talents. The language is also an obstacle as many of the students are not good in English; the language used by some introduced identification programs is English, and this causes them to be unidentified as talented.

Finally, experts in the interviews, in general, were adhered to their institution's definition and identification methods. However, not all of them were satisfied with how they describe or clarify the meaning of talents. The consensus among the interviewed regarding the conflict between giftedness and talent was apparent. On the other hand, most of them were not able to explain or even to define the differences. Experts' participation in the formulation of a definition is critical. However, most of them admitted that they did not share in this formulation, and only a limited number of experts formulated the definition. Many experts, teachers, managers, parents, and even talented individuals were not aware of the existence of any descriptions, and the importance of the definition. Most of the teachers did not even know that there is a definition for talented and gifted individuals, or they did not memorise it. The identification tool, on the other hand, constitutes a challenge for experts, teachers, and managers in the first stage. Many of the exported international identification programs need customisation to suit the highly able students in a differentiated environment and to align with the strategies, visions, and national identity and agendas in the UAE.

(3)-Development & support

RQ 3. How is the development of gifted and talented students implemented in the UAE?

Table 32. Interview questions in alignment with RQ (3)

Development &	What factors influence talent development (examples)	Experts- Teachers- Managers
Process	How do you develop the gifted programs?	Experts- Teachers- Managers
	What do you/ school do to develop your child's talent?	Parents
	What supported the development of your talents?	Gifted & Talented
	What did you achieve -benefit from your talent?	Gifted & Talented
	Did anybody encourage you? Who and how?	Gifted & Talented

Table 33. Interview results in alignment with RQ (3)

Development and processes	
a) The current talented and gifted provisions and development practices, according to experts, teachers, and managers are random and not systematic, and not based on an institutional philosophy.	Experts, teachers, and managers.
b) There is no overall institution, strategy, or plan to regulate giftedness and talent development practices, as mentioned by all experts, teachers, and managers in addition to gifted and talented adults.	Experts, teachers, and managers.
c) No policy supports the implementation of qualified development programs. No system that links the current practices together. No policy provides guidance and regulations on the processes and procedures.	Experts, teachers, and managers.
d) No gifted and talented centres provide gifted and talented students with a conducive, stimulating, and challenging learning environment according to all participants from all categories.	Parents, Experts, teachers, managers, and gifted and talented.
e) No collaboration and coordination between the institutions to provide a qualified system that regulates the usage and efficiency of gifted and talented programs and initiatives, according to experts, teachers, and managers.	Experts, teachers, and managers.
f) The current gifted and talented practice is not sustainable and does not support gifted and talented students and teachers' needs in the long term, according to all participants from all categories.	Experts, teachers, and managers.

This question explores the development programs implemented for the development of the highly able students in public and private schools in the UAE.

The knowledge about the development of these highly able students' and the different programs implemented in and outside the schools to ensure this development and to make it an ongoing appropriate achievement is an important issue that affects the whole process if not applied in the right time and manner.

Finding out to how extent participants know about development programs and the role of these developmental programs, the interviews included specific questions aimed to explore the participants' awareness and knowledge in this regard.

The information gained from these different categories who work in the field of gifts and talents' development or related to it resulted in realising how some experts were mostly unfamiliar with the other institutions' identification and development process and procedures. Most of the interviewed teachers indicated that they rely on their experience to nurture the highly able students. However, they occasionally do not know if they satisfied their potentials equally.

For example, an expert who works in one of the educational institutions said:

"We have plenty of gifted and talented developmental programs and initiatives that cover all educational levels in the UAE. However, they are not empowered with enough experts, teachers, and criteria to make them sustainable. Most of them are temporary and unsustainable, they stop or pause, and sometimes, they continue without an aim. "

Interviewees revealed critical issues regarding the development programs, their creation, and implementation procedures. There were common consents among the interviewees who agreed and consent that these educational institutions have already created and implemented numerous

numbers of gifted and talented development programs for various domains of gifts and talents. Though, and because there is no ongoing assessment nor continues professional development available for the teachers, it leads to a lack of awareness about the programs and their ability to support the gifted and talented. These issues would appear in the form of a shortage in identification and development programs despite many programs are available. It prevents providing any support or the ability to implement the right decisions that will ensure their development sustainability and will empower them with the appropriate enablers. The majority of teachers nominated to teach and look after these gifted and talented students are teachers who work in schools and implement the regular curriculum and the traditional teaching style and system. Therefore, such teachers, who are not experienced and do not have long and continuous knowledge in dealing with highly able students would negatively influence the process of identification and development and minimise its effect. The assessment itself is sometimes not appropriate and does not match the gifted or talented student who is coming from a different country. If the institutions implement the development programs according to the state that created the application and developed it in coherence with its culture, citizens' needs, values, and its national agendas, this will end up in failure in our country or any other state, and the program will not achieve anything. The findings as well highlighted the massive shortage in the key enablers that support the implementation of the development programs and minimise their benefit. Such an issue might not be evident nor addressed at present. However, their negative implications will emerge gradually in further extended periods. According to experts, most of the implemented development programs work consecutively for a while, but then they discontinue once they face any issue. The development of highly able students requires sustainability and continuity to reflect on their

development success. There should be teachers with knowledge on how to solve any obstacles, and they are professionals in implementing such programs.

An expert in the field of giftedness said that one of the observed challenges during the implementation of the talents' development programs is not being encouraged or welcomed by the highly able parents. The majority of parents do not appreciate gifts and talents in arts or any domain other than the academic field. They do not view the highly able children as an investment or assume their success in future. They think talents' development programs are just activities that will not educate their children or secure their jobs in future. The regular school curriculum, in their opinion, is the only opportunity by which their children can secure their future.

One of the experts said:" Another problem that faces any developmental programs is the lack of understanding and awareness about the meaning and the concept of the development of gifted and talented, and how this development should reflect on the surrounding environment in the community. This problem caused the drop out of countless students who were not encouraged by their parents and were not influenced to continue in them." In addition to the above, the expert said:" These students should learn how to arrange their priorities and manage their timing. They need to learn to support themselves and manage their lives and their emotions. They want to know how to grow and progress with their talents while going on with their regular lives. Parents could mess their lives, but it is all because they are not convinced that gifts and talents would support their children's future and secure them jobs. "what I see as the most significant challenge". "This challenge makes the gifted and talented development very difficult, and sometimes it ruins the talent itself".

An expert indicated:

If the high ability students do not get training, this will negatively affect their personalities.

"It can turn out to be the most critical implications of introversion and isolation. There are substantial general observations that directly affect gifted and talented individuals. Lack of mentoring and development programs is one of them; the absence of a teacher rewarding program; the teachers' focus on providing the curriculum and not to train students on how to use their skills to understand the subjects; the absence of a management system that regulates gifted and talented students' education from early school life. The expert as well pointed at the existence of the traditional conservative thought in the community that views the scientific approach as the best approach for a career. The non-diversity in talent development programs make them unchallenging and dull, as well.

A mother of a talented girl who arrived in the UAE two years ago said:

"My daughter was able to develop her talent in (street dance) in an academy in Finland, but now, and as we moved to Dubai, it is difficult to find a specialised academy to support her talent.

She is developing her talent with the support of YouTube."

This mother was unaware of the existence of any developmental programs in the UAE. She criticised her daughter's school for piling her with homework, which is distracting her from developing her talent.

"It is a private school. However, it lacks the equipment of many instruments and tools that could help my daughter develop or satisfy her talent."

Another mother indicated that the parents, who come to know their children are gifted or talented, begin to live and see daily nightmares because of this knowledge. It is oppressing because they do not know how to treat them or if they need to do anything for them. They are afraid that they may ruin their talents if they interfere in it.

A father of a grade 12 son agreed with her and confirmed what she said by adding:

"The stress begins when they are unaware of what to do after they discover that their child is gifted and talented and how to deal with this information. Life becomes more difficult and challenging for parents more than their children do. To know or not to know, both are distressing.

A mother of a grade eleven son and a grade one daughter said:

"I usually expose my children to different experiences, and I spend time to observe their behaviours. I used my experience with my elder son, who identified as talented in his school to encourage my daughter, too. Now I assume that she may exhibit any talent, as I believe that abilities inherited. However, she is still very young to know or to explain her desires, and I am the one who keeps thinking if I will succeed with my efforts exerted to make her display any talent.

A mother of a male student who is in grade ten said:

"My son has a passion for helping countries maintain a healthy, rubbish free environment. He established a group called: (Green 4 Life). He acted like a leader when he gathered several friends and used to go to the neighbourhood to collect the plastic bottles, glasses, cardboards, papers, and any recyclable items where I would take to the recycling centres every Saturday to support him. In Amman, our country, he discovered a group called (Green Wheels) that attracted

his attention and passion. The group's main task is to collect the plastic bottles and the leads, and they categorise them and send them somewhere where they use them to make wheelchairs for sick and disabled people. My son loves this kind of work, and he can gather a considerable number of people and can convince them to work with him. In Amman, they called him the 'Green Wheels Ambassador, but now, I do not know how he will manage his life with his leadership talents and where he could go to start his ambitious projects.

She continued saying:" I thought the school could offer my son any support to flourish his talent, but this was not true. I had to move my son to a more expensive school because his former school does not offer any growth opportunities for talented children like him. I do not have access to any identification tools to have a reliable insight on his abilities; there are no opportunities apart from having private music or sports lessons and training in a club with expensive charges.

A father of a gifted boy in math and robotics said:

"My son's former school had no provision or services for gifted and talented students, but the new school has after-school activities, and my son enjoys these activities. I noticed now that he mainly focuses on Math. He is also passionate about robotics. He used to complain about how boring the math classes were in his previous schools. Now he loves it, and he is doing very well in this school.

A mother indicated when asked what she knows about talent development programs:

"Not a lot, but now I am seriously thinking of opting Home Schooling, where I can spend time with my children to make them responsible for their learning."

The results obtained from the interviews show that the development of talents operates through several initiatives and programs in the fields of science, academics, and culture. However, these newly launched initiatives could not reflect the talents' development progress yet. Similar development programs require ongoing evaluation to clarify the program plans, improve communication among partners, and gather needed information for program improvement and effectiveness. (Milstein, B, Wetterhall, S, and Evaluation Working Group, 2000).

However, it is not difficult to assume the existence of several difficulties that face the application of these programs, such as (1) the shortage in the enablers that would support the program's success in the future. (2) The lack of specialists in the field (3) the shortage of training given to gifted and talented teachers, trainers and parents. (4) The absence of coordination between the related institutions, which makes it challenging to learn about other talents' development programs in other institutions.

Another finding that emerged from these experts' dialogues was that teachers and managers in some schools have little awareness of the existence and importance of talented programs and initiatives process, although some of them thought it is easy to guess what they entail based on their knowledge of other teaching programs in the regular classes. Experts were aware of the impact of not having a national policy. A few participants were aware of the national concern regarding these developmental programs and their achievements. They thought that having a policy for gifted and talented would help regulate the applications of these gifted programs. It would also enable them to achieve better and higher results. Some participants expressed disenchantment with the existence of multiple institutions in charge of implementing talented developmental programs. The lack of enablers and the nonexistence of a regulating policy would negatively reflect on all of them and support a non-collaboration that enhances undesired systematic competing.

An expert who works in an educational institution said:

"There are plenty of talents' developmental programs and initiatives that cover all aspects of talents and all educational levels in the UAE. However, because they are still new, they are yet to have measurements taken accompanied by the qualified enablers nor any accredited accelerators, which will make them a sustainable, fruitful, and measurable program."

A talents' program developer expert commented by saying:

"I know that every institution has implemented different programs and initiatives to support talents' development in a different emirate, but I know that there is a need to equalise and regulate these programs in the UAE. There are no enablers, we have all types of talents' developmental programs for different domains of giftedness and talents, but we do not have sustainable programs to train qualified teachers and enable them to train other teachers as well and so on continuously."

Another finding that emerged from these interviews was the unavailability of qualified teachers to identify and nurture talented learners in schools.

A manager in an educational institution said:

"There were some teachers selected to enrol in an intensive Diploma program that is offered by Hamdan Bin Mohammed Al-Smart University. This university as well has signed a partnership agreement for the implementation of the vocational diploma program in gifted education. This program aimed at training and developing a selection of teachers who were chosen from the government schools as teachers in the field of giftedness education and at rehabilitating a distinguished Emirati teacher in the process of detection, care, and development of talented people."

The researcher found that this program includes (20) teachers every duration and runs for five months. It represents an advanced step towards the formation and the development of human capacities and capabilities that are necessary to build a scientific and practical system.

Managers in some of these institutions showed little awareness about this diploma's achievements. Few experts had a piece of knowledge about the national concern on the accomplishment of this diploma, and some participants expressed their disenchantment with this diploma because it is limited to a small number of teachers.

According to an expert:

"It also selects teachers from regular education and special education to train them for (13) hours only, which is not enough to make teachers able to identify and nominate who is talented and who is not."

Finally, looking at what the interviewed indicated and what results obtained from them reveals an issue in the development process, strategy, and enablers. Such results will constitute the recommendations in the recommendation section.

(4) Challenges

RQ(4) What are the main challenges facing the gifted and talented identification and development in the UAE?

Table 34. Interview questions in alignment with RQ (4)

	In your opinion, what are the most obstacles and challenges facing talented people and their families?	Parents
Challenges	What are the challenges you face in raising up your child's talents?	Parents
	What are the challenges you faced in the UAE with your talents'	Gifted &
	development?	Talented

Table 35. Interview results in alignment with RQ (4)

Challenges

a) The support given for the subjects related to arts and cultural talents is very little and not encouraging according to all participants from all categories. b) There are rare scholarships granted for artistic talents but plenty to the academic subjects only according to all participants from all categories c) Lack of qualified instruments for identification, lack of trained and qualified teachers to identify gifted or talented learners, according to experts, teachers, and managers. d) Talented and gifted students can study one speciality in the university, which prevents them from considering the speciality of their choice as a talent, e) Non-academic subjects are costly because they are not available in public schools as given by private schools. f) The lack of differentiation in teaching methods in schools makes talented students forced to underachieve in their academic subjects. g) Curriculum provision: the curriculum taught in schools is inadequate to fulfil the gifted and talented students and the differentiation of the curriculum is unable to accommodate all types of talents, and there is no differentiation of the curriculum is unable to accommodate all types of talents, and there is no differentiated or specialised qualified curriculum for talented and gifted students. h) Training of skills: Teachers at schools teach subject content only without paying attention to skills. They do not even know how to train skills. There is no attention paid to develop the students' abilities; students do not understand innovation and innovative products. Teachers' skills do not undergo training, and they are mostly not qualified to discover or identify talented students in their classes. 1) Human capital: the participants raised the lack of specialist professionals to identify or to provide counselling, coaching, and training for gifted and talented sudentes and talented and managers. Parents, experts, teachers, and managers. and managers. Parents, experts, teachers, and managers. Parents, experts, teacher	Challenges	
subjects only according to all participants from all categories c) Lack of qualified instruments for identification, lack of trained and qualified teachers to identify gifted or talented learners, according to experts, teachers, and managers. d) Talented and gifted students can study one speciality in the university, which prevents them from considering the speciality of their choice as a talent, e) Non-academic subjects are costly because they are not available in public schools as given by private schools. f) The lack of differentiation in teaching methods in schools makes talented students forced to underachieve in their academic subjects. g) Curriculum provision: the curriculum taught in schools is inadequate to fulfil the gifted and talented students and the differentiation of the curriculum is unable to accommodate all types of talents, and there is no differentiated or specialised qualified curriculum for talented and gifted students. h) Training of skills: Teachers at schools teach subject content only without paying attention to skills. They do not even know how to train skills. There is no attention paid to develop the students and their teachers in schools. g) University specialities: lack of higher education institutes and academics founded for teaching music and other arts along with cultural subjects is a challenge for those who are talented and who have no option to select any speciality subjects in the university. k) The mismatch between university speciality and jobs, on the other hand, creates founded for teaching music and other arts along with cultural subjects in another challenge for those who graduate without guidance on how or where to find a job. l) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, yet, it is difficult to assess their efficiency or how successful they were. However, this is an oppo	and not encouraging according to all participants from all categories.	Experts, teachers, managers, and gifted and talented.
managers. d) Talented and gifted students can study one speciality in the university, which prevents them from considering the speciality of their choice as a talent, e) Non-academic subjects are costly because they are not available in public schools as given by private schools. f) The lack of differentiation in teaching methods in schools makes talented students forced to underachieve in their academic subjects. g) Curriculum provision: the curriculum taught in schools is inadequate to fulfil the gifted and talented student's potential. There is no curriculum differentiation method adopted, and the differentiation of the curriculum is for the academic giftedness achievements only. The curriculum is unable to accommodate all types of talents, and there is no differentiated or specialised qualified curriculum for talented and gifted students. h) Training of skills: Teachers at schools teach subject content only without paying attention to skills. They do not even know how to train skills. There is no attention paid to develop the students' abilities; students do not understand innovation and innovative products. Teachers' skills do not undergo training, and they are mostly not qualified to discover or identify talented students in their classes. f) University specialities: lack of higher education institutes and academics founded for teaching music and other arts along with cultural subjects is a challenge for those who are talented and who have no option to select any speciality subjects in the university. k) The mismatch between university speciality and jobs, on the other hand, creates another challenge for those who graduate without guidance on how or where to find a job. l) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, this is an opportunity to apply for an assessment program to resume a culture of	subjects only according to all participants from all categories	managers, and gifted and
prevents them from considering the speciality of their choice as a talent, e) Non-academic subjects are costly because they are not available in public schools as given by private schools. f) The lack of differentiation in teaching methods in schools makes talented students forced to underachieve in their academic subjects. g) Curriculum provision: the curriculum taught in schools is inadequate to fulfil the gifted and talented student's potential. There is no curriculum differentiation method adopted, and the differentiation of the curriculum is for the academic giftedness achievements only. The curriculum is unable to accommodate all types of talents, and there is no differentiated or specialised qualified curriculum for talented and gifted students. h) Training of skills: Teachers at schools teach subject content only without paying attention to skills. They do not even know how to train skills. There is no attention paid to develop the students' abilities; students do not understand innovative products. Teachers' skills do not undergo training, and they are mostly not qualified to discover or identify talented students in their classes. l) Human capital: the participants raised the lack of specialist professionals to identify or to provide counselling, coaching, and training for gifted and talented students and their teachers in schools. j) University specialities: lack of higher education institutes and academics founded for teaching music and other arts along with cultural subjects is a challenge for those who are talented and who have no option to select any speciality subjects in the university. k) The mismatch between university speciality and jobs, on the other hand, creates another challenge for those who graduate without guidance on how or where to find a job. l) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, ye	teachers to identify gifted or talented learners, according to experts, teachers, and	•
schools as given by private schools. Experts, teachers, managers, and gifted and talented. Experts, teachers, managers, and gifted and talented. Experts, teachers, and managers. g) Curriculum provision: the curriculum taught in schools is inadequate to fulfil the gifted and talented student's potential. There is no curriculum differentiation method adopted, and the differentiation of the curriculum is unable to accommodate all types of talents, and there is no differentiated or specialised qualified curriculum for talented and gifted students. h) Training of skills: Teachers at schools teach subject content only without paying attention to skills. They do not even know how to train skills. There is no attention paid to develop the students' abilities; students do not understand innovation and innovative products. Teachers' skills do not undergo training, and they are mostly not qualified to discover or identify talented students in their classes. I) Human capital: the participants raised the lack of specialist professionals to identify or to provide counselling, coaching, and training for gifted and talented managers. Parents, experts, teachers, and managers. Parents, experts, teachers, and managers. Parents, experts, teachers, and managers. Parents, experts, teachers, and managers. Experts, teachers, and managers. Experts, teachers, and managers. Parents, experts, teachers, and managers. Experts, teachers, and managers. Experts, teachers, and managers.		managers, and gifted and
students forced to underachieve in their academic subjects. g) Curriculum provision: the curriculum taught in schools is inadequate to fulfil the gifted and talented student's potential. There is no curriculum differentiation method adopted, and the differentiation of the curriculum is for the academic giftedness achievements only. The curriculum is unable to accommodate all types of talents, and there is no differentiated or specialised qualified curriculum for talented and gifted students. h) Training of skills: Teachers at schools teach subject content only without paying attention to skills. They do not even know how to train skills. There is no attention paid to develop the students' abilities; students do not understand innovation and innovative products. Teachers' skills do not undergo training, and they are mostly not qualified to discover or identify talented students in their classes. 1) Human capital: the participants raised the lack of specialist professionals to identify or to provide counselling, coaching, and training for gifted and talented students and their teachers in schools. j) University specialities: lack of higher education institutes and academics founded for teaching music and other arts along with cultural subjects is a challenge for those who are talented and who have no option to select any speciality subjects in the university. k) The mismatch between university speciality and jobs, on the other hand, creates another challenge for those who graduate without guidance on how or where to find a job. 1) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, yet, it is difficult to assess their efficiency or how successful they were. However, this is an opportunity to apply for an assessment program to resume a culture of		Experts, teachers, managers,
the gifted and talented student's potential. There is no curriculum differentiation method adopted, and the differentiation of the curriculum is for the academic giftedness achievements only. The curriculum is unable to accommodate all types of talents, and there is no differentiated or specialised qualified curriculum for talented and gifted students. h) Training of skills: Teachers at schools teach subject content only without paying attention to skills. They do not even know how to train skills. There is no attention paid to develop the students' abilities; students do not understand innovation and innovative products. Teachers' skills do not undergo training, and they are mostly not qualified to discover or identify talented students in their classes. I) Human capital: the participants raised the lack of specialist professionals to identify or to provide counselling, coaching, and training for gifted and talented students and their teachers in schools. j) University specialities: lack of higher education institutes and academics founded for teaching music and other arts along with cultural subjects is a challenge for those who are talented and who have no option to select any speciality subjects in the university. k) The mismatch between university speciality and jobs, on the other hand, creates another challenge for those who graduate without guidance on how or where to find a job. l) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, yet, it is difficult to assess their efficiency or how successful they were. However, this is an opportunity to apply for an assessment program to resume a culture of		
attention to skills. They do not even know how to train skills. There is no attention paid to develop the students' abilities; students do not understand innovation and innovative products. Teachers' skills do not undergo training, and they are mostly not qualified to discover or identify talented students in their classes. I) Human capital: the participants raised the lack of specialist professionals to identify or to provide counselling, coaching, and training for gifted and talented students and their teachers in schools. j) University specialities: lack of higher education institutes and academics founded for teaching music and other arts along with cultural subjects is a challenge for those who are talented and who have no option to select any speciality subjects in the university. k) The mismatch between university speciality and jobs, on the other hand, creates another challenge for those who graduate without guidance on how or where to find a job. l) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, yet, it is difficult to assess their efficiency or how successful they were. However, this is an opportunity to apply for an assessment program to resume a culture of	the gifted and talented student's potential. There is no curriculum differentiation method adopted, and the differentiation of the curriculum is for the academic giftedness achievements only. The curriculum is unable to accommodate all types of talents, and there is no differentiated or specialised qualified curriculum for	_
identify or to provide counselling, coaching, and training for gifted and talented students and their teachers in schools. j) University specialities: lack of higher education institutes and academics founded for teaching music and other arts along with cultural subjects is a challenge for those who are talented and who have no option to select any speciality subjects in the university. k) The mismatch between university speciality and jobs, on the other hand, creates another challenge for those who graduate without guidance on how or where to find a job. l) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, yet, it is difficult to assess their efficiency or how successful they were. However, this is an opportunity to apply for an assessment program to resume a culture of	attention to skills. They do not even know how to train skills. There is no attention paid to develop the students' abilities; students do not understand innovation and innovative products. Teachers' skills do not undergo training, and they are mostly	
founded for teaching music and other arts along with cultural subjects is a challenge for those who are talented and who have no option to select any speciality subjects in the university. k) The mismatch between university speciality and jobs, on the other hand, creates another challenge for those who graduate without guidance on how or where to find a job. l) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, yet, it is difficult to assess their efficiency or how successful they were. However, this is an opportunity to apply for an assessment program to resume a culture of	identify or to provide counselling, coaching, and training for gifted and talented	
another challenge for those who graduate without guidance on how or where to find a job. 1) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, yet, it is difficult to assess their efficiency or how successful they were. However, this is an opportunity to apply for an assessment program to resume a culture of	founded for teaching music and other arts along with cultural subjects is a challenge for those who are talented and who have no option to select any	-
challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, yet, it is difficult to assess their efficiency or how successful they were. However, this is an opportunity to apply for an assessment program to resume a culture of	another challenge for those who graduate without guidance on how or where to	•
	l) A significant speciality is available in one university only, this causes other challenges such as the availability of an opportunity, and who can obtain it. m) Most of the implemented identification and development programs are recent, yet, it is difficult to assess their efficiency or how successful they were. However, this is an opportunity to apply for an assessment program to resume a culture of	and managers. Experts, teachers, and

In these sections, several challenges represented as obtained from the results with alignment to the themes derived from the results of the discussions.

(a)Challenges in knowledge and awareness

This major challenge occurs in the misunderstandings and beliefs of the community and their attitudes towards gifted and talented individuals. Challenges can be at home, in schools, and the classroom setting. According to (NAGC)'s "list of the most prevalent myths in gifted education" (nagc.org), many of these turned to be solid myths, believed, and accepted for a long time. These myths are including (1) the belief that "gifted and talented children can learn on their own" (nagc.org) because their curiosity will lead them to learn and know what they need to know without any interference. This category of individuals indeed has their innate ability; however, this ability and these children require guidance to know "what it means to be gifted or talented and how to nurture their giftedness and talent to benefit out of them" (nagc.org)

(2) People assume "the happiness of the gifted and talented students and that they always enjoy being exceptional" (nagc.org), however, the highly able persons they are not satisfied all the time, and they may feel lonely or isolated most of the time. Their behaviour requires schools and parents to seek psychological advice to help their children know how to deal with their giftedness or talents and to live happily like the others. (nagc.org, 2019)

Teachers and parents assume that "those students find it interesting to know all the answers in the classroom" (nagc.org). The thing that some people do not understand that while talented and gifted students have an incredible ability and curiosity, but they will not engage in any material they do not find interesting. The worst challenge for them is to feel bored, as this feeling will hinder their interest, and they will not make any effort to learn or to involve themselves in the class discussions. The most challenges the gifted and talented students' face that their parents think if they are gifted or talented, they do not require development because their teachers can recognize and support any highly able students in their classroom. This issue requires the attention of the

teacher or the parents to help this type of extraordinary students find their interest and passion for topics they like. The fact about how most of the teachers are not qualified or trained on how to identify or recognise the students with exceptional abilities would increase the gap and minimise their abilities. Moreover, gifted and talented students are more comfortable to teach, and they learn faster than others do according to the myth. However, and despite being highly able, teachers need to change their methods of teaching and present the subject in distinct ways. Therefore, to teach these students, teachers must exert more effort to keep them interested and willing to learn. The high expectations of teachers or parents or even the community would help the development of these students and urge them to achieve better. However, very high expectations might oppress and pressure them. These highly able students do not like to feel pressured, though people around them should understand this. They as well assume that the giftedness or talents will make the gifted and talented students achieve without effort, however, highly able children will likely be at the top of a class and get high grades unless they were interested in the subject.

A talented girl said in the interview said "When I told my mother that my school identified me as talented in arts, she was not happy, and she asked me to look for another useful talent in me.

Frankly, this upset me because I love drawing, and I can't find my interest or passion in another field to make my mother happier."

A mother of a talented son in entrepreneurship said: "My husband felt ashamed from our son when his school called him for a meeting and told him that our son creates original items and sells them in school. His father stopped him from doing his own business as he called it and told him that what he is doing appears like he is poor and in need of money. He told him we do not need the money, and people will think we do if you sell your creations."

The lack of awareness or knowledge about giftedness and talents, the need of freedom required by these individuals to feel they can create and produce is one of the most considerable challenges that hinder these students from exhibiting their abilities and interpreting them into achievements.

A parent said that he is not worried about searching for any development programs to support his son's talent because he is sure that his son's talent will find its way and will lead him to decide and choose his path without the help or interference of any tools or outsiders."

A teacher of gifted and talented said that she had no clue about any other institutions' programs or development strategies for gifted or talented. She said that she heard about these programs during this interview only and for the first time.

(b)Challenges in definition and identification

According to (NAGC)'s" list of the most prevalent myths in gifted education, all children labelled as gifted have strengths, and they display curiosity, which is one of the giftedness and talents traits" (nagc.org). This mistake leads the children to underachieve because people assume by this definition that these highly able would succeed and achieve better marks than his /her peers in the classroom achieve.

A teacher said: "it is easy nowadays to label any hardworking student as talented, while I think this will change the concept of talents and will make anyone who works in any work is called talented."

An expert who worked as a gifted and talented program developer said:" We are not only in a debate regarding the differences between gifted and talented, but we also debate the usefulness of many of the identification tools. Some of these are outdated, and if it comes to us, there is a

need for different types of definition and identification tools customised to suit the environment and many other aspects that would prevent losing talented individuals or identifying a non-talented as a talented.

The UAE's vision, strategies, and national agendas depend on a variety of values such as leadership, innovation, and creativity. Therefore, creativity and leadership should emphasise the selection of abilities in the definition of a gifted and talented individual in the UAE, and this would help the educators and the related stakeholders provide them with the appropriate services accordingly.

Development of the best identification programs depends on how the highly able students defined. The definition should include specific abilities and skills. Developers of programs can determine the type of identification if they have a clear definition for them. The definition helps to nominate and select any gifted and talented student and their enrollment in specific development programs. An appropriate definition helps the educators as well to decide the type of education, the method of identification, and the development programs that will support their talents' progression.

(c) Challenges in development and process

Challenges in the development of the highly able students rise mainly from the absence of specialised, differentiated and gifted education. The nature of general education does not enhance these students' development. It does not challenge their specific abilities as well. Gifted and talented students usually find regular education boring and uninteresting, and they fail to get along with it. Furthermore, the style of regular learning and teaching relies heavily on memorisation and lecturing. This system results in increasing dropouts and numbers of underachievers, and it narrows the participation of the highly able students and hinders their skills from progression.

According to an expert:

"The argument here is about the domination of the traditional teaching style in all theoretical and practical subjects at schools. Many children dropped out of schools, and many of them may be gifted or talented."

According to another expert, "The bestseller development programs are the enrichment programs that occur after school or trips and visits that involve all students and do not allow any gifted and talented student to feel he or she is a unique person."

A father of a teenager gifted said, "I do not encourage my son to waste this valuable time to attend any enrichment programs provided after school. Drawing and arts are the girls' interests. He can spend quality time studying physics or math subjects. He is talented in math, and he will get high grades if he focuses on his subjects. I wonder why schools do not increase science subjects in boys' education. If the school introduces robotics programs, I may consider this program only and encourage him to attend this one specifically."

A psychologist indicated, "In parents' psychology and mindsets, any program excluded from the primary curriculum is useless and applied for fun and entertainment only. They view arts as one of the feminine interests, and it will take long years to convince them that cooking, for example, is the place for many successful male chefs. It is part of arts and culture, and they are defined as talented accordingly."

5.4 Results from focus groups

This research aimed to study the development system in the educational institutions in the UAE to detect the status of identification and development in schools. As previously discussed in the

chapter (3), the focus group discussions intended as part of this study methodology to obtain data. The researcher varied the methods for the sake of getting more data from different resources. Group discussions were infused using three scenarios in alignment with the themes. They used the scenarios to tell their stories and inform the topic with real data. In this way, the researcher was able to generate discussions among the participants. These participants met for the first time; however, they were familiar with the highly able individuals' issues. They were able to describe the highly able students statue currently and suggest what they think should change in the future. Their discussions were helpful to recommend the design of a better ecosystem eager to support the highly able progress.

The findings and discussions among (64) participants in these focus groups, will be highlighted in the following section, noting that this study will not disclose these participants names or designations to protect their anonymity.

5.5 Findings from Focus Groups

As previously discussed in the chapter(3), setting in the focus group sessions was prepared to guide the discussions. It intended to allow the groups' participants to imagine the life cycle of the highly able students. They were using the scenarios and the stories presented to them in their discussions. The debates aimed to obtain the opinions and answers from the participants that constitute the stakeholders in the lives of gifted and talented individuals. The selection of the participants for the focus groups was similar to the selection of the participants for the interview. Why did this happen? Because the researcher was looking for answers to the same themes that respond to the research questions. Obtaining answers from different participants for the same question supports the reliability of the study and to generate preliminary data. Any real information obtained from lively

discussions and brainstorming would allow the study to have better insight into the issues. The groups' participants included educational experts, teachers, managers, parents of highly able students, highly able students and adults. All of them had different experiences in the field of gifted programs in the UAE, and they are familiar with its educational system and environment. The focus group participants debated different aspects of the critical issues.

Table 36. Participants in Focus Groups

Participant Group	Code	Number of participants
Experts -including psychologists	E	8
Teachers	T	4
Managers	PM	2
Parents	P	12
Young talents	YT	18
Adolescence talents	AT	21

5.6 Analysing focus group discussions

Four main themes emerged from the interview data. These four themes constituted the main topics for discussions in the focus group, as they were the core issues that considered a comprehensive ecosystem supposed to serve and support the highly able students. The four themes including knowledge and awareness, definition and identification, development and processes, and finally challenges, formed the ground from which the scenarios given to each group came, with stories intended to add more insights to the results and findings from the interviews.

Various challenges emerged within the stories mocked in the group discussions in response to the given scenario. The problems in alignment with the themes were: (1) lack of support given to parents who need expert advice to raise their awareness and learn how to identify and nurture their highly able children. (2) The misconception of definitions due to not having a unified definition.

(MOE) In their opinion should impose a unified national definition within a federal national policy to regulate the services and provisions to support the highly able persons in the country. (3) The availability of numerous identification and development programs and tools. According to the expert's focus group, many of these tools lack many enablers that empower them to serve and support the gifted and talented students' differentiated needs. These challenges mainly focused in: Age-specific Tools; Tools that recognise a specific talent domain; Tools that only test one or two skills and the majority of these tools or programs disconnect. The small number of specialists and experts in this field is another challenge. Such a shortage of professionals in different domains of expertise affects the area and reflects negatively on the identification and development processes as well. Furthermore, the available experts do not undergo periodical professional development to update their methods and knowledge in this area. On the other hand, teachers who are in charge of teaching and developing highly able students are teachers who work in special or regular education. They go through a short-term diploma, which is not adequate for a critical process regarding nurturing the highly able students. Moreover, the lack of a mechanism to bring together all experts and specialists is another critical challenge in the process. Gifted and talented specialists need to share knowledge and experience, communicate with parents and teachers, and develop a policy that unifies all concepts. The development and disseminate of a policy would standardise practices and regulations is an essential procedure towards a thorough solution. The establishment of specialised academies in different domains of talents is a must. Conducting researches that explore the type of current identification processes would help to identify the gaps and propose improvements. Experts group thought that the usage of a platform to get expert advice would help students and parents gain the experience they need. They also thought that the professional

development of a gifted and talented teacher must go through systematic education and not just a short diploma or brief courses.

Table 37. Themes analysis

Following analysing the findings from the focus group discussions, below is a summary of results generated from these discussions

Knowledge and awareness	Participants	
Three Groups pointed out that the family & community do not have adequate knowledge about giftedness and talents, and that negatively affects their support for their children.	Parents Gifted and talented students Experts	
Four groups pointed out that parent prefers and encourage their children to study academic programs over artistic talents due to the insufficient understanding of the significance of talents for the UAE national and international profile.	Parents, gifted and talented students Gifted and talented adults Experts	
The society perspective reflected in the discussions in more than two groups. The talented adults, as well as students, agreed the society still finds art talents as a complementary component, and the investment favours the academic talents.	Parents Experts, teachers, and managers.	
The students do not receive adequate guiding on aligning their career options with the needs of the workforce.	Parents, experts, teachers, managers, Gifted and talented students and adults.	
In one of the groups, there was a consent that gender is pertinent in determining the appropriateness	Gifted and talented students and adults.	
Definitions and identification of Giftedness & Talents		
The current talented and gifted provisions and practices are random, not systematic, and not based on an institutional philosophy.	Experts, teachers, managers,	
There is no overall organisation to regulate giftedness and talent definitions and practices.	Experts, teachers, managers,	
The absence of a policy framework that supports and links the current practices and provides guidance and regulations on the processes and procedures of identification.	Experts, teachers, managers,	
The lack of comprehensive gifted and talented identification programs in schools to provide the G&T students with a conducive, stimulating, and challenging environment.	Experts, teachers, managers, Parents	
The current gifted and talented practice is not sustainable.	Experts, teachers, managers,	
Lack of collaboration, and coordination between the organisations, associations who provide gifted and talented programs.	Experts, teachers, managers,	
Development and processes		
Human capital: the participants raised the lack of specialist professionals to identify or to provide counselling, coaching, and training for gifted and talented individuals.	Experts, teachers, and managers.	
Curriculum provision: the taught curriculum is inadequate to fulfil the talented student's potential, the curriculum is unable to accommodate all types of talents, there is no curriculum for talented and gifted students	Experts, teachers, and managers.	
University courses: lack of higher education institutes for teaching music and other arts subjects, challenges in selecting speciality subjects in the university,	Parents, Experts, teachers, managers, and gifted and talented.	

the mismatch between university speciality and jobs, and a significant speciality is available in one university only. Skills' training: teachers teach subject content, there is no attention paid to Experts, teachers, and develop the students' skills, students do not understand innovation and managers. innovative products Challenges a) The support given for the subjects related to arts and cultural talents is very little Parents, and not encouraging according to all participants from all categories. Experts, teachers, managers, and gifted and talented. b) There are no scholarships granted for artistic talents but to the academic subjects Parents, experts, teachers, only according to all participants from all categories managers, and gifted and talented adults. c) Lack of qualified instruments for identification, lack of trained and qualified Experts, teachers, and teachers to identify gifted or talented learners, according to experts, teachers, and managers. managers. d) Talented and gifted students can study one speciality in the university, which Parents, experts, teachers, managers, and gifted and prevents them from considering the speciality of their choice as a talent, talented. e) Non-academic subjects are costly because they are not available in public Parents, schools as given by private schools. Experts, teachers, managers, and gifted and talented. f) The lack of differentiation in teaching methods in schools makes talented Experts, teachers, and students forced to underachieve in their academic subjects. managers. Experts, teachers, and g) Curriculum provision: the curriculum taught in schools is inadequate to fulfil the gifted and talented student's potential. There is no curriculum differentiation managers. method adopted, and the differentiation of the curriculum is for the academic giftedness achievements only. The curriculum is unable to accommodate all types of talents, and there is no differentiated or specialised qualified curriculum for talented and gifted students. h) Training of skills: Teachers at schools teach subject content only without paying Experts, teachers, and attention to skills. There is no attention paid to develop the students' abilities. managers. Teachers' skills also do not undergo training, and they are mostly not qualified to discover or identify talented students in their classes. i) Human capital: the participants raised the lack of specialist professionals to Parents, experts, teachers, identify or to provide counselling, coaching, and training for gifted and talented and managers. students and their teachers in schools. j) University specialities: lack of higher education institutes and academics Parents, experts, teachers, founded for teaching music and other arts along with cultural subjects is a and managers. challenge for those who are talented and who have no option to select any speciality subjects in the university. k) The mismatch between university speciality and jobs, on the other hand, creates Experts, teachers, and another challenge for those who graduate without guidance on how or where to managers. find a job. 1) A significant speciality is available in one university only, this causes other Parents, experts, teachers, challenges such as the availability of an opportunity, and who can obtain it. and managers. m) Most of the implemented identification and development programs are recent, Experts, teachers, and yet, it is difficult to assess their efficiency or how successful they were. However, managers. this is an opportunity to apply for an assessment program to resume a culture of programs' ongoing evaluation in the future. Financial support: the participants pointed out challenges with financial backing Parents,

Experts, teachers, managers,

and gifted and talented.

such as little support for the non-academic subjects, scholarships, instruments,

teachers, students can study one speciality in the university, and non-academic

items are costly, lack of differentiation in teaching in schools.

5.7 Evidence from Focus Groups discussions

Focus groups consisting of different categories of stakeholders brought together to gain more information related to this research topic. They discussed the four themes presented in a manner of scenarios. They told their stories in different ways. They discussed the challenges and suggested solutions to overcome these challenges. The groups' discussions generated information added to the data obtained from the interviews. The issues of awareness, definitions, identification, and development programs in the UAE were the main themes discussed.

Following a summary of the evidence derived from the discussions conducted within the focus groups

5.7.1 The Parents Group

Parents' focus group consisted of parents who were accompanying their talented children. They all expressed their worries about the future of their children, and they look at the challenges in this period as an obstacle that should not be there. The following paragraph will look at these challenges and try to define them as discussed during the session. The scenario in this group was suggesting that parents have to choose the future track for their talented child who is interested in arts. They have to decide which educational path they prefer or impose on their children. Arts or Science were the two defined topics, and parents were required to imagine guiding their children within one or both tracks. They had to write and draw a plan that will support the child in his way to the future. They had to write down the challenges they expect their child would face while moving on within each track.

Evidence from the parents' group

Parents' opinions split into two. Fathers of the gifted males rejected the idea of their children studying arts and drew a road map for their children that leads them towards STEM education.

Mothers described arts as a not important subject because it is an enrichment program that is not

ranked like other subjects and does not require academic education. They thought that STEM is the only way towards the future.

Parent1: (Mother): There is no time to participate in all events, and our children are distracted by many different competitions, exhibitions, and any other function in the country.

Parent 2(Father): The school is not providing any programs in Robotics or science to sharpen their skills and prepare them to be future scientists as I dream."

Parent 3: (Mother): I am worried about my child's health. He does not express enough to know how he feels. There is no counsellor in his school to talk to him. It is the school's responsibility to follow up on the student's health. The other thing that worries me is the differences between parents and children's perspectives. They do not think like us.

Parent 4: (Father): My son suffers from being multi-talented. He is a genius in math, physics, biology, and robotics. He faces difficulty in choosing a university speciality because he does not know which skill will help him achieve.

Parent 5: (Father): I have three talented children. One of my daughters is a musician but cannot find higher institutes of music to develop her abilities in music. The other child is passionate about photography, but again, there are no specialised academies for them. The offered programs at the different institutions here are whether short or not qualified or costly.

Parent 6: (Mother): There is no sustainability in gifted programs. There are no talent centres. My son feels distracted now because he cannot find a more extended program that supports his talent in Cinema.

Parent 7: (Father): My son is finding difficulty in certain conditions of admission to universities.

There are no mentors available at school to guide them and direct them to the right speciality.

5.7.2 The Experts' group

Experts focus group consisted of experts, teachers, and managers. Some teachers were graduates from the gifted teachers' diploma. They all work in the UAE in different educational institutions, and they are acquainted with the situation of the highly able students' in the country. The scenario given to this group was about the identification and development programs in the country.

"We have a massive number of specialised tools prepared to serve those who are gifted and talented in the UAE, yet most of them are still new to evaluate, and there are still various gaps regarding:

- The inability of most of these existing tools to detect the types of gifts and talents that are difficult to detect (unique gifts and talents)
- There is an ample supply (the labour market provides many fields of business), and university graduates are in large numbers. Studies have shown that there is no consistency between supply and demand.
- There is a need for development programs to bridge or reduce the gap between supply and demand.

Evidence from the experts' group

Expert1: Efforts in the field of giftedness and talents requires a council or an entity to coordinate all these institutions' efforts. There is a need for a national policy that mandates assessments and

evaluations and criteria for these tools and programs, and in conclusion, if there are no provisions, there are no excellent services as well.

Teacher1: There is no sustainability in the discovery and follow-up of talents. The shortness in the program's periods and their durability distracted the gifted and talented students and their parents. Such a process was challenging for them. It is challenging to follow up with the identified students because some of them leave the school. There are no improved e-portfolios for them that can always support follow-ups with them and in any other school.

Manager: The presence of a gifted specialist in each school as a specialised teacher would hugely support gifted and talented students. They should be able to mentor not only gifted and talented students but also other teachers, leaders in school, and parents according to clear policy and methodology at the state level. These specialists will address any challenges through their experiences in the field of education and will support the implementation of the 21st-century strategies in discovering the abilities and skills of students in different domains of gifts and talents.

5.7.3 The Gifted and Talented focus group

In this group, a selection of highly able young males and females' students (artistic and scientific) came together to discuss the issues reflected in the scenarios. They were from different age groups, interests, genders and had various experiences in different gifted and talented development programs.

The scenario: Draw your life journey from today until you graduate from the university following the below script:

1-Today you knew that you are a gifted /talented student. From here, define the base stations in your learning journey. What do you need? Who will support you at every station? How do you plan for every stage?

2-The next crucial stage: Today, you graduated from college and started looking for a job. What kind of profession you think will suit your gifts/talents?

3-Can you draw the work you want to do in a circle and try to suggest the skills needed to master this work?

4-Now, imagine that you found the work or project you are passionate about: What is the product that you produce?

Evidence from the Gifted and Talented focus group

Highly able students thought that parents put more pressure and high expectations on them.

Others said that their parents want to impose their choices of study, interest and career on them.

The society's mindset is not adapted accepting the domain of Arts and Culture as careers.

Universities as well do not provide scholarships in the domains of arts and culture nor any organised education. The highly able students in this group seemed frustrated by the oppression of the educational system and commented on the lack of teachers' awareness and support.

Talent 1(male): I was born with a problem with hearing. I could not cope with the classroom work or with my teachers, who were unable to diagnose my silence due to my illness. I was talented, but I needed support. My teachers could not identify my talents, but my mother tried to help me. She moved me to different schools and exerted efforts to diagnose my hearing problems.

She exposed me to different hobbies and crafts until I excelled sculpturing. Today I am an artist.

I see myself creating more sculptures and investing in them.

Talent 2(male): My parents are not aware of the importance of talents, so I do not know how they can help me to move on the path I choose. Besides, it is difficult to convince them that I want to do my major in music, which is my talent. They cannot see my future in anything but in my academic achievement. It is oppressing.

Talent 3(female): I love music; I am encouraged by my father, who rides me to Fujairah to attend music classes at Al Fujairah Academy. I want to become a pianist; I see myself sitting in front of the piano and playing music in the opera, but there is no way to practice what I like because I cannot practice in the Academy anymore. I will graduate from school and because there are no university programs to do so. I will travel to study abroad and satisfy my passion for music.

Talent 4(male): There is no support for me at school neither at home. I am failing the lessons in school because my parents want me to get a certificate in what I am not interested in learning.

They think I want a future job in an office, which I dislike. I cannot explain to my teacher why I underachieve in math while I get the highest marks in PE. I want to become an athlete, and I do not achieve well in any academic tests. I do not know why I must study what I do not like! I want to be an athlete and I need to practice and not to sit in a classroom.

Talen 5(male): Sometimes, I feel like it is my fault for being talented in arts, sports, and math. I do not know how to solve the problem of what to study and when or how to manage my talents.

No one tells me what to do, and I cannot find the university that would help me combine two majors or even plan better objectives. It is challenging to choose any of them.

5.7.4 Documentation

Evidence from Documentation

(1)Knowledge and Awareness

In support of the findings obtained from both methods "interviews and focus groups", an analysis was conducted on the related documents to investigate the development programs implemented by the educational institutions in the UAE. Another search was to find out if these institutions developed a law or a policy that support the highly able students' and nurture their needs. The investigation aimed to explore the documents issued and related to the improvements in education, including the development of these highly able individuals in the country. The reviewed materials were mostly resolutions, ministerial decrees, and other official publications that give more insights about the five institutions' activities and achievements in the field of education, in addition to other reports like "The United Arab Emirates National Commission for Education, Culture, and Science, *UNESCO* reports". (moe.gov.ae)

The findings from the document and their analysis supported the four themes generated from the "interviews and the focus groups" such as knowledge and awareness, definition and identification; development process; and challenges. This section follows a similar manner like the manner observed while discussing the results of the interviews and the focus group. The findings were as follows:

Starting with the goals of the educational institutions in the UAE, they emphasise on the student's personality and ability to grow in a creative environment. They also highlight society's needs for equal education and opportunities for more creativity and innovation.

The UAE constitution, permanently adopted in 1971 with amendments through 2004, "is the primary source of laws and legislations and the fundamental basis that governs the political, economic and social affairs in the UAE. It allocates the legislative, executive and international jurisdiction between the federation and the emirates and provides essential directions regarding education in the country". It states in "(Article (17) of the constitution that "education is a primary means of social development". "It is made compulsory and free, and the educational plans will be prescribed by law for the circulation of education all over the country." (Emirates, U.A, 2016)

The UAE depends mainly on education. It applies different services in the educational system to infuse its development and advancements(Warner, R.S. & Burton, 2017). The progressive education system in the UAE is adopted to fulfil the market needs for human capital development. The global agenda of the UAE urges it to develop a sustainable model of education. The majority of educational institutions in the UAE are empowering the type of education that supports the development of both students and teachers. They aim to make the students more aware and able in social life, better educated and ready to function in a knowledge-based economy. Human capital development is what the education stakeholders are looking at to attain the UAE National Agenda aims for education.

In the (Assessment project of 111 UAE UNESCO Associated Schools to identify the efficient schools), the UAE Commission reported the commitment of the UAE to achieve its goals and plans, especially those concerned with UAE schools and related to the educational system. The Commission referred by that to the UNESCO's associated schools-specific objectives in accordance with its medium-term strategy (2014-2021): Education integration for international citizenship, education for sustainable development in learning and education, experimentation of the creative aspects to achieve integration, strengthening the exchange of information, experiences

and leading practices in the UAE network schools. They also must take part in achieving the strategic goal, which is empowering all learners to become global and creative citizens'. This empowerment infuses the development of in-depth knowledge, strong values, and more specific skills. (*moe.gov.ae*).

Moreover, the *World Data on Education Report (VII Ed.2010/11)* included (*MOE*) Education Strategy (2010-2020) that aims to deliver a student-centred model focused on improving student's outcomes, school life, and equality. The overall outcomes of the strategy, according to the report, were to make students ready for a lifetime career and teachers' excel in supporting them (*De Education*, 2010)

Therefore, and according to the "World Data on Education Report", the UAE new organisational structure identifies and determines the educational tasks employed to the (MOE). These tasks include implementing advanced education for all citizens of the UAE, setting educational plans, advanced curricula, advanced examination systems, and adult ongoing lifelong education programs. Moreover, it shall establish schools and institutions and monitor their performance. The new educational system as well defined the role of the (MOE) to develop and implement policies, standards, and plans and all other supporting services." (De Education, 2010).

The UAE government, therefore, has longtime before entrusted the Ministry of Education (MOE) to provide the best education to its youth. The Ministry of education, according to *Federal Law No 1-M7*, bears the responsibilities of supervising and improving the educational affairs in the country (De Education, 2010). The Cabinet Decree No.1 of 5 January 1987 defines the objectives of the (MOE). It focuses on achieving these objectives and structuring the formation of a supreme committee to oversee the Education Policy and identifies its tasks. "*The Cabinet Decree No.7 of 15 May 1991*" deals with the establishment of a Ministerial Advisory Committee to study all

matters dealing with education policies. However, and in the light of the existing policies to oversee the education of different categories in the UAE, policies related to gifted education and development programs are not within their tasks. Such a policy is in need to regulate and help the implementation of high-quality development programs, and this policy is supposed to be a part of the (MOE) future policies.

On the other hand, the UAE government has launched several educational institutions to oversee the development of the educational system in the other Emirates such as (ADEK), which is authorised to deliver education in the emirate of Abu Dhabi and policy development since 2005.

The establishment of the (KHDA) in 2006 was to develop the sectors concerned about "knowledge and human resources" in Dubai. The role of (KHDA) is to assure that these sectors apply quality in their missions and to support and improve education and human development. Another objective is to develop a comprehensive strategy for "knowledge and human resources development" (khda.gov.ae). The KHDA accordingly established the National Institute for Vocational Education (NIVE.ae) in 2006. It also established the "University Quality Assurance International Board" (UQAIB) in 2006 to assess and license universities operating in the education free zone in Dubai (khda.gov.ae) it is responsible for private school inspection, too. However, and while individual educational policies and strategies do exist in several institutions in Dubai, a national policy related to highly able students is not among them.

The UAE government exerts efforts to facilitate collaboration among stakeholders. It aims to infuse the development of highly able students' skills. It operates to ensure that institutions services are meeting the expectations of gifted and talented students. The role of each educational institution will not appear enough unless a national policy regulates its efforts. The unavailability

of a policy asserts the concerns revealed by different participants in the interviews and focused group regarding gaps in the programs.

Finally, there is a need for a collaborative strategy that enhances communication among all of the educational institutions. This strategy would help them fulfil their roles collaboratively and enhance the talented and gifted services in the UAE. It would organise each authority task to support a systemized harmonised ecosystem capable of the development and progression of these highly able students in the UAE.

(2) Definition & Identification

Definition of talents (MOE)

According to the "General Rules for the Provision of Special Education Programs and Services in (Public & Private Schools)" (moe.gov.ae), the definition of gifted and talented is:

"A gifted and talented is the one with outstanding ability or aptitude in one or more areas of intelligence, or creativity, or academic achievement or special talents and abilities such as oratory, poetry, drawing or handicrafts, sports, or drama, or leadership capacity and others. More often, the performance of outstanding, talented students in the above areas is distinct from his peers who are in the same age group." (moe.gov.ae)

However, this definition does not differentiate the giftedness and talents. It does not specify any other traits or abilities that differentiate the gifted and the talented; it instead points at specific areas or unique talents and abilities in specific types and domains of gifts and talents.

Definition of talents (MCKD)

It appeared that the "Ministry of Culture and Knowledge Development" did not generate a definition for the highly able persons in culture and arts. However, the Ministry is in the process

of developing a policy for talented individuals, and this suggests that it may include a definition in this policy. In general talents at (MCKD) means those who produce, publish, perform, or win in the literature and arts or cultural contests or other different domains of talents.

Definition of talents (ADEK)

Talents are the students with outstanding abilities and skills that lead to outstanding high performance and their expected accomplishments, and potential are at the forefront of most of their peers in one or more of the following areas"(Adek.abudhabi.ae):

- Intellectual ability,
- Specific subject competence such as mathematics and science, meeting maturity, leadership, mechanical
- Theatrical and theatrical arts, such as literature, theatre and fiction, psychological and motor ability, such as dancing or playing sports

Their needs require specific consideration within the mainstream educational programs. (Adek.abudhabi.ae)

Definition of talents (KHDA)

The definitions of gifted and talented students at (KHDA) reflects the "Differentiation Model of Giftedness and Talent" and align with international best practices according to (KHDA) inspection report".(khda.gov.ae)

"Talented refers to "students able to transform their "giftedness" into exceptional performance, and they are creative in one or more ways. They show significantly higher performance than their peers do in the same age group. Students are likely to show unique talents and abilities in aspects such as poetry, painting, sports, and theatre" (moe, gov, ae).

"Gifted refers to a student who is inexperienced and spontaneously- expresses exceptional natural ability in one or more domains of social skills such as intellectual, creative, social, and physical skills. Gifted students, while extraordinary potential will be present, may underachieve. They are students who have the potential to study or have the potential to achieve much higher rates than their peers in one or more areas of their academic study. These students show distinct performance levels from their peers". (moe.gov.ae)

Definition of talents (HA)

"A gifted student is the student who has exceptional preparation and potential or has markedly superior performance than his / her peers in general mental ability and or academic achievement, creative thinking, and or leadership that the community appreciates.

This definition limits the giftedness to special preparations and describes the gifted students as the one who has (exceptional) and (markedly superior performance).

Identification of talents (MOE)

Identification of gifted and talented in (MOE) depends on six methods or stages.

The stages of identification are:

- 1) Differentiation-A differentiated method to implement the curriculum without special education provisions.
- 2) Pace-Highly able to learn at a faster pace and teachers required to give them extra challenges.

- 3) Assessment-Ongoing assessment to match the curriculum to student's needs.
- 4) Grouping-Flexible grouping of highly able students in the classroom.
- 5) Level of work-The curriculum needs to include complex abstract ideas to challenge their abilities.
- 6) Enrichment-Extended enrichment with challenging opportunities and programs within and beyond classrooms.

Identification of talents (MCKD)

Competitions, camps and contests are the followed procedures in the (MCKD) to identify talented and gifted individuals. The presence of (9) cultural community centres affiliated to the (MCKD) and distributed around the UAE makes them a primary ground for identification(*mckd.gov.ae*).

Identification of talents (ADEK)

The process of identification at (ADEK) starts at grade (4) -Cycle (1) -grade (6) -Cycle 2 -and Grade (10) in Cycle 3(Adek.abudhabi.ae).

This identification process depends on five criteria's that the student must meet (3) of them at least:

- (1) Achievement: This means having a (97) per cent in a standardised test such as (EMSA -PIPS) in a specific area. Alternatively, having a (90) % in total in all standardised tests areas like (EMSA-PIPS). Alternatively, as one of the top 10 in reading and understanding in English or Arabic languages (Adek.abdudhabi.ae).
- (2) Characteristics Checklist): It contains (15) characteristics, and a student must have more than half of them. The teacher must be good in using these checklists, should be familiar with the student, and knows him/her by a direct relation in the classroom. Parents and peers.

- (3) Screening: This assessment uses an approved test like the IQ test. The score of the student must be (120-above).
- (4) Final Grades: The gifted student is very high-performance evidence in the last two academic years will be on his/her final report card and that he/she has been among the top (10 %).
- (5) Product Assessment Rating: The student's product must obtain (very good) or (outstanding) in product rating form if evaluated by a specialist.

Identification of talents (KHDA)

The identification of gifted and talented, according to (KHDA) inspection report "is not an isolated activity. Identification is the first crucial stage of a continuum of provisions that should provide appropriate and personalised learning programs to enable all gifted students to become talented" (khda.gov.ae).

The critical aspects considered in (KHDA) when personalising education programs for highly able students are:

- Differentiation: A differentiated curriculum for all students with different learning methods and ability levels.
- Pace: Learning at a faster pace.
- Assessment: An ongoing diagnostic tool for matching curriculum delivery to student's needs.
- Groupings: Students with extraordinary gifts and talents benefit from flexible grouping within the classroom.
- Level of work: The level of understanding of the highly able students is usually above usual age-related expectations.

Enrichment: The majority of highly able learners are in mainstream classes while they
need more opportunities for advancement, extension, and acceleration within and beyond
the classroom.

Identification of gifted (HA)

Identification at (HA) is a critical stage that supports the stage of building students' development programs. Identification of gifted students is to provide them with appropriate services and enough educational programs in every domain of giftedness (ha.ae). This process, according to the foundations' documents, contributes to control the system of how to enrol the gifted children in the programs designed for their development. The supplement programs aim to attract the most worthy students to provide them with developmental services. They determine the type and level of gifts to define the needs of the gifted students and guide them to enrol in the appropriate programs for their area of competence. At (HA) there are different stages of gifted student's identification process such as:

- (1) Screening: To accommodate the most significant possible number of students and engage them in the identification procedures. It includes:
- (a) Nominations of students "by teachers, parents, and peers or self –nomination".
- (b) Screening methods "checklists, test results, portfolios, teacher reports".
- (c) The narrowing: To identify gifted students based on specific criteria by using qualitative and quantitative tools.
- (d) Inclusion: To make decisions concerning assigning selected students in the second stage of educational programs in line with their abilities, preferences, and interests. This stage works via

personal interviews and lists of inclinations and interests. Identification tools used are IQ, creativity, and academic achievement tests(ha.ae).

The identification standards and tools split into two types: "Quantitative tools that measure mental abilities are" intelligence tests, the gifted identification kit, academic achievement tests, and innovative Problem solutions test". "Qualitative tools include a portfolio, a notecard, behavioural statements, personal interviews, lists of professional interests, and inclinations".

The gifted identification kit is to discover and identify gifted students through the provision of adequate identification tools. (HA) Developed specific tools to commensurate with the UAE environment. They develop measuring tools for each stage to measure that stage.

(3) Development & Processes

Development of Talents in the UAE

In UAE Federal Law on child rights (*Wadeema's Law*) no.3 of 2016 issued on 8/3/2016, articles (6), (25), (30), (31) and (32), stressed on the child's rights for development and knowledge.

Article (6) included the liability of the concerned entities in the UAE to implement policies and programs developed by them to work in all areas related to the child. The cultural rights of the child in this Law's (Chapter 6-Article (25) incorporated that the capable concerned authorities in the UAE should provide the child with his/her rights for knowledge, innovation and creativity means to participate in the recreational, cultural, artistic and scientific programs that are adequate to the child's age(khda.gov.ae). "Article (30) was about the necessity of forming specialised councils, associations, clubs, and centres to develop children's cultural, artistic, scientific, and physical talents". "Article (31) assured that every child has the right to be educated, and the country should provide every child with an equal opportunity. Article (32) mandated several measures in

the field of education, including the development of the educational system to achieve its purposes of development of every child's mental, physical, emotional, social, and moral traits (*khda.gov.ae*). To achieve the core goals of UAE Vision 2021 and UAE Centennial 2071, the UAE has taken enormous steps towards equipping the young generations with the needed skills for the future. It issued a report title: (*The UAE Green Jobs Program: Jobs and Skills for the UAE's Green Economy Transformation*) (moccae.gov.ae). According to the report, the number of green jobs will rise from (49,520 in 2018 to 83,422 by 2030). The different national approaches in the UAE focus on youth and in implementing various development programs to help the highly able students achieve their potential in different types and domains of giftedness and talents. The UAE National Innovation strategy focuses on education as well as other aspects such as health, technology, transport, and others to represent a mixture of opportunities and challenges for the young students and long-life learners in the UAE.

Development of Talents (MOE)

The UAE (MOE) as well issued the (*Advanced Skills Strategy*) in November 2018 (government.ae), which sets out the national framework that will consolidate the concept of lifelong learning for citizens and residents of the UAE. The Advanced Sciences Agenda launched in 2018 aims as well to support the (UAE's Centennial Plan 2071) by focusing on (8) scientific priorities and (three) consecutive strategies from 2018-2021(government.ae.)

The (MOE) implements the developmental programs at an early age. Not only because children learn faster, but also to ensure providing the highly ables students' all types of development opportunities. (moe.gov.ae)

Development of Talents (MCKD)

The talent development at (MCKD) focuses mainly on implementing different enrichment programs and initiatives such as the summer camp for children and young people. This camp provides them with skills development in the fields of science, arts, design, and architecture. The camp stimulates the participants' passion for acquiring knowledge, as well as alerting them regarding future jobs based on their aspirations. However, it is not an ongoing camp, and students join it during summer only.

Development of Talents (ADEK)

The Department of Education and Knowledge (ADEK) launched its (2018) second edition of talent Discovery Program) (Adek.abudhabi.ae). This initiative nurtures eight fields of arts such as painting, colouring, graphic design, photography, music, ballet dancing, traditional performing singing, and acting. It accepts all students, citizens, and residents, from grades one through twelve, who study in public and private schools in Abu Dhabi. This initiative aim is to identify the highly able students in visual and performing arts. Following they undergo training and development to sharpen their skills. Each talented student has an individual development plan tailored to his or her potential and to ensure that all students have access to their talent-based programs. The program includes mentoring support, training and scholarships to enlist them in the world's leading universities. It also offers these students a variety of opportunities to showcase their products in local and international events and exhibitions. The first phase of the program is experimental. It accepts only 90 of the highly able students from various technical fields in it (Adek.abdudhabi.ae).

Development of Talents (KHDA)

Rahhal is a talent development program, initiated by (KHD) to enlist any desired institution to provide learning that supports transforming all people into lifelong learners."

Rahal's project focuses on preparing students for life experiences by allowing educational institutions and service providers to deliver quality services to expand their knowledge. It offers a platform that helps integrate life with education and enables students to study in regular schools, as well as partially follow up on another place in society, within an innovative system that includes assessment tools, and allows students and learners to seek and benefit from learning opportunities. It also enables students to acquire and recognise life skills on an equal footing with academic skills, thereby driving Dubai's regional and international leadership in promoting education. Rahal promotes educational opportunities for all members of the community, whether children or adults, and will support learners who are motivated, talented students, and individuals with unique abilities. At the same time, Rahhal program provides adults with flexible education that can be used to enhance their careers or enrich their lives(khda.gov.ae).

Development of Gifts (HA)

The gifted development at (HA) depends on the National Plan for Gifted Education(ha.ae). This plan implements the latest practices in the field of gifted programs and education. It is a framework for all methods applied to promote talents and giftedness. (HA) Believes in the importance of caring for gifted. Therefore, it is working towards a qualitative shift in this area. The National Plan consists of seven significant talents'-oriented programs integrated to discover and nurture talented people as follows: "1. Discoverable students. 2. Sponsorship programs for gifted students. 3.

Professional teachers in the field of giftedness. 4. Awareness and scientific dissemination in the field of talent. 5. Partnerships with relevant institutions in this area. 6. Hamdan Bin Rashid Al Maktoum Center for Talent and Creativity. 7. Hamdan Schools for gifted students" (ha.ae). The Hamdan Center for Talent and Innovation is the first to provide care, guidance, training, and digital manufacturing opportunities to students in public schools. The centre includes two sections: One is for discovery and nurturing of the gifted students, while the other is for the services provided to them in the Fab Lab. The Fab Lab is the place where innovators and digital manufacturers nurtured. In addition to that, there are mentoring programs for the students and their parents, including workshops and training programs. They aim at providing gifted students with the latest scientific and engineering techniques in various fields of science. They constitute of robotics, aeronautics and space industry, as well as Mathematics and Science. Training programs in this institution support the development of the gifted students' personal, scientific, and leadership aspects. They also promote self-learning and creative thinking skills. The second section is the innovation centre, established according to the international standards and specifications. It provides logistical support to (8,000) beneficiaries annually. The centre succeeded in identifying (225) gifted students. (47%) Males and (53%) females. The Center aims to provide seven enrichment programs annually to develop different interests and gifts among students during the year (2018).

(3) Challenges

Current attitudes to inspiring and celebrating talent may be counterproductive to identifying and supporting talent's development. Many see talent as something elusive, that is hard to locate, and for which there is a narrow window of opportunity for identification. As a result, this is demotivating for parents, children, and those that support them, leading to a feeling of helplessness and paralysis. Each stakeholder within the talent ecosystem believes that the responsibility towards

talent development falls on someone else. The government has employed the responsibility of the highly able students to the educational institutions. However, parents think it is all the responsibility of schools and teachers. These students, additionally believe everyone around them, should be offering the services. This diffusion of responsibility has contributed to the "bystander effect," and no one taking enough action.

Implementing development for the highly able students alone will not be enough, nor it will solve the challenges. Decision-makers in education must also focus on regulating the development of other skills (e.g., business creativity, resilience) that would enable talented people to become responsible for their talents and build their successful careers. Conversations have primarily focused on developing talents. However, talent is a small component of success. It will be essential to support their skills that are necessary for the desired development.

Several studies, papers and previous interviews have pointed at the missing federal law that safeguards talented individual's rights to be educated by special gifted education. The existing law and policies safeguard the rights of disabled people with special needs while gifted and talented individuals lack such service.

Talented children can sometimes suffer while fitting in with their average peers in a regular classroom. Gifted and talented students do not have the facility of education by an exceptionally gifted curriculum or skills development program. The children with high IQ and advanced capabilities sit for hours in typical classrooms and listen to ordinary lectures. Moreover, it is usual that a parent of a profoundly gifted student might feel isolated as well. Among the root causes of this isolation, according to an expert in talents' development, is the absence of awareness education and training for parents, which can leave them feeling that they lack the knowledge they need to handle and develop a gifted child. (The national.ae)

The lack of awareness drives the fact that there is a lack of appreciation for being talented in arts or any other artistic domain in the community. It is one of the most substantial challenges, as this means there is a need to change the mindset to appreciate the importance of talents and understand why we need them. The society should be aware of the importance of gifts and talents as investments in the long run, besides, becoming responsible towards them. Development of talents is not the sole responsibility of the educational institutions; institutions are also liable to educate parents, highly able students and the society on how to deal with gifted and talented individuals and how to support them.

Finally, the following chapter discusses the findings obtained from the methods used in this study.

Chapter Six. Discussions, Recommendations & Conclusion

Introduction

This chapter concludes this study by discussing the key findings of the research presented in the previous chapter to address the research questions. The questions centred on the role of five educational institutions to implement gifted and talented identification and development programs in the UAE. The goal of this study was to explore their achievements and provide recommendations to help improve the implementation of development programs. The gained results from data and information taken from the interviews focus group discussions, and document reviews were about the effectiveness of implementing such development programs on the talented students. A summary of the research and findings discussed and interpreted. The findings and discussions provide implications and limitations of the study. However, the study recommends future research in this issue and other related to the area of highly able students.

The design of the research questions was to guide the research to obtain descriptive data that would provide the study insight into the selected initiatives' current practices.

RQ1.What is the role of development programs implemented by governmental, educational institutions to develop gifted and talented students in the UAE?

RQ2. How are gifted and talented students defined and identified in the UAE?

RQ3. How is the development of gifted and talented students implemented in the UAE?

RQ4. What are the main challenges facing the gifted and talented identification and development in the UAE?

6.1 Discussing Key Findings

The discussion implemented within this section is upon the four themes derived from the analysis of the findings of data gained from interviews, focus groups, and document review.

The Main Themes

Awareness and knowledge, Misconception of definitions and identification. Development processes, procedures, and challenges.

6.1.2 Gifted and talented ecosystem

The findings, in general, revealed the existence of several gaps in the gifted and talented ecosystem. These gaps reflect in the process of implementing the development programs and the duration of these development programs in addition to the evaluation process of these developmental programs.

The design of the current study was a qualitative method. It accordingly urged gathering and analysis of data based on interviews, focus groups, and document review. The results of the data analysis showed the information obtained from different sources. However, this research did not intend to criticise any institution. The participants included parents of talented students, experts in the field of gifted and talented, gifted program managers, teachers of gifted, talented, gifted and talented students, and gifted and talented adults as well. They were responding to the questions from their perspectives and without pointing at any institution or their responsibility.

The researcher tried to include different participants in the interviews selected from the critical relations around the talented students and those who affect their identification and the journey of their talents' development. The teachers and program managers come from different educational institutions or had experience in similar institutions. Some of these teachers are graduates of gifted

diplomats and programs. These teachers implement some development programs; therefore, if they thought that some of these programs lack a variety of essentials, they intended to support their improvement. Several participants, such as parents and highly able students, agreed that they do not know about many development programs. They revealed that they are not aware of any policy that exists. Others said that several programs lack specific enablers to be capable of benefiting gifted and talented students. Most of the participants believed that most of the highly able students do not receive appropriate services in public schools according to their different needs. Others believe that these programs are not mature enough to fulfil their objectives. Accordingly, below are the four key themes that lead the discussion in this chapter.

This chapter discusses the findings presented in the previous chapter to address the research questions. The researcher, therefore, implemented ethical principles to collect, analyse, interpret, and report the findings. Therefore, the researcher suggests that educational institutions should trust the conclusions of this study if they choose to implement the recommendations.

RQ(1). What is the role of development programs implemented by governmental, educational institutions to develop gifted and talented students in the UAE?

Findings from the collected data make it significant to reveal that the implementation of development programs does not fulfil the needs of these highly able students. The documents of these institutions and the published reports submitted to the UNESCO show the concern and the interest of the (MOE) in supporting highly able students (Al Obaidli, 2006). The findings from the data collected from the interviews and the focus groups did not reflect this concern and interest.

According to (Gaad et al., 2006), "the provisions and the programs found in government schools do not mirror such care and the significance of the educational institutions" (*Gaad et al.*, 2006).

The interviews with experts and teachers and the analysis of their published documents show that a national policy that provides these institutions with a framework to implement and evaluate standardised development programs is missing. Furthermore, there is no training or awareness plan to encourage the value of talents among the parents. Their role seems to be limited to implementing such programs.

Several teachers who participated in the study believe that it is essential to implement development programs for highly able students. Others indicated that the development programs implemented and exist in schools. Most of them seem not aware that these programs' need enablers to make them beneficial and sustainable. These enablers represent a policy as a framework to define gifted and talented, regulates development programs, implement continuous teachers' professional development, define program standards, provide programs constant evaluation, and a clear gifted and talented explanation. The definitions of the highly able students or the identification process are highly critical. Teachers and parents need to acknowledge them to provide these students with differentiated development programs that suit their abilities and type of talent. A research study examined the issues regarding the provision in the performing arts, sport, and found that the identification of talented in art base on performance only and ignores the other dimensions like teamwork, leadership, commitment, and academic skills (Neelands et al., 2005). Unfortunately, there is a little awareness and few published documents that give information about the definition of the highly abled and the standards of the identification programs. There is no guidance to guide the educators on such procedures.

Most of the programs provided for the highly able students in the country are limited to field trips, enrichment after school programs, and competitions. They do not help much to elevate and promote their abilities and talents. (Al Obaidili, 2006). The after-school enrichment programs are not used equally by all schools nor are all activated. Also, many teachers are inexperienced; they have received little training in implementing development programs. Other enablers such as resource rooms, technological tools, clubs, and other skills training courses do not seem as designed for highly able students in every school, and if they are available, they do not use them as teaching aids but more as entertainment centres for all the students.

Development and identification programs and processes require more variety, widely conducted, or used appropriately. The data gained from the interviews and the focus groups have reflected how the identification of the highly abled students based mostly on achievement tests and teachers' nominations. Creativity tests, psychological and social evaluations, and IQ tests are not standard methods. There is a need for proper identification process that will meet the highly able needs with essential support. Moreover, highly able students may be stressed and unhappy if they do not get their essential needs. This difficulty might affect their behavioural and emotional attitudes with conflicts, underachievement, and leads to harmful behaviours. Such behaviours include being overly aggressive, the tendency to violence, school dropout, refusal to work with others, and suicide (Davis & Rimm, 2004). Many gifted children might have a higher risk of suicide (Webb, J.T., Gore. & Amend, 2007).

In conclusion, highly able students deserve an appropriate education and an opportunity to get their needs for development to become productive and skilful. However, several gaps in the development services need a focused concern in order to address them adequately.

RQ(2): How are gifted and talented students defined and identified in the UAE?

The interviews and focus groups reflected the absence of formal provisions that regulate identification and development programs. It is due to the lack of national policy according to them. Moreover, there is a misconception regarding the gifted and talented definitions, which makes them vague and not conducted appropriately in most of the schools. The giftedness referred to as the potential for higher academic achievement in sciences or languages, while the talents include the performing arts and sports. However, not everyone agrees with the difference between giftedness and talents (Gardner, 2000). For example, some said that there is no reason to separate gifted and talented into quite distinct areas. Others suggested that this separation refers to a view that promotes academic subjects for being more critical than arts. (Ofsted, 2005) stated, "It has been much more difficult for schools to identify the students potential rather than their realised talent. "The gifted and talented students assigned in regular classes that offer regular students regular unchallenging education. Gifted education is missing, and a curriculum that supports the gifted and talented development after identification is not there.

The identification programs need development, and their procedures need improvements too. They are not obliged to specific timing and inexperienced teachers implement most of them. They do not evaluate the students' progress, and they are random and insufficient. The UK teacher net, which is the UK government website for teachers, recommends "Identification of highly able students, must be an ongoing process by multiple sources and strategies." The UK teacher agreed that students' abilities emerge at different ages, and they develop in different circumstances (Attfield, 2009). According to the findings from interviews and focus groups, experts, and teachers indicated that the identification of the highly able students is not varied or multidimensional. Therefore, institutions need to consider the advantages and the disadvantages of the identification

tools before deciding on them and to agree on a method that identifies the highly able students. The identification of the underachievers who have high abilities is an essential process too, and It needs to prioritise in schools. Identifying gifted and talented students by their potential may sometimes fail to recognise them as capable, and this will possibly frustrate them and force them to demonstrate awkward behaviour.

The (MOE) already developed a gifted and talented policy that contains the essential provisions such as a definition. It organises identification and development programs too. However, it is still a draft awaiting approvals to go in force. The (MOE) role is to change and implement reforms in education to provide all students with a high standard education system and to support their development (UAE Vision and National Agenda 2021). However, there are no laws or policies that serve highly able students yet (Mazumder, 2016). The UAE government pays much attention to gifted and talented identification and development. It is the highest priority in the country, and this should make the educational institutions eager to develop a policy that supports this attention and helps to implement the proper identification procedure. The Ministerial Resolution no.94/2(2002), regulates the institutions' role towards highly able students. They should provide these students with development programs along with education. These educational institutions need to impose ongoing professional training for the teachers and provide consultation for the highly able students and their stakeholders. Some of the plans implemented for these students are, on the other hand, still new and in need for more time to evaluate their success.

According to a guide booklet (inspection framework) published by the (MOE) and includes definitions, characteristics, application forms, checklists, and teaching strategies, UAE public schools should set plans for gifted and identification programs (moe.gov.ae). Teachers of the highly able students must complete an ongoing assessment and update their profiles to support

their identification and support their differentiated needs. The results of the interviews and focus groups conducted with educators, parents, and talented and gifted students, showed that most of them did not go through the mentioned identification process. They were not aware of many identification processes and methods or any of the teaching strategies that encourage gifted differentiated teaching and education. Most of the development programs were limited to enrichment programs, and they were not evaluated or accredited. The exemption of development programs from the school curriculum makes these programs look like additional unnecessary programs and not as core programs intended for the highly able students' ongoing development.

Furthermore, educators think it is not possible to implement any differentiation or compact curriculum within the regular classes where different abilities included. Parents complain of not having time to send their children to additional development programs offered to them after school while some of them believe there is no need to send their highly able children to such programs because these programs are not included in the school core formal curriculum, not thoroughly planned and will not affect their children's academic achievement. As not all the educational institutions can interfere in the schools' plans, the (MOE) which is the educational regulating authority in the country, has encouraged schools to give time for these programs and to allow students to participate in them. According to the (MOE) objectives in its Strategic Plan (2017-2021), it aims to "ensure quality, efficiency and good governance of the educational and institutional performance, including the delivery of teaching" (moe,gov.ae), though, and as findings showed, most of the development programs can be incompetent if they lack the appropriate planning. Furthermore, many of these development programs, as said, are limited to function as extra-curricular activities, and most of these programs need specialists to assist the highly able students while participating and gaining some benefits.

In conclusion, the status of gifted and talented identification and development programs seems in need of more in-depth exploration. Few programs are implemented for limited domains of talents while there are no comprehensive, well-organised plans for them. The identification programs need effective processes and procedures based on clear definitions and an understanding of the description and characteristics of giftedness and talent. There are no policies established to regulate the services offered to the highly able students. These services provided with misperception in schools, and a comprehensive national strategy is highly needed to organise the scene.

Moreover, many programs do not exist equally in all schools. There is a shortage of qualified or trained teachers who can contribute effectively in implementing the identification programs. Highly able students still enrol in regular classes, and such classes do not provide specialised services like differentiated curriculum or compacted curriculum. Additionally, most of the programs do not follow the identification methods, and the first approach depends on the achievement scores and teachers' nominations.

RQ(3): How is the development of gifted and talented students implemented in the UAE?

According to the researching instruments used in this study, including document review, interviews, and focus groups', their findings approved that the absence of the policy affects the effectiveness and quality of any implemented development program. The interviewed experts and teachers admitted that despite the existence of a few plans to improve the quality of education and to meet the needs of the highly able students, challenges remain unmoved due to the absence of regulations. The non-existence of identification and development standards minimises their effect and influence. Each institute has its standards, and this confuses the field. There is a great need for an integrated, unified approach and a national strategy that dictates regulations and plans. The UAE is witnessing more launching of advance development programs dedicated to the

development of the highly able persons in it. However, the absence of a policy that regulates and aligns the developmental programs with the needs of all the users from the highly able students in schools makes these programs unappreciated.

The policy is essential to organise and mandate gifted education and development for any highly able student at any time and under any circumstances. This education should become mandatory to support the identification of these students and infuse the services for them. The UAE's first education system incepted at the establishment of the UAE in 1971. This education since its last major updating in 2014, has transformed its practices to meet the country's future needs by adding new subjects in the curriculum, such as design and technology. Policies regarding the education systems developed, reformed and modified over the years. However, no individual policy oversees the existing situation and adapts to world changes and talents' development challenges. The absence of a system for gifted and talented students results in not having any guidelines or regulations that support their development. The educational institutions are responsible for the implementation of several programs, including programs for the welfare of students. UAE launched several initiatives to boost the development of strategies that would help gifted and talented student's growth, including competitions and arts performing auditions. However, from the findings of the conducted interviews, focus groups and documents' analysis, the absence of a written published official policy regarding the gifted and talented identification and development remains an unsolved challenge.

Having a policy for the development of the highly able students supports as well the country's development. Nurturing and meeting their unique needs is everyone's' responsibility. Adapting the education system by particular policies will enhance the abilities of this category and ensure that the UAE benefits from their skills, unique talents, and aptitudes. There should be a collective

understanding and agreement amongst all stakeholders on how to define, identify and serve the gifted and talented. Providing these services and having sustainable financial support should also be shared.

In conclusion, the UAE requires a state policy that organises the gifted education and development and provides the highly able students and individuals with their needs. There should be a clear definition that defines who is gifted and who is talented. The desired policy will also regulate and organise the application of the identification and development programs. It will allocate the responsibilities of each authority and institution towards gifted and talented services and needs for development. It will determine the standards of the identification and the development programs and will give instructions and regulations on who is qualified for gifted and talented identification and development. The policy should regulate the plans and the strategies to offer the gifted and talented the appropriate development. It will ensure all gifted and talented students will receive suitable programs, have access to all of the institutions that provide them with development initiatives and programs, receive equal education, have access to various activities, and makes sure that they are satisfied and challenged in their schools.

RQ(4): What are the main challenges facing the gifted and talented identification and development in the UAE?

The UAE focuses on this valuable category to pursue its economic development and step in the future. It exerts efforts to elevate the quality of education and the development of highly able students. To do that, these students need a unified definition that helps in identifying and developing their skills and abilities appropriately. The absence of a national policy that organises their education and the implementation of their development programs agreeing to standards increases the challenges faced by this category. Furthermore, this issue would be more

complicated, knowing that each educational institution in the UAE is following a different framework and definitions to describe giftedness and talents. (khda.gov.ae).

Schools and parents should collaboratively exert efforts to secure a thriving environment in which all gifted and talented children can blossom and find what helps them to grow and not to become disruptive or reclusive. The absence of a policy that mandates processes and procedures such as definition and identification leaves them to the estimations of their parents and teachers who need to have enough experience to spot them. The usual talented and gifted provisions and practices applied now are random and not systematic, and they do not depend on an institutional philosophy due to not relying on a regulating policy.

The other most significant challenge is the implementation of identification programs by non-experienced teachers and providing highly able students with their needs of development through non-qualified and unable programs. Identifying highly able students should be a significant priority for any institution and school. Failing to distinguish this category occurs due to an inappropriate implementation of the identification method or as a reason for not implementing it in time or by people, as it should be. The identification method has both advantages and disadvantages, and this may lead to these students going unrecognised and non-talented identified as talented. Being gifted or talented and not getting the attention deserved or not challenged in the classroom leads to disruptive behaviour. The findings from the documents, interviews, and focus groups stated that many gifted and talented students hide their abilities to stay with their friends in their class. They avoid that by performing poorly in classes, and they show their distress more than their happiness if they feel they would be isolated from their peers. It is the responsibility of all educators and staff in schools to identify the highly able students and to meet their needs. A lead teacher may be responsible for issues related to these students. However, this responsibility is unlikely to have a

significant impact without the support and help of the school team and parents. Most kindergarten children, as well, display gifted and talented characteristics. However, if people fail to identify them at that time, their talent will rapidly decrease as they progress through formal education.

The educational institutions are aware that these students need early identification and more development programs. However, they also know that schools have difficulty in balancing between trying to develop their abilities while preparing them for the traditional exam-based preuniversity qualifications. (The national. ae, 2015).

Parents, on the other hand, are having an issue amongst providing a challenge for their children or allowing them to behave and be children. Putting more expectations on these students as well as pressure or demands may ruin their lives. However, believing that their talents develop in their own time is a challenge if we know that this talent may disappear if not spotted at a time.

Participants in the interviews and focus groups pointed out challenges with financial support. The little support for the nonacademic subjects makes them unappreciated and undesirable. Scholarships granted for science or (STEM) only. This makes arts and cultural studies unappreciated. Developing few instruments for music practices interprets music as not a desirable subject, lack of continuous and sustainable identification process and lack of specialised teachers who can identify such highly able students risk losing more skilled and talented individuals. All these forces push the talented students in arts and culture to study another academic subject. Nonacademic subjects are, on the other hand, very costly if the highly able student in this field wanted to pursue them outside their schools or universities, and they are not available in nearby locations nor sufficiently organised.

Finally, the current taught curriculum provision is inadequate to fulfil the talented student's potential, and the curriculum is unable to accommodate all types of gifted and talents because there is no specific differentiation or challenging curriculum for talented and gifted students.

Table 38. Theme Analysis in alignment with research questions

1	Table 38. Theme Analysis in alignment with research questions				
What is the role of development programs implemented by governmental, educational institutions to develop gifted and talented students in the UAE? How are gifted and talented students defined and identified in the UAE?	Knowledge and awareness What participants know Definition and Identification Participants attitudes	The family and community do not have adequate knowledge about giftedness and talents, and that affects their support for their children. There was a lack of knowledge among the participants about the existence of a talents' services policy or development programs. There is little training given to teachers who oversee talented training. There was a lack of teachers 'continuous training and development. The talented programs are enrichment programs, and not considered as talented education because they are not sustainable and not included in the school curriculum. There is no explicit guidance or awareness on talented education, rather general principles. Selected teachers to implement the identification or development programs are not qualified enough. Human capital: the participants raised the absence of national policy as the cause of lack of specialist professionals to identify or to provide counseling, coaching, and training for gifted and talented students.			
How is the development of gifted and talented students implemented in the UAE?	Development of talented students	Development programs are not following any policy or standards, teachers teach subject content, and there is no attention paid to develop the students' skills.			
What are the main challenges facing the gifted and talented identification and development in the UAE?	Challenges faced by parents/talented students Participants recommendations	Absence of a national gifted and talented policy to regulate gifted and talented services. Absence of a gifted and talented law to equalize, mandate, and regulate the gifted and talented services. Lack of financial support: Participants pointed out challenges with financial backing such as little support for the non-academic subjects; scholarships, instruments, teachers, students can study one speciality in the university, nonacademic subjects are costly, lack of differentiation in teaching in schools. Curriculum provision: Taught curriculum is inadequate to fulfil the talented student's potential, the curriculum is unable to accommodate all types of gifted and talents; there is no differentiation or challenging curriculum for talented and gifted students. The current talented and gifted provisions and practices are random and not systematic, and not based on an institutional philosophy due to the absence of national policy. There is no overall organization to regulate the giftedness and talent practices. Lack of collaboration amongst the institutions to provide gifted and talented development programs. The current G&T practice is not sustainable.			

In conclusion, the absence of an overall organisation that regulates the giftedness and talent practices, processes, and education is a considerable challenge. Lack of collaboration and coordination between the organisations, associations, and institutions to provide development programs to highly able students and to ensure their sustainability makes addressing their needs critical. Without agreement among stakeholders around the roles and responsibilities of each of them, issues and challenges that hinder the implementation of development are likely to remain, affecting negatively on this category's progress and productivity in the future.

6.1.3 Recommendations

Table 39.Interview Questions in alignment with recommendations

Tuble 55 times view Questions in unfilment with recommendations			
	What are your suggestions for more talent's development in the UAE?	Experts-	
		Teachers-	
		Managers	
		Parents	
		Gifted &	
		Talented	
Suggestions & Recommend ations	What do you suggest investing in your child's abilities and talents?	Parents	
	What are your suggestions and recommendations regarding talents' development in the UAE?	Parents	
	What are your recommendations regarding increasing talents' services in the UAE?	Gifted & Talented	

This study explored the implementation process of some of the development programs by five prominent educational institutions in the UAE. The study constructed on an exploratory sequential qualitative method design is to give applicable recommendations regarding improving the situation of highly able students' development in the UAE. One of the research questions was about the challenges faced by these institutions while implementing development programs and initiatives.

This research was to answer the questions and to find out the type of appropriate recommendations to provide throughout this study.

The question was as follows: What are the main challenges facing the gifted and talented identification and development in the UAE?

The findings obtained from the collected and analysed data in chapter five, along with the reviewed related literature in chapter two, are to offer the suggestions proposed in this chapter.

These recommendations intended though to enhance and improve gifted and talented service and development in the UAE.

Recommendations follow the arrangement of the topics. They are not final recommendations, but they are significant and a few depending on the results of this study.

Recommendations include the following areas: National Policy for highly able persons, the definition, identification and development of these highly able, methods, resources, and support.

Recommendations for further research available at the end of this chapter.

6.1.4 National Policy

Meeting the educational needs of the highly able students is not about providing something entirely different, but about constructing the new needs on proper general school provision (Eyre & McClure, 2001).

Education policy, according to researchers, is a strategy implemented by a governing board, driven by an educational issue that existed at a time (Van Tassel-Baska, 2006). This policy aims at attending to excellence in education. (Rasmussen & Lingard, 2018) It attempts to raise educational standards in general and to identify the highly able students to make special provisions for them.

Proper provision for these students is one of the most critical components of the personalisation and equal opportunities agendas that drive recent government initiatives. (Eyre &Lowe, 2013)

Changing an existing policy or improving its contents is not an easy task, and it needs to perform through different stages, moreover, improvement efforts in general education policies (AlObaidli, 2006). There is always a need for written policies by the decision-makers who govern and approve the policies. This policy is assumed to mandate the identification, development programs and any reasonable provisions for highly able students' progression. Denmark, for example, is an excellent practice regarding equality-oriented tradition in education. It pursued strategies of developing talent throughout the educational system to cater more to the needs of gifted and talented students in its schools (Rasmussen&Lingard, 2018).

Consequently, policies are penetrating to the different needs of the gifted and talented students served in schools. They could interpret and implement these services efficiently and appropriately if they were sensitive to their needs. They should positively affect gifted and talented students lives by providing them with benefits and systematic services. The policy for highly able individuals must support the needs of gifted and talented students and persons in general (Gallagher &Reis, 2004). To design a policy for the gifted and talented, it must be open to improvements and does not follow a standard model according to (Passow&Rudnitski, 1994).

The absence of a unified national policy for gifted and talented students in the UAE allowed the issuance of the variance of policies that were designed individually and by different institutions. The policies left to the efforts of private schools and institutions, and this confuses the national reform regarding gifted education and makes it less cohesive, comprehensive, and inclusive than it should be(Van Tassel-Baska, 2009). Accordingly, there is a need for an overall educational

policy to go inline with the new global trends and developments. This policy design should base on the UAE vision of education and emerge from it's National Agenda's objectives.

The (UAE National Agenda 2021 and the UAE Centennial 2071), stand on four aspects that comprise education, economy, government development, and community cohesion. The goals of the UAE are focusing primarily on the educational system. The ability of this system to implement the required development, such as advanced technology, engineering would help the UAE achieve its goals (UAE cabinet.ae). Thus, the need for a policy that implements, regulates and organises gifted education is a vital requirement in the UAE. It has to ensure that these highly able students are assigned their opportunity to acquire development and skills needed to become independent, responsible, and to contribute to the UAE's achievements.

To fulfil its role in this area, the (MOE) in its capacity as the principal educational institution and the primary decision maker regarding the educational policies in the UAE must develop and administrate such policy to govern, fund and provide these students with their needs all over the UAE. The policy should include flawless guidelines and directions for the identification and development of these students. It should as well focus on their experiences and skills and should include clues on how gifted and talented students are ready to participate in the society and to be a part of an active life in which they can solve problems in any situation. The policy is to inspire the gifted and talented towards their country's service and to promote the characteristics of a loyal citizen. The form of this policy should be able to provide all highly able students, teachers, and any stakeholders involved their differentiated needs and supported to be able to support them. Its objective is to design and organise the ecosystem around the gifted and talented students in a way that makes it able to support their needs equally and as early as possible. It should include all professional development regulations and needs for different materials and required logistics and

resources including books, electronic tools and devices, identification tests, games, competitions, resources rooms, and all necessary learning resources that would support the creation of a supportive environment for them, their parents, teachers and counsellors. (Eyre &Lowe, 2013)

These well-developed policies form the foundations of sustainable education systems. They help countries achieve their educational development goals and contribute to continuous learning. (UNESCO.org). "Planning policy for the highly able students should not be limited to one institution. All stakeholders and institutions related to their education, well-being and services should be involved in this task. Their involvement should endorse awareness about the importance of their differentiated education and development. The policy creators may constitute various stakeholders and people from different categories such as experts, psychologists, media specialists, curriculum developers, educators, parents, and gifted and talented. It is crucial to define its purpose and set its intended goals ahead. The stakeholder's assembly who oversees the policy planning and setting should research and review current Acts and policies before placing a new one. The UAE philosophy of education must be consistent with how they vision gifted and talented education. The other educational institutions should be involved as well and observe the policy designing and implementation to ensure that their staff will be capable of the implementation of the policy. An evaluation plan should come within the policy with a clear purpose and vision. Institutions, programs and any other related stakeholders should be involved in the different stages of the evaluation plan of the policy. This evaluation plan will help all stakeholders, including the educational institutions; identify areas of success through the results. They will be able also to provide their recommendations and point at the areas of improvement and the required modifications. There must be a schedule set for follow up this evaluation plan at different stages. Another plan for conducting a policy review includes the assessment of the policy recommendations and the development of specific plans. The policy review may consist of consultations if requested by the policy developers to discuss the ideas and coordinate efforts with international agencies. This policy review helps the educational authorities reinforce their educational system and include any modification and improvements in it. It also contributes to the development of the educational authorities' system competencies. (UNESCO.org)

Having a policy developed and reviewed by all the stakeholders to look after these gifted and talented students is an efficient tool to regulate their services and help them satisfy their potential needs appropriately.

6.1.5 Unified Definition

Highly able students are individuals with exceptional abilities defined according to their skills and developed to a higher level than their year group (Eyre, 2012)."

Socrates and Plato viewed giftedness as the ability of a person to learn and gain higher knowledge and understanding faster than the others. In later centuries, the concept of giftedness moved from philosophers to scientists. People identified as gifted or talented only if they had made significant contributions to society like Galton, Mozart, Albert Einstein, and Thomas Edison (George, 2013). However, using this method limits the number of gifted and talented individuals and eliminates most of them from consideration. Very few people can make such contributions, and if the world followed this method, we would hardly find someone defined as gifted or talented.

The variety of definitions, terms, and criteria to describe these exceptional students and individuals have always created confusion among scholars and people in the societies in general. Thus, these students are not a similar group with similar qualities, but they represent a wide range of characteristics, and they do not demonstrate the same traits. (George, 2013)

The main characteristics of the highly able students to have extraordinary abilities and to be committed to what they do within or outside the school curriculum. Their gifts and talents may be in one or more areas, and it became apparent at different stages in their lives. We can find gifted and talented individuals in any group, including people with special needs, or underachievers in schools.

Consequently, recent gifted and talented definitions started to focus on identifying children who have the potential to excel as well as the children who have already shown excellence in some areas. (George, 2013)

The concept of intellectual intelligence moved to become multiple talents, according to (Gardner, 2000). The theory of multiple intelligence discards using one method to identify any gifted or talented student (George, 2013). Therefore, this should consider any child could be gifted or talented if he/she is abler than his /her peers are and had achieved better than those who are in his age. However, policymakers need to adopt a definition that will interpret the concept of "giftedness" and "talent." This definition should help determine the students who will be involved in a specific program of development. The definition is for the decision-makers to be clear about what decision they should make to secure the education and development of this category.

The definition that describes these gifted and talented must be accessible and comprehensible for everyone. It should include the UAE vision, philosophy of education, and the characteristics of the gifted and talented individuals in the country .it should provide all schools and educational institutions in the UAE with a clear description of their concepts.

This definition should help every highly able student get equal opportunities from a variety of services such as the development enrichment programs. This definition must help the school's

educators and staff acknowledge these students' needs and support them to get their differentiated academic, emotional, and social needs. However, it should not put on them any pressure or lead others to bully them or to assume higher expectations from them.

The definition should provide support to these students and not make people view them as non-humans. It is an essential issue to understand that gifted and talented individuals are humans with emotions and can suffer if they do not like what they do, or if not getting what they want. All educators, parents and stakeholders should recognise the gifted and talented students emotional, social and financial needs, understand their strengths and weaknesses and support them to become able to use their gifts and Tlnets properly. They may be superior in some areas and weak in others. They may be suffering from learning difficulties or any other social, physical, or behavioural challenges.

6.1.6 Identification Process

The identification process is what these highly able students' need to find their way and recognise their abilities. The crucial step in identification is to agree on how to identify them, when and who should do it. The main problem in the identification process is not having a unified definition that guides the identification methods and tools. Existing identification is mainly dependant on achievement tests, teachers and parents' nominations. Other identification methods are testing one dimension, while others are limited to a certain age.

The need for usage of more comprehensive assessment methods is to avoid ignoring many gifted and talented students. Their unqualified teacher or parents may not identify those tested by one tool.

The multidimensional method would identify those students who exhibit more than one talent. This method means using standardised assessment data, employing a multi-method approach to identify different areas of ability. It also means consulting a range of people around the students, monitoring the students in different situations, placing them away from the classroom, and watching them performing or doing anything out of class context.

Some methods of identification do not require specialists or expert's knowledge, such as screening checklists. A trained class teacher can complete the list and monitor the students who exhibit gifts or talents.

The identification process must start at an early age. Delaying this identification risks that this child is misunderstood and become an underachiever due to being unchallenged in a boring regular class.

Identification in the early years, according to (Freeman, 1998) is challenging. He highlighted four critical dimensions in the assessment of very young gifted and talented children: Lively minds, awareness, ability to learn, and independence. (George, 2013)

Renzulli's three-ring concept of giftedness (*Renzulli, J.S., 2016*), helped the educators for a long time look for more skills in their students and be eager to recognise their creativity combined with intellectuality and task commitment.

Moreover, identification should be a continuous, fair, and transparent process that does not neglect any child from any group. It should be flexible and able to include any child at any time. It should be flexible and comfortable enough to identify them at an early age or to identify those who will emerge later. Identification within the school must be a part of school life and curriculum. Teachers are the principal tool in the identification process. They should be able to spot any gifted

or talented student within a class or outside it, and his or her identification process should rely on a reliable tool, such as the portfolio approach supported with a range of different qualitative and quantitative data.

Identification should be an ongoing process and part of school life, and schools must prepare a challenging and supportive environment that helps these students emerge and exhibit their abilities. School leaders, teachers, or other educators in schools should be able to spot any hidden gifts or talents among students who come from all different cultural backgrounds. They should be aware that underachievers and students with specific learning or physicals disabilities can be gifted and talented with hidden gifts or hidden talents as well.

Open communications between educators in different educational institutions, and between teachers and gifted and talented students, and their parents would ensure all students appropriately identified and nurtured (Eyre, 2012)

6.1.7 Development Process & Systems

All highly able students should receive differentiated services, including a variety of development programs and initiatives that nurture their different abilities.

The most important provision is the ecosystem that surrounds these students inside and outside schools. Preparing this ecosystem makes it able to become a tool to identify and develop these students at early ages and throughout their lives. The gifted and talented needs differ from one to another. However, the emotional, psychological, social, and financial needs are the essentials.

The progress of the ecosystem starts with the development of the whole-school approach towards the highly able students. Having a policy that helps the schools or the institutions follow its mandating provisions would outline this development and structure it. Having a clear, smooth, and flexible, multidimensional identification methods would expand the identification process and the identified students. Ensuring that all teachers, responsible for teaching and guiding the highly able students, understand and know how to spot gifted or talented students would strengthen the process and increase its reliability. Teachers should keep a precise checklist and record of all gifted and talented students to follow up with their identification and development. The new technology tools made it easier currently to follow up on their development and record their achievements accurately. They should continuously review and monitor the gifted and talented students' progress to improve the school's environment to become able to facilitate the development of gifted and talented students (George, 2013).

Gifted and talented students need developmental programs provided to them in a variety of options that suits their differences and help them to satisfy their potentials equally. These programs may include an extension provided to gifted and talented students within the mainstream classroom. The extension involves these students who follow the regular curriculum but to a different complexity. Effective teachers who can provide these highly able students with the extension program are those who possess good knowledge of the subject, and they work with at a higher level than they usually work at with an age group. They must be with excellent pedagogical skills and the ability to recognise the student's abilities and personalities (Attfield, 2009).

Other programs like acceleration apply to one or more than a subject, and it involves moving the gifted or talented students to work with a group of older students.

Grouping these students based on their abilities to meet their different needs can help these gifted and talented students in a variety of ways. Forming different settings could be a way that supports all types of abilities. The grouping includes different types of groups according to their abilities,

learning style, shared interest and socialising interests. The most crucial aspect of this program is planning the educational objective and reason for it. Different forms of grouping can provide productive learning and help gifted and talented students stay motivated. (Attfield, 2009)

Compacting describes a shortening of course materials. It happens when a gifted or talented student is familiar with a significant portion of course content, and this makes this student cover the course in less than the required amount of time.

Mentoring is also another tool used for regular and highly able students' empowerment. It has proved to be useful and supportive for the gifted and talented students. It helped them plan their ambitions and know-how to reach their potential. The mentor-mentee relationship depends on time arrangement, mutual trust and respect, and some other requirements to make it successful. The right mentor helps the gifted and talented students take hold of the proper opportunities, plan and come up with their solutions, develop confidence, maximise their potential, develop their skills, and gain self-awareness (Long, Fallon, Murphy, Derbyshire & Hewitt-Craft, 2010)

Enrichment programs proved to result in more benefits in learning for these highly able students. These students enjoy new and supplementary experiences and different opportunities outside the standard curriculum. This enrichment program may include trips, clubs, museums, and galleries visits, the use of different resources, developing a broader curriculum to enrich and enhance provision, additional languages, drama, music, astronomy, debating groups, chess, community projects, competitions, and many other projects or intellectual activities(Attfield, 2009)

Offering a variety of programs ensures providing every gifted and talented student with the best developmental opportunity according to their different needs and requirements. To design an

individual development program for each gifted and talented student ensures that each student has access to the best developmental opportunity.

Creativity also is an essential attribute for the development of gifted and talented potential because they need special services and needs while learning in schools. However, school context is not developing creativity properly, nor it offers creative solutions or methods for this category within curriculums or environment (Piske, F.H.R, Stoltz & Machado, 2014)

Teachers need to be creative by differentiating their teaching methods and students' learning styles in order to allow highly able students to improve their learning in an inspiring environment. Utilising the available community centres in the UAE can help schools provide their exceptional students as well as other students, in general, an opportunity to learn through different methods and places outside school days

The continuous reforms in the UAE educational system provide an opportunity to improve the curriculum and to ensure that the education system is adapting all developmental programs and is flexible enough to implement them quickly and for every student. Teachers, on the other hand, should be involved in continuing professional development to be able to apply differentiated teaching methods and approaches to meet the needs of different student's abilities in one classroom. They are required to design different activities and a variety of methods and programs in a challenging way.

It is, therefore, essential to note that many factors can stimulate different improvements in educational institutions. Such improvements would support the development of more sustainable beneficial programs that ensure gifted and talented meet their needs and can satisfy their potentials.

This study has provided some recommendations while there are many more that need more researches and studies. The significant recommendation is having a national policy agreed upon by all stakeholders to organise and mandate the gifted and talented quality services in the UAE. This policy will regulate and mandate identification and development programs and will ensure that they will be applied appropriately to develop these students' abilities and potentials. All gifted and talented students will get support through accredited, high-quality programs.

Developing a policy means all educational institutions need to collaborate and agree on how to implement new services to identify and develop the highly able students and to improve the existing services to make them achieve sustainable progress all over the UAE. This policy will impose crucial changes to the current systems and programs. It will monitor the process and will require adhering to international standards and criteria. It will develop more plans in favour of gifted and talented ongoing development.

6.2 Recommendation for Future Research

* The current situation of gifted and talented identification and development programs in the UAE needs further studies and researches that aim to explore and investigate it in-depth. These recommended researches should be able to examine the effectiveness of the continuous establishment of nurturing initiatives and if they are providing these highly able students with their differentiated needs?

* The evaluation of the rapid changes in the UAE and the investigation of the educational system improvements made to fulfil the (UAE Vision 2021) and the UAE preparations for the (UAE Centennial 2071), needs to be researched and investigated.

*Gifted and talented different physical, emotional, social, financial, and psychological needs that studies ignore and to seek more topics related to this category as humans.

* To conduct longitude studies, track the changes in the UAE society and people's mindsets; to study the social effects of this rapid advancement on the awareness and knowledge of the people regarding the value of gifted and talented individuals.

Conclusion

This study aimed to explore the area of identification and development programs implemented by five educational institutions in the UAE for the sake of highly able students development and progress. It aimed as well to give appropriate recommendations according to the results and findings. A carried out comprehensive literature to understand various concepts and theories behind the giftedness and talents. Different research instruments used in this study, including interviews, focus groups, document reviews and library search obtained more data and extended information. This data was studied and analysed to identify the services provided for gifted and talented students and the programs implemented for the sake of their development in the UAE.

Realizing the enormous amounts of global books, studies, researches, articles, journals, conference papers and debates conducted about the issues of the gifted and talented worldwide, makes this research a significant addition to the UAE's efforts to promote and support the importance of these students and the necessity of implementing the most reliable identification and development programs and tools in the country. It was also significant to learn more about different theories and debates provided to the field by prominent scholars, psychologists, thinkers, politicians, journalists, and researchers to enrich literature and help any future researchers conduct replicated researches and studies in their countries.

As this study was born to investigate the role of five educational institutions in the UAE, it aimed accordingly to explore different stakeholders opinions by interviewing four main categories in this area. It applied various methods and planning tools to answer the questions of the research.

This study based its rationale on the urgent need for proper identification and development tools to support the highly able students and prepare them to contribute to the aspirations of the UAE according to its (UAE National Agenda, UAE Vision 2021, and UAE Centennial 2071).

Moreover, these exceptional students are the future investment and the country's hope to support its economy as a replacement for oil if provided with the appropriate development after being correctly identified.

The study and the analysed findings found out that the identification and development processes and programs, despite their excessive availability in the UAE are still in need of crucial improvements and modifications. While the five educational institutions subject of the study provides a variety of these identification methods and development programs for gifted and talented students, however, some programs are in need for additional enablers, more enhancements, and various necessary modifications to achieve their roles. The educational institutions need to collaborate and communicate to evaluate these programs and mutually plan to elevate their effectiveness and exchange experiences in this regard. They need to extend an appropriate ecosystem that would provide highly able students inside and outside schools with the proper services and development. Many of these programs require governing standards, provisions, and criteria applied through a national policy that regulates the implementation and quality of the services provided for the gifted and talented students in the UAE.

Implementing appropriate, highly qualified programs in all public schools would effectively support the proper identification and development of highly able students. They as well provide them with an equal opportunity through differentiated services.

Furthermore, a national policy unifies and regulates the gifted and talented provisions. There is also a need for an appropriate comprehensive description of the concepts of gifted and talented. Having a flexible, modernised definition should allow all of the highly able students with different abilities to satisfy their needs and obtain customised services to thrive. The definition should describe and reflect the cultural values and the leadership vision of the UAE. Moreover, identification methods should be multidimensional and not depending on one tool or process to avoid excluding many gifted and talented students.

In conclusion, the study found that there is a critical need for improvements that include the improvements in the identification and development programs provided to the gifted and talented students in the UAE. The different gaps that disable their efficiency and prevent the gifted and talented students from benefiting out of them require certain amendments and modifications to make them beneficial and reliable and to build a more comprehensive ecosystem that meets the needs of all students in general and gifted and talented students.

The role of the five institutions towards implementing the identification and development programs for the gifted and talented students would be fully accomplished and successful with a national policy that regulates the services and mandates the ongoing assessment and evaluation. The participation of teachers, parents and all of the stakeholders in the country would promote the awareness regarding the role of each of them and their collaborative responsibility towards the development of the gifted and talented students in the UAE.

References

Abbott, A., Button, C., Pepping, G.J. and Collins, D., 2005. Unnatural selection: talent identification and development in sport. *Nonlinear dynamics, psychology, and life sciences*.

Al Obaidli, A.I.A., 2006. *Educating the Gifted and Talented in the UAE: Status and Recommendations* (Doctoral dissertation, The British University in Dubai (BUiD)).

ALGHAWI, M.A., 2016. *Needs Assessment of Gifted Education Programmes in Dubai; an investigative case study of governmental Primary Schools* (Doctoral dissertation, The British University in Dubai (BUiD)).

AlGhawi, M.A., 2017. Gifted education in the United Arab Emirates. Cogent Education.

Alhebsi, A., Pettaway, L., and Waller, L., 2015. *A history of education in the United Arab Emirates and Trucial Shiekdoms*. The Global eLearning Journal.

Allana, G., 2010. *Is Academic Excellence in School Children Based on IQ:* A review of private schools in Dubai regarding the gifted and talented (Doctoral dissertation, The British University in Dubai (BUiD)).

Alnabah, N., 1996. Education in the United Arab Emirates. Abu Dhabi, UAE: Alflah.

Armstrong, T., 2003. The multiple intelligences of reading and writing: Making the words come alive. ASCD.

Attfield, R., 2009. Developing a Gifted and Talented Strategy: Lessons from the UK Experience/Attfield R. Berkshire, CfBT Education Trust.

Babu, G.R., 2008. Research Methodology in Social Sciences: For M. Phil, MSW, MBA, Pre-Ph. D and Post-doctoral Courses. Concept Publishing Company.

Badri, M., Al Qubaisi, A., Mohaidat, J., Al Dhaheri, H., Yang, G., Al Rashedi, A. and Greer, K., 2016. *An analytic hierarchy process for school quality and inspection: Model development and application.* International Journal of Educational Management.

Baker, C.N. and Hoerger, M., 2012. Parental child-rearing strategies influence self-regulation, socio-emotional adjustment, and psychopathology in early adulthood: Evidence from a retrospective cohort study. *Personality and individual differences*.

Basham, J.D., Israel, M., Graden, J., Poth, R. and Winston, M., 2010. A comprehensive approach to RTI: Embedding universal design for learning and technology. *Learning Disability Quarterly*

Bassey, M., 1999. Case study research in educational settings. McGraw-Hill Education (UK).

Benson, E., 2003. Intelligent intelligence testing. *Monitor on Psychology*.

Betts, G.T. and Neihart, M., 1988. Profiles of the gifted and talented. Gifted child quarterly.

Bloom, B.S., and Sosniak, L.A., 1985. Developing talent in young people. Ballantine Books.

Boring, E.G., 1959. Lewis Madison Terman: 1877–1956. *Biographical memoirs of the National Academy of Sciences*.

Borland, J. H., & Wright, L. (1994). Identifying young, potentially gifted, economically disadvantaged students. *Gifted Child Quarterly*.

Bostrom, R.P., and Nagasundaram, M., 1998, January. Research in Creativity and GSS. *In Proceedings of the Thirty-First Hawaii International Conference on System Sciences*.

Bowen, G.A., 2009. Document analysis as a qualitative research method. *Qualitative research journal*.

Bryan, L., and Joyce, C., 2005. The 21st-century organization. The McKinsey Quarterly, 3, pp.21-29. Cadbury, D., 2006. *Space race: The epic battle between America and the Soviet Union for dominion of space. HarperCollins.*

BUDAIR, S., 2018. Examining the Implementation of Gifted and Talented Pilot Program in ADEK's Public Schools (Doctoral dissertation, The British University in Dubai (BUiD)).

Burns, R.B., and Bursn, R.B., 2000. Introduction to research methods.

Campbell, D.T., and Stanley, J.C., 2015. Experimental and quasi-experimental designs for research. Ravenio Books.

Campbell, J.P., 1971. Personnel training and development. Annual review of psychology,

Campbell, J. R., & Walberg, H. J. (2011). Olympiad studies: Competitions provide alternatives to developing talents that serve national interests. *Roeper Review*.

Campbell, J.R., Wagner, H. and Walberg, H.J., 2000. Academic competitions and programs designed to challenge the exceptionally talented — international handbook of giftedness and talent, 2.

Cantrell, M.A. and Lupinacci, P., 2007. Methodological issues in online data collection. *Journal of advanced nursing*.

Carmines, E.G. and Zeller, R.A., 1979. *Reliability and validity assessment* (Vol. 17). Sage publications.

Ceci, S.J. and Liker, J.K., 1986. A day at the races: A study of IQ, expertise, and cognitive complexity. *Journal of Experimental Psychology: General*.

Chan, D.W., 2011. Characteristics and competencies of teachers of gifted learners: The Hong

Kong student perspective. Roeper Review.

Chapra, M.U., 1979. The Islamic welfare state and its role in the economy. Islamic Foundation.

Clough, P. and Nutbrown, C., 2012. A student's guide to methodology. Sage.

Colangelo, N. and Dettmann, D.F., 1983. A review of research on parents and families of gifted children. *Exceptional children*.

Coleman, L.J., Guo, A. and Dabbs, C.S., 2007. The state of qualitative research in gifted education as published in American journals: An analysis and critIQue. *Gifted Child Ouarterly*.

Coleman, M.R., 2001. Surviving or thriving? 21 gifted boys with learning disabilities share their school stories. *Gifted Child Today*.

Coleman, M.R., 2003. *The identification of students who are gifted*. Arlington, VA: ERIC Clearinghouse on Disabilities and Gifted Education.

Creswell, J.W. and Poth, C.N., 2017. *Qualitative inquiry and research design: Choosing among five approaches*. Sage publications.

Dai, D.Y., 2010. The Nature and Nurture of Giftedness: A New Framework for Understanding Gifted Education. Education & Psychology of the Gifted Series. Teachers College Press. 1234 Amsterdam Avenue, New York, NY 10027.

Davis, G.A. and Rimm, S.B., 2004. Characteristics of gifted students. Education of the gifted and talented.

Davis, K., Christodoulou, J., Seider, S. and Gardner, H.E., 2011. *The theory of multiple intelligences*.

De Lisle, J., 2011. The benefits and challenges of mixing methods and methodologies: Lessons learnt from implementing qualitatively led mixed methods research designs in Trinidad and Tobago.

Denscombe, M., 2014. *The good research guide: for small-scale social research projects*. McGraw-Hill Education (UK).

Denys, E.P., 1973. Philip Melanchthon's Unique Contribution to Education.

Denzin, N.K. and Lincoln, Y.S., 2005. 2005. Handbook of qualitative research.

deS Price, D., 1984. The science/technology relationship, the craft of experimental science, and policy for the improvement of high technology innovation. *Research Policy*.20.

Dickens, W.T. and Flynn, J.R., 2006. Black Americans reduce the racial IQ gap: Evidence from standardization samples. *Psychological Science*.

Donahue, D.M., 1993. Serving students, science, or society? The secondary school physics curriculum in the United States, 1930–65. *History of Education Quarterly*.

Doran, J. and Bizony, P., 2011. Starman: *The truth behind the legend of Yuri Gagarin. Bloomsbury Publishing USA.*

Drew, C.J., Hardman, M.L. and Hosp, J.L., 2007. *Designing and conducting research in education*. Sage Publications.

Eyre, D. and Lowe, H. eds., 2013. Curriculum provision for the gifted and talented in the secondary school. Routledge.

Eyre, D. and McClure, L. eds., 2001. Curriculum provision for the gifted and talented in the primary school: English, Maths, Science and ICT. Routledge.

Eyre, D., 2012. An effective primary school for the gifted and talented. In *Curriculum provision* for the gifted and talented in the primary school. David Fulton Publishers.

Fasko Jr, D., 2001. An analysis of multiple intelligences theory and its use with the gifted and talented. *Roeper Review*.

Feist, G.J. and Runco, M.A., 1993. Trends in the creativity literature: An analysis of research in the Journal of Creative Behavior (1967–1989). *Creativity research journal*.

Feist, G.J., 2006. The development of scientific talent in Westinghouse finalists and members of the National Academy of Sciences. *Journal of adult development*.

Feldhusen, J.F., 1996. How to identify and develop special talents. *Educational Leadership*.

Feldman, D.H., 1993. Child Prodigies: A Distinctive Form of Giftedness1. *Gifted Child Quarterly*

Feldman, D.H., 1999. The development of creativity. *Handbook of creativity*,

Fierros, E.G., 2004. How multiple intelligences theory can guide teachers' practices: Ensuring success for students with disabilities. *Denver, CO: The National Institute for Urban School Improvement*.

Flinders, D.J., 1992. In search of ethical guidance: Constructing a basis for dialogue. *Qualitative studies in education*.

Flynn, J.R., 2007. What is intelligence? Beyond the Flynn effect. Cambridge University Press.

Frankel, J.R., Wallen, N.E. and Hyun, H.H., 2012. *How to design and evaluate research in education*. (8. Baskı).

Friedman, T., 2005. The world is flat: a brief history of the globalised world in the 21st century. *Allen Lane-Penguin Books: London*.

Gaad, E., Arif, M. and Scott, F., 2006. Systems analysis of the UAE education system. *International Journal of Educational Management*.

Gagné, F., 1985. Giftedness and talent: Re-examining a re-examination of the definitions. *Gifted child quarterly*.

Gagné, F., 1995. From giftedness to talent: A developmental model and its impact on the language of the field. *Roeper Review*.

Gagné, F., 2005. From gifts to talents. Conceptions of giftedness.

Gallagher, J.J. and Reis, S.M., 2004. Public policy in gifted education. Corwin Press.

Gallagher, J.J., 1994. Current and Historical Thinking on Education for Gifted and Talented Students.

Galton, F., 1869. Hereditary genius: An inquiry into its laws and consequences. Macmillan.

Galton, Francis (1874). "On men of science, their nature and their nurture". Proceedings of the Royal Institution of Great Britain.

Gardner, H., 1992. Multiple intelligences (Vol. 5, p. 56). Minnesota Center for Arts Education.

Gardner, H., 2004. Globalization: Culture and education in the new millennium (M. Suarez-Orozco & D. Qin-Hilliard, Eds.). Berkeley.

Gardner, H., 2011. Frames of mind: The theory of multiple intelligences. Hachette Uk.

Gardner, H.E., 2000. Intelligence reframed: Multiple intelligences for the 21st century. Hachette

George, D., 2013. Gifted education: Identification and provision. David Fulton Publishers.

Giofrè, D., Pastore, M., Cornoldi, C. and Toffalini, E., 2019. Lumpers vs. splitters: Intelligence in children with specific learning disorders. *Intelligence*.

Gladwell, M., 2008. Outliers: The story of success. Hachette UK.

Golafshani, N., 2003. Understanding reliability and validity in qualitative research. *The qualitative report*.

Gottfredson, L.S., 2005. Suppressing intelligence research: Hurting those we intend to help. *Destructive trends in mental health: The well-intentioned path to harm.*UK.

Gould, S.J., 1996. *The mismeasure of man*. WW Norton & Company.

Greene, M.J., 2006. Helping build lives: Career and life development of gifted and talented students. *Professional School Counseling*.

Grigorenko, E.L. and Sternberg, R.J., 2001. Analytical, creative, and practical intelligence as

predictors of self-reported adaptive functioning: A case study in Russia. *Intelligence*.

Gross, M.U., 2006. Exceptionally gifted children: Long-term outcomes of academic acceleration and nonaccelerating. *Journal for the Education of the Gifted*.

Hackett, P., 1998. The selection interviews. CIPD Publishing.

Hally, T., 2015. A Brief History of IQ tests. *Pridobljeno*.

Hansen, J.B. and Feldhusen, J.F., 1994. Comparison of trained and untrained teachers of gifted students. *Gifted Child Quarterly*.

Heller, K.A., Mönks, F.J., Subotnik, R. and Sternberg, R.J. eds., 2000. International handbook of giftedness and talent. *Elsevier*.

Hertberg-Davis, H., 2009. Myth 7: Differentiation in the regular classroom is equivalent to gifted programs and is sufficient: Classroom teachers have the time, the skill, and the will to differentiate adequately. *Gifted Child Quarterly*.

Hew, K.F. and Brush, T., 2007. Integrating technology into K-12 teaching and learning: Current knowledge gaps and recommendations for future research. *Educational technology research and development*.

Hilliard, A.G., 1994. What good is this thing called intelligence and why bother to measure it?. *Journal of Black Psychology*.

Hodge, K., 2013. Gifted and talented education. ACT Government. Education and Training.

Hoerr, T.R., 2000. Becoming a multiple intelligences school. ASCD.

Hollingworth, L.S., 1920. The psychology of subnormal children. Macmillan.

Hollingworth, L.S., 2015. Children above 180 IQ Stanford-Binet: *origin and development. World Book Company*.

Holmes, R., Dahan, H.M. and Ashari, H., 2005. *A guide to research in the social sciences*. Pearson/Prentice Hall.

House, F.B., Gruber, A., Hunt, G.E. and Mecherikunnel, A.T., 1986. History of satellite missions and measurements of the Earth radiation budget (1957–1984). *Reviews of geophysics*.

Huitt, W. and Hummel, J., 2003. Piaget's theory of cognitive development. *Educational psychology interactive*.

Hurd, P.D., 1958. Science literacy: Its meaning for American schools. *Educational leadership*.

Hvidt, M., 2007. Public–private ties and their contribution to development: The case of Dubai. *Middle Eastern Studies*.

Jensen, A.R., 2002. Galton's legacy to research on intelligence. *Journal of Biosocial Science*.

Jolly, J.L., 2009. Historical perspectives: The national defense education act, current STEM initiative, and the gifted. *Gifted Child Today*.

Joppe, G., 2000. Testing reliability and validity of research instruments. *Journal of American Academy of Business Cambridge*.

Kamin, L.J., 2012. The science and politics of IQ. Routledge.

Kell, H. J., Lubinski, D., & Benbow, C. P. (2013). Who rises to the top? Early indicators. *Psychological Science*.

Kendall, L., 2005. Excellence in cities: *The national evaluation of a policy to raise standards in urban schools* 2000-2003.

Lai, F., Liu, C., Luo, R., Zhang, L., Ma, X., Bai, Y., Sharbono, B. and Rozelle, S., 2014. The education of China's migrant children: The missing link in China's education system. *International Journal of Educational Development*.

Lanvin, B. and Evans, P., 2016. The Global Talent Competitiveness Index 2017 Talent and Technology. Fontainebleau, France: *INSEAD*, *Adecco*, *and HCLI*.

Legg, S. and Hutter, M., 2007. A collection of definitions of intelligence. *Frontiers in Artificial Intelligence and applications*,.

Leslie, M., 2000. The vexing legacy of Lewis Terman.

Li, L. and Delisle, J.R., 1990. Gifted education in China. Gifted Education International.

Lippman, W., 1922. Debunking intelligence experts: Walter Lippmann speaks out. *Six articles archived from the New Republic*.

Long, T., Fallon, D., Murphy, M., Derbyshire, D. and Hewitt-Craft, L., 2010. Understanding the children's social care workforce.

Lubinski, D. and Benbow, C.P., 1994. The study of mathematically precocious youth: The first three decades of a planned 50-year study of intellectual talent. *Beyond Terman: Contemporary longitudinal studies of giftedness and talent.*

Lubinski, D., Webb, R.M., Morelock, M.J. and Benbow, C.P., 2001. Top 1 in 10,000: A 10-year follow-up of the profoundly gifted. *Journal of applied Psychology*.

Maccoby, E.E. and Jacklin, C.N., 1974. Myth, reality and shades of gray-what we know and dont know about sex differences. *Psychology Today*.

Macleod, R. B. (1938). William Stern (1871-1938). Psychological Review..

Mansour, K., 2011. Current Status of Gifted and Talented Learners in Cycle 2 Government

Schools in Abu Dhabi Emirate: An Analytical Study (Doctoral dissertation, The British University in Dubai (BUiD)).

Martin, A.J., 2005. Exploring the effects of a youth enrichment program on academic motivation and engagement. *Social Psychology of Education*.

Matthews, D., 1988. Gardner's multiple intelligence theory: An evaluation of relevant research literature and a consideration of its application to gifted education. *Roeper Review*.

Matthews, M.S. and Shaunessy, E., 2010. Putting standards into practice: Evaluating the utility of the NAGC pre-K—grade 12 gifted program standards. *Gifted Child Quarterly*.

Mazumder, L.K., 2016. Restructuring the economy through sustainability initiatives in UAE-a case study analysis. *The Business & Management Review*.

McClain, M.C. and Pfeiffer, S., 2012. Identification of gifted students in the United States today: A look at state definitions, policies, and practices. *Journal of Applied School Psychology*.

McDougall, W.A., 1985. Sputnik, the space race, and the Cold War. *Bulletin of the Atomic Scientists*.

McGrew, K.S., 2009. CHC theory and the human cognitive abilities project: Standing on the shoulders of the giants of psychometric intelligence research.

Merriam, S.B., 1988. Case study research in education: A qualitative approach. Jossey-Bass.

Mertens, D.M., 2010. Transformative mixed methods research. *Qualitative inquiry*.

Mertens, D.M., 2014. Research and evaluation in education and psychology: Integrating diversity with quantitative, qualitative, and mixed methods. Sage publications.

Merzon, E.E., Panfilov, A.N., Panfilova, V.M. and Shterts, O.M., 2013. The Multifactor Pedagogical Model of the Development of Giftedness in Children and Youth. *Middle-East Journal of Scientific Research*.

Miles, M.B. and Huberman, A.M., 1994. *Qualitative data analysis: An expanded sourcebook.* sage.

Mönks, F.J., 1992. Development of gifted children: The issue of identification and programming. *Talent for the future*.

Neelands, J.G.A., Band, S., Freakley, V. and Lindsay, G., 2006. Hidden talents: A review of state-supported provision and policy for talented pupils in England. *Occasional Paper*, 2.

Neihart, M., Reis, S.M., Robinson, N. and Moon, S., 2002. *The social and emotional development of gifted children: What do we know?* Sourcebooks, Inc.

Neu, T.W., Baum, S.M. and Cooper, C.R., 2004. Talent development in science: A unique tale of one student's journey. *Journal of Secondary Gifted Education*.

Neubauer, A.C., Jauk, E., Benedek, M. and Dunst, B., 2014. The relationship between intelligence and creativity: New support for the threshold hypothesis by means of empirical breakpoint detection. *Personality and Individual Differences*.

Nikolova, K. and Taneva-Shopova, S., 2007. Multiple intelligences theory and educational practice. *Annual Assen Zlatarov*.

O. Nyumba, T., Wilson, K., Derrick, C.J. and Mukherjee, N., 2018. The use of focus group discussion methodology: Insights from two decades of application in conservation. *Methods in Ecology and evolution*.

O'Sullivan, A., 2004. Reading and Arab college students: Issues in the United Arab Emirates higher colleges of technology. *In Proceedings of the First International Online Conference on Second and Foreign Language Teaching and Research: Beyond Borders*.

Orleman, P. and Borner, H., 2008. The talent supply chain: leveraging the global talent market. In *From Strategy to Execution*. Springer, Berlin, Heidelberg.

Painter, D., 2002. The Cold War: an international history. Routledge.

Pal, H.R., Pal, A. and Tourani, P., 2004. Theories of intelligence. Everyman's science.

Palinkas, L.A., Horwitz, S.M., Green, C.A., Wisdom, J.P., Duan, N. and Hoagwood, K., 2015. Purposeful sampling for qualitative data collection and analysis in mixed method implementation research. *Administration and policy in mental health and mental health services research*.

Park, G., Lubinski, D. and Benbow, C.P., 2007. Contrasting intellectual patterns predict creativity in the arts and sciences: Tracking intellectually precocious youth over 25 years. *Psychological Science*.

Parker, J.P., 1996. NAGC standards for personnel preparation in gifted education: A brief history. *Gifted Child Quarterly*.

Passow, A.H. and Rudnitski, R.A., 1994. Transforming policy to enhance educational services for the gifted. *Roeper Review*.

Perkins, D., 1995. Outsmarting IQ: *The emerging science of learnable intelligence*. Simon and Schuster.

Pfeiffer, S.I. ed., 2008. *Handbook of giftedness in children: Psychoeducational theory, research, and best practices.* Springer Science & Business Media.

Pinker, S., 2010. The cognitive niche: Coevolution of intelligence, sociality, and language. *Proceedings of the National Academy of Sciences*,.

Piske, F.H.R., Stoltz, T. and Machado, J., 2014. Creative Education for Gifted Children. *Online Submission*.

Plomin, R., 2001. Genetics and behavior. *PSYCHOLOGIST-LEICESTER*.

Punch, K.F., 2013. *Introduction to social research: Quantitative and qualitative approaches.* sage.

Purcell, J.H. and Eckert, R.D., 2006. *Designing services and programs for high-ability learners:* A guidebook for gifted education. Corwin Press.

Purposeful sampling for qualitative data collection and analysis in mixed method implementation research. *Administration and policy in mental health and mental health services research*.

Rasmussen, A. and Lingard, B., 2018. Excellence in education policies: Catering to the needs of gifted and talented or those of self-interest? *European Educational Research Journal*.

Reid, E. and Horváthová, B., 2016. Teacher training programs for gifted education with focus on sustainability. *Journal of Teacher Education for Sustainability*.

Reid, E., 2015. Development of Gifted Education and an Overview of Gifted Education in the USA, Canada, Equator, and Mexico. *Slavonic Pedagogical Studies Journal*.

Reid, H.M., 2013. *Introduction to statistics: fundamental concepts and procedures of data analysis*. Sage Publications.

Reis, S.M., 1995. Talents in Two Places: Case Studies of High Ability Students with Learning Disabilities Who Have Achieved. Research Monograph 95114.

Renzulli, J.S., 2012. Reexamining the role of gifted education and talent development for the 21st century: A four-part theoretical approach. *Gifted Child Quarterly*.

Renzulli, J.S., 2016. The three-ring conception of giftedness: *A developmental model for promoting creative productivity*.

Renzulli, Joseph S. & Reis, Sally M.: *The Schoolwide Enrichment Model* - Second Edition; Creative Learning Press, Mansfield 1997.

Rimando, M., Brace, A.M., Namageyo-Funa, A., Parr, T.L., Sealy, D.A., Davis, T.L., Martinez, L.M. and Christiana, R.W., 2015. Data collection challenges and recommendations for early career researchers. *The Qualitative Report*.

Rimm, S.B., 1987. Why do Bright Children under Achieve. Gifted Child Today Magazine.

Culross, R.R., 2015. A Talent Development Perspective on the Olympic Athlete. *American Journal of Sports Science and Medicine*.

Ritchie, J., Lewis, J. and Elam, G., 2003. *Designing and selecting samples* (pp. 77-108). London: Sage.

Robinson, W., Campbell, J., Mazzoli, L. and Hewston, Ruth (2006) Developing Expertise: School-based Case Studies. (Occasional Paper 12). Working Paper. National Academy

for Gifted and Talented Youth, Coventry.

Robson, P., 2002. The economics of international integration. Routledge.

Roth, G. and Dicke, U., 2005. Evolution of the brain and intelligence. *Trends in cognitive sciences*.

Sanjari, M., Bahramnezhad, F., Fomani, F.K., Shoghi, M. and Cheraghi, M.A., 2014. Ethical challenges of researchers in qualitative studies: The necessity to develop a specific guideline. *Journal of medical ethics and history of medicine*, 7.

Sapsford, R. and Jupp, V. eds., 1996. Data collection and analysis. Sage.

Schwartzman, H.B., 1993. Ethnography in organizations (Vol. 27). Sage.

Scott, D. and Usher, R., 2010. *Researching education: Data, methods and theory in educational enquiry*. Bloomsbury Publishing.

Shaunessy, E., 2003. State policies regarding gifted education. *Gifted Child Today*.

Shavinina, L.V., 2009. On giftedness and economy: The impact of talented individuals on the global economy. *In International handbook on giftedness*. Springer, Dordrecht.

Shields, P.M. and Rangarajan, N., 2013. A playbook for research methods: Integrating conceptual frameworks and project management. New Forums Press.

Shurkin, J.N., 1992. *Terman's kids: The groundbreaking study of how the gifted grow up*. Little, Brown and Co.

Siegle, D., 2004. The merging of literacy and technology in the 21st century: A bonus for gifted education. *Gifted Child Today*.

Silverman, D., 2006. Interpreting qualitative data: Methods for analyzing talk, text and interaction. Sage.

keith Simonton, D., 2016. Early and late bloomers among 120 classical composers: Were the greatest geniuses also prodigies? *Musical Prodigies: Interpretations from Psychology, Education, Musicology, and Ethnomusicology.*

Simonton, D.K., 1999. Talent and its development: An emergenic and epigenetic model. *Psychological review*.

Simonton, D.K., 2000. Genius and giftedness: Same or different. *International handbook of giftedness and talent*.

Simonton, D.K., 2012. Quantifying creativity: can measures span the spectrum?. *Dialogues in clinical neuroscience*.

Siti Nur Fazilah, A., 2015. Awareness and acceptance of Singaporean Muslims towards Islamic

banking products and services/Siti Nur Fazilah Abdul Aziz (Doctoral dissertation, Universiti Malaya).

Smil, V., 2005. Creating the twentieth century: Technical innovations of 1867-1914 and their lasting impact. Oxford University Press.

Smith, C., 2006. Multiple cultures, multiple intelligences: Applying cognitive theory to usability of digital libraries. *Libri*.

Smith, P.L. and Ragan, T.J., 1996. Impact of RM Gagne's Work on Instructional Theory.

Spearman, C. (2005). The Abilities of Man: Their Nature and Measurement. *The Blackburn Press*

Stanley, J.C., 1996. In the Beginning: The Study of Mathematically Precocious Youth (SMPY).

Stern, W., 1914. The psychological methods of testing intelligence (No. 13). Warwick & York. Terman, L.M., 1940. *Psychological approaches to the biography of genius. Science*.

Sternberg, R.J. and Grigorenko, E.L., 2002. The theory of successful intelligence as a basis for gifted education. *Gifted Child Quarterly*.

Sternberg, R.J. and O'Hara, L.A., 1999. 13 Creativity and Intelligence. *Handbook of creativity*, 251.

Sternberg, R.J. and Sternberg, R.J. eds., 1982. *Handbook of human intelligence*. CUP Archive.

Sternberg, R.J., 1983. How much Gall is too much gall? Review of Frames of Mind: The theory of multiple intelligences. *Contemporary Education Review*.

Sternberg, R.J., 1986. A triarchic theory of human intelligence. In *Human assessment: Cognition and motivation*. Springer, Dordrecht.

Sternberg, R.J., 1996. IQ counts, but what really counts is successful intelligence. *NASSP Bulletin*.

Sternberg, R.J., 2005. The theory of successful intelligence. *Interamerican Journal of Psychology*.

Sternberg, R.J., 2012. Contemporary theories of intelligence. *Handbook of Psychology, Second Edition*, 7.

Sternberg, Robert J & Davidson, Janet E. (ed.): *Conceptions of Giftedness -* Second Edition; Cambridge University Press, Cambridge 2005.

Sternberg, Robert J. *Beyond IQ: A Triarchic Theory of Human Intelligence*. Cambridge University Press, 1985.

Stutchbury, K. and Fox, A., 2009. Ethics in educational research: introducing a methodological

tool for effective ethical analysis. Cambridge Journal of Education.

Subhi-Yamin, T., 2009. Gifted education in the Arabian Gulf and the Middle Eastern regions: History, current practices, new directions, and future trends. *In International handbook on giftedness. Springer, Dordrecht.*

Subotnik, R.F., Duschl, R.A. and Selmon, E.H., 1993. Retention and attrition of science talent: A longitudinal study of Westinghouse Science Talent Search winners. *International Journal of Science Education*.

Subotnik, R.F., Olszewski-Kubilius, P. and Worrell, F.C., 2011. Rethinking giftedness and gifted education: A proposed direction forward based on psychological science. *Psychological science in the public interest*.

Tannenbaum, A.J., 1986. Giftedness: A psychosocial approach. Conceptions of giftedness

Tavakol, M. and Dennick, R., 2011. Making sense of Cronbach's alpha. *International journal of medical education*.

Taylor, S., 2017. Contested Knowledge: A Critical Review of the Concept of Differentiation in Teaching and Learning. *Warwick Journal of Education—Transforming Teaching*.

Tellis, W.M., 1997. Introduction to case study. *The qualitative report*.

Thomson, G.H. (1947). "Charles Spearman. 1863-1945". Obituary Notices of Fellows of the Royal Society.

Tuttle Jr, F.B., 1988. *Characteristics and identification of gifted and talented students*. NEA professional Library, PO Box 509, West Haven, CT 06516.

UNESCO, I., 2010. World data on education. *Principles and General Objectives of Education*.

Vaeyens, R., Lenoir, M., Williams, A.M. and Philippaerts, R.M., 2008. Talent identification and development programmes in sport. *Sports medicine*.

VanTassel-Baska, J., 2000. The on-going dilemma of effective identification practices in gifted education. *The Communicator*.

Van Tassel-Baska, J., 2006. State policies in gifted education. *Designing services and programs for high-ability learners: A guidebook for gifted education.*

VanTassel-Baska, J., 2009. United States policy development in gifted education: A patchwork quilt. In *International handbook on giftedness*. Springer, Dordrecht

Visser, B.A., Ashton, M.C. and Vernon, P.A., 2006. g and the measurement of Multiple Intelligences: A response to Gardner. *Intelligence*.

Wadsworth, B.J., 1996. *Piaget's theory of cognitive and affective development: Foundations of constructivism.* Longman Publishing.

Wagner, T. and Compton, R.A., 2012. Creating innovators: The making of young people who will change the world. Simon and Schuster.

Walliman, N., 2017. Research methods: The basics. Routledge.

Warne, R.T., 2019. An evaluation (and vindication?) of Lewis Terman: What the father of gifted education can teach the 21st century. *Gifted Child Quarterly*.

Waterhouse, L., 2006. Multiple intelligences, the Mozart effect, and emotional intelligence: A critical review. *Educational Psychologist*.

Watson, D., 2006. Understanding the relationship between ICT and education means exploring innovation and change. *Education and Information Technologies*.

Webb, J.T., Gore, J.L. and Amend, E.R., 2007. *A parent's guide to gifted children*. Great Potential Press, Inc..

Weinberg, R.A., 1989. Intelligence and IQ: Landmark issues and great debates. *American psychologist*.

Wellisch, M. and Brown, J., 2012. An integrated identification and intervention model for intellectually gifted children. *Journal of Advanced Academics*.

Wellisch, M., 2016. Gagne's DMGT and underachievers: The need for an alternative inclusive model. *Australasian Journal of Gifted Education*.

Westberg, K. L. (1999, Summer). What happens to young, creative producers? NAGC: *Creativity and Curriculum Division Newsletter*.

Wiggins, G., 1998. Educative Assessment. Designing Assessments to Inform and Improve Student Performance. Jossey-Bass Publishers, 350 Sansome Street, San Francisco, CA 94104.

Wilson, G., 1999. Father of Dubai: Sheikh Rashid bin Saeed Al-Maktoum. Media Prima.

Yin, R.K., 1994. Discovering the future of the case study. Method in evaluation research. *Evaluation practice*.

Yin, R.K., 2014. Case study research: design and methods 5th ed. Thousand Oaks.

Yin, R.K., 2017. Case study research and applications: Design and methods. Sage publications.

Zabloski, J. and Milacci, F., 2012. Gifted Dropouts: Phenomenological Case Studies of Rural Gifted Students. *Journal of Ethnographic & Qualitative Research*, 6(3).

Zhang, Z., 2017. Gifted education in China. Cogent Education.

Ziegler, A., Stoeger, H., Harder, B. and Balestrini, D.P., 2013. Gifted education in German-speaking Europe. *Journal for the Education of the Gifted*.

Online Resources

Ibrahim, Mohammed. (2016). *National Special Curriculum for UAE talented students by Al Khaleej Newspaper*. Available at: www.alkhaleej.ae/alkhaleej/page/05518632-14ac-4a86-95e5-09f7f1bd6c4b.

Accessed April 17, 2019

Hamdan Bin Rashid Awards (2019). *The national plan for gifted programs in the UAE*. Available at: ha.ae/en/web/guest/the-national-plan-for-gifted-programs-in-the-uae . Accessed May 3 2019

Oxford dictionaries. *Definition of Identification in English*. Available at https://en.oxforddictionaries.com/definition/identification. Accessed May 4, 2019

Encyclopedia of Education. *Gifted and Talented Education*. Available at: https://www.encyclopedia.com/education/encyclopedias-almanacs-transcripts-and-maps/gifted-and-talented-education.

Accessed May 05, 2019

National Association for Gifted Children. *Jacob Javits Gifted & Talented Students Education Act*. Available at: https://www.nagc.org/resources-publications/resources-university-professionals/jacob-javits-gifted-talented-students
Accessed May 5, 2019

Emirates, U.A. (2016). *United Arab Emirates' Constitution of 1971 with Amendments through 2004*. Available at:

https://www.constituteproject.org/constitution/United Arab Emirates 2004.pdf Accessed May 9, 2019

UAE Centennial 2071. *Government .ae*. Available at: https://government.ae/en/about-the-uae/strategies-initiatives-and-awards/federal-governments-strategies-and-plans/uae-centennial-2071.

Accessed May 9, 2019

Washington DC. Sheikh Zayed bin Sultan Al Nahyan, Founder of the UAE. Embassy of the United Arab Emirates. Available at: https://www.uae-embassy.org/about-uae/history/sheikh-zayed-bin-sultan-al-nahyan-founder-uae. Accessed on May 17, 2019

Future Skills for Youth: Available at https://www.government.ae/information-and-services/jobs/future-skills-for-youth.

Accessed on May 24th, 2019

National Innovation Strategy. The Official Portal of the UAE Government. Available at: https://government.ae/en/about-the-uae/strategies-initiatives-and-awards/federal-governments-strategies-and-plans/national-innovation-strategy. Accessed May 26, 2019

Education in the UAE. Embassy of the United Arab Emirates, Washington D.C. Available at https://www.uae-embassy.org/about-uae/education-uae. Accessed May 26, 2019

Ahmed Afshan (2011). *The UAE's History Lesson*. Available at: https://www.thenational.ae/uae/the-uae-s-history-lesson-1.460608. Accessed May 26, 2019

Clark, John Bates. Available at:

https://encyclopedia2.thefreedictionary.com/John+Bates+Clark.

Accessed on November 8, 2019

The Multifactor Pedagogical Model of the Development of Giftedness in Children and Youth .Available at

https://pdfs.semanticscholar.org/fd50/f7f7bb918fdd9bef0603cc5e9450e5799545.pdf Accessed on November 8, 2019

Tests & Assessments. Available at:

https://pdfs.semanticscholar.org/fd50/f7f7bb918fdd9bef0603cc5e9450e5799545.pdf Accessed on November 8, 2019

Academic Competitions and Programs Designed to Challenge the Exceptionally Talented. Available at:

http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.535.1138&rep=rep1&type=pdf Accessed on November 8, 2019

A Brief History of Gifted and Talented Education. Available at:

https://www.nagc.org/resources-publications/resources/gifted-education-us/brief-history-gifted-and-talented-education

Accessed on November 8, 2019

Science Literacy: It's meaning for American Schools. Available at: http://edcipr.com/wp-content/uploads/2016/09/Hurd_1958_Science-literacy.pdf

Accessed on November 8, 2019

Gifted Education in the U.S. Available at https://www.nagc.org/resources-

publications/resources/gifted-education-us

Accessed on November 8, 2019

Gifted Education. Available at: https://en.wikipedia.org/wiki/Gifted_education
Accessed on November 8, 2019

Creativity. Available at: https://en.wikipedia.org/wiki/Creativity Accessed on November 8, 2019

Renzulli's Three-Ring Conception of Giftedness. Available at: http://www.gigers.com/matthias/gifted/three_rings.html
Accessed on November 8, 2019

The Three-Ring Conception of Giftedness: A Developmental Model For Promoting Creative Productivity. Available at: https://pdfs.semanticscholar.org/b0a6/0bdc11e9fefa8c2b1ae71f0fbd324c8fac12.pdf Accessed on November 8, 2019

Intelligence. Available at: https://allpsych.com/psychology101/intelligence/ Accessed on November 8, 2019

Galton, Francis. Available at: https://www.intelltheory.com/galton.shtml Accessed on November 8, 2019

The official, authoritative site of multiple intelligences. Available at: https://www.multipleintelligencesoasis.org/
Accessed on November 8, 2019

Intelligence. Available at: https://www.apa.org/topics/intelligence/ Accessed on November 8, 2019

A Collection of Definitions of Intelligence. Available at: https://arxiv.org/pdf/0706.3639.pdf
Accessed on November 8, 2019

What is Intellectual Giftedness? What does Intellectual Giftedness mean? Available at: https://www.youtube.com/watch?v=zA6dfBSGVp8
Accessed on November 8, 2019

Intellectual Giftedness. Available at: https://en.wikipedia.org/wiki/Intellectual_giftedness Accessed on November 8, 2019

Spearman, Charles. Available at: https://en.wikipedia.org/wiki/Charles_Spearman#_Theory_of_Intelligence Accessed on November 8, 2019

Piaget's theory of cognitive development. Available at: https://en.wikipedia.org/wiki/Piaget%27s theory of cognitive development Accessed on November 8, 2019

Armstrong, Thomas (2005). Multiple Intelligences of Reading and Writing. Available at: http://www.ascd.org/publications/books/102280/chapters/Literacy,-Multiple-Intelligences,-and-the-Brain.aspx

Accessed on November 9, 2019

Herndon, Eve. (2018). What are Multiple Intelligences and how do they affect learning? Available at: https://www.cornerstone.edu/blogs/lifelong-learning-matters/post/what-are-multiple-intelligences-and-how-do-they-affect-learning
Accessed on November 9, 2019

National Agenda. Available at: https://www.vision2021.ae/en/national-agenda-2021 Accessed on November 9,2019

Education in UAE-K-12 Education. Available at: http://www.uaecd.org/k-12-education Accessed on November 9,2019

Education as a national priority. Available at:

https://www.government.ae/en/about-the-uae/leaving-no-one-behind/4qualityeducation Accessed on November 9, 2019

Ministry of Education Strategic Plan 2017-2021. Available at: https://www.moe.gov.ae/En/AboutTheMinistry/Pages/MinistryStrategy.aspx Accessed on November 9, 2019

Gokulan, Dhanusha (2018). Education in the UAE: Then, now and tomorrow. Available at: https://www.khaleejtimes.com/kt-40-anniversary/education-in-the-uae-then-now-and-tomorrow

Accessed on November 9, 2019

Emirates Foundation. Available at: https://www.emiratesfoundation.ae/ef/about-ef/ Accessed on November 9, 2019

The Abu Dhabi Education Council (ADEK). Available at: https://www.adek.gov.ae/https://www.taaleem.ae/en/academics/adec.html Accessed on November 9, 2019

Special Education Department. Available at: https://cedu.uaeu.ac.ae/en/departments/special/index.shtml Accessed on November 9, 2019

Gifted and Creativity Center. Available at: http://www.ha.ae/en/web/guest/about-the-

center

Accessed on November 9, 2019

The Role of International Assessments and School Inspections in the Reform of Education in Dubai. Available at: https://www.cambridgeassessment.org.uk/images/126095-seminar-the-role-of-international-assessments-and-school-inspections-in-the-reform-of-education-in-dubai-dr-abdulla-al-karam-jameela-al-muhairi-fatma-al-marri.pdf
Accessed on November 10, 2019

University Quality Assurance International Board. Available at: https://www.khda.gov.ae/CMS/WebParts/TextEditor/Documents/UQAIB_EN.pdf
Accessed on November 10, 2019

*UAE Green Jobs Program: Jobs & Skills for the UAE's Green Economy Transformation.*Available at:

file:///C:/Users/7323/Downloads/UAE%20Green%20Jobs%20Programs_Brochure.pdf Accessed on November 10, 2019

Wadeema law. Available at:

https://www.khda.gov.ae/CMS/WebParts/TextEditor/Documents/Children_Law_English.pdf

Accessed on November 10, 2019

Advanced Skills Strategy. Available at: https://government.ae/en/about-the-uae/strategies-initiatives-and-awards/federal-governments-strategies-and-plans/advanced-skills-strategy Accessed on November 10, 2019

National Advanced Sciences Agenda 2031. Available at: https://government.ae/en/about-the-uae/strategies-initiatives-and-awards/federal-governments-strategies-and-plans/national-advanced-sciences-agenda-2031

Accessed on November 10, 2019

How do I know if a gifted program is of high quality? Available at:

 $\underline{https://www.nagc.org/resources-publications/resources/frequently-asked-questions-about-gifted-education}$

Accessed on November 11, 2019

Why Are Gifted Programs Needed? Available at: https://www.nagc.org/resources-publications/gifted-education-practices/why-are-gifted-programs-needed
Accessed on November 12, 2019

The Dubai Data Publishing and Exchange Act in 2015, (Law No. (26) Of 2015). Available at: https://dlp.dubai.gov.ae/ar/Pages/Default.aspx

Accessed on November 12, 2019

Intelligence. Available at: https://www.dictionary.com/browse/intelligence Accessed on November 13, 2019

Cognitive Ability Test. Available at: https://www.wikijob.co.uk/content/aptitude-tests/test-types/cognitive-ability-test
Accessed on November 13, 2019

Appendix A

Interview Questions Guide

Experts-Teachers-Programs Leaders.

Project Study: The Role of Governmental Educational Institutions in implementing talents' development programs in the UAE

Time of Interview				
Date of Interview				
Place of Interview				
Interviewer				
Interviewee:	Age	Job	qualifications	Years of experience
Gender:				

My name is **Basema Younes.** Thank you for accepting to participate in this research study. The purpose of this interview is to collect data for my PhD project study on the practices implemented by UAE Government Educational Institutions to identify and develop talents. The interviews will provide the researcher with the best information on your perspectives and background knowledge you have on (TID &TDE) in this initiative.

The research questions are:

- 1. What is the role of development programs implemented by governmental, educational institutions to develop gifted and talented students in the UAE?
- 2. How are gifted and talented students defined and identified in the UAE?
- 3 How is the development of gifted and talented students implemented in the UAE?
- 4. What are the main challenges facing the gifted and talented identification and development in the UAE?

What is the talents' definition from your perspective?	
What factors influence talent development (examples)	
What are the identification tools used to detect talent in your institution?	
How do you develop gifted programs?	
How well do you know about gifted programs and talent care programs in the UAE?	
In your opinion, what qualities of talent that facilitate their identification?	
How can gifted and talented develop their talent?	
In your opinion, what are the most obstacles and challenges facing talented people and the families?	ieir
What opportunities are there to promote talent investment in the UAE?	
What are your suggestions for more talent's development in the UAE?	
o you think teachers need special educational provision and training for the G/T?	
your opinion, how likely, the G/T have emotional problems.	

Appendix B

Interview Questions Guide

Parents of Talented Students

Project Study: The Role of Governmental Educational Institutions in implementing talents' development programs in the UAE

Time of Interview				
Date of Interview				
Place of Interview				
Interviewer				
Interviewee	Gender	Age	Nationality	Qualifications

My name is **Basema Younes.** Thank you for accepting to participate in this research study. The purpose of this interview is to collect data for my PhD project study on the practices implemented by UAE Government Educational Institutions to identify and develop talents. The interviews will provide the researcher with the best information on your perspectives and background knowledge you have on (TID &TDE) in this initiative.

The primary research questions are:

- 1. What is the role of development programs implemented by governmental, educational institutions to develop gifted and talented students in the UAE?
- 2. How are gifted and talented students defined and identified in the UAE?
- 3 How is the development of gifted and talented students implemented in the UAE?
- 4. What are the main challenges facing the gifted and talented identification and development in the UAE?

Interview Questions	
Definition of talent from your perspective	1
How did you discover your child's talent?	2
What do you/ school do to develop your child's talent?	3
To what extent do you know about talents program-Policy in the UAE?	4
How did you come to know about the talents' identification programs in the UAE?	5
What are the characteristics that made you recognise your child's talent?	ϵ
What are the challenges you face in raising up your child's talents?	7
What are your suggestions and recommendations regarding talents' development in the UAE?	8
What are the challenges you face in raising up your child's talents? What are your suggestions and recommendations regarding talents' development in the UAE?	

Appendix C

Interview Questions Guide

Talented Individuals

Project Study: The Role of Governmental Educational Institutions in implementing talents' development programs in the UAE

		Interview Ques	tions
the UAE?			
	n challenges faci	ng the gifted and	talented identification and development in
3 How is the develop	pment of gifted a	nd talented studen	its implemented in the UAE?
-			
2. How are gifted an	d talented studer	nts defined and ide	ntified in the UAE?
institutions to develo	op gifted and tale	ented students in th	ne UAE?
1. What is the role o	f development pr	ograms implemen	ted by governmental, educational
The primary resear	rch questions ar	e:	
background knowled	dge you have on	(TID &TDE) in th	is initiative.
interviews will provi	ide the researche	r with the best info	ormation on your perspectives and
		•	tions to identify and develop talents. The
•			project study on the practices
My name is Basema	Younes. Thank	vou for accepting	to participate in this research study. The
Interviewee	Name	class	Type of Talent
Interviewer			
Place of Interview			
Date of Interview			
Time of Interview			

How do you define talents?	11
How did you know you are talented?	2
What supported the development of your talents? Who did?	3
How do you want to benefit from your talent in the future?	4
Did you participate in any development program inside or outside the UAE? How?	6
To what extent did the institution support your talent development?	7
What are the challenges you faced in the UAE in relation to your talents' development?	
Did anybody encourage you? Who and how?	